

between the

BC FAMILY MAINTENANCE AGENCY LTD. (BCFMA)

and the

B.C. GENERAL EMPLOYEES' UNION (BCGEU)

Effective from April 1, 2022 to March 31, 2025

230920v2

1007-151

TABLE OF CONTENTS

1.1 Intent of Agreement. 2 1.2 Future Legislation 3 1.3 Conflict with Regulations. 3 1.4 Human Rights Code 3 ARTICLE 2 - STATEMENT OF MANAGEMENT RIGHTS. 3 2.1 Management Rights. 3 2.2 Exercise of Management Rights. 3 3.1 Bargaining Unit Defined. 3 3.2 Bargaining Unit Recognition 4 3.3 Correspondence. 4 3.4 No Other Agreement. 4 3.5 No Discrimination 4 3.6 Recognition and Rights of Stewards. 4 3.7 Bulletin Boards 5 3.8 Union Insignia 5 3.9 Right to Refuse to Cross Picket Lines 5 3.10 Time Off for Union Business 5 3.11 Union Nop Card 6 3.12 Notification of Staff Representatives 6 3.13 Shop Steward Representatives 7 5.1 Deduction of Dues 7 5.2 Remitting of Dues and	ARTICLE 1 - I	NTRODUCTION	2
1.3 Conflict with Regulations 3 1.4 Human Rights Code 3 ARTICLE 2 - STATEMENT OF MANAGEMENT RIGHTS 3 2.1 Management Rights 3 2.2 Exercise of Management Rights 3 3.1 Bargaining Unit Defined 3 3.2.1 Bargaining Unit Defined 3 3.2 Bargaining Unit Recognition 4 3.3 Correspondence 4 3.4 No Other Agreement 4 3.5 No Discrimination 4 3.6 Recognition and Rights of Stewards 4 3.7 Bulletin Boards 5 3.8 Union Insignia 5 3.9 Right to Refuse to Cross Picket Lines 5 3.10 Time Off for Union Business 5 3.11 Union Shop Card 6 3.12 Notification of Staff Representatives 6 3.13 Shop Steward Representatives 7 5.1 Deduction of Dues 7 5.2 Remitting of Dues and Employee Information 7 5.3	1.1	Intent of Agreement	2
1.4 Human Rights Code 3 ARTICLE 2 - STATEMENT OF MANAGEMENT RIGHTS 3 2.1 Management Rights. 3 3.2.2 Exercise of Management Rights. 3 3.1 Bargaining Unit Defined. 3 3.1 Bargaining Unit Defined. 3 3.2 Bargaining Unit Recognition 4 3.3 Correspondence. 4 3.4 No Other Agreement 4 3.5 No Discrimination 4 3.6 Recognition and Rights of Stewards 4 3.7 Bulletin Boards 5 3.8 Union Insignia 5 3.9 Right to Refuse to Cross Picket Lines 5 3.10 Time Off for Union Business. 5 3.11 Union Shop Card 6 3.12 Notification of Staff Representatives 6 3.13 Shop Steward Representatives 7 5.1 Deduction of Dues 7 5.2 Remitting of Dues and Employee Information 7 5.3 Exclusive Membership Rights 8 ARTICLE 5 -	1.2	Future Legislation	3
ARTICLE 2 - STATEMENT OF MANAGEMENT RIGHTS 3 2.1 Management Rights 3 2.2 Exercise of Management Rights 3 3.2.2 Exercise of Management Rights 3 ARTICLE 3 - UNION RECOGNITION AND RIGHTS 3 3.1 Bargaining Unit Defined 3 3.2 Bargaining Unit Recognition 4 3.3.1 Bargaining Unit Recognition 4 3.4 No Other Agreement 4 3.5 No Discrimination 4 3.6 Recognition and Rights of Stewards 4 3.7 Bulletin Boards 5 3.8 Union Insignia 5 3.9 Right to Refuse to Cross Picket Lines 5 3.10 Time Off for Union Business 5 3.11 Union Shop Card 6 3.12 Notification of Staff Representatives 6 3.13 Shop Steward Representation 7 4.1 Union Membership 7 5.1 Deduction of Dues 7 5.2 Remitting of Dues and Employee Information 7 5.3<	1.3	Conflict with Regulations	3
2.1 Management Rights. 3 2.2 Exercise of Management Rights. 3 ARTICLE 3 - UNION RECOGNITION AND RIGHTS. 3 3.1 Bargaining Unit Defined. 3 3.2 Bargaining Unit Defined. 3 3.3 Correspondence. 4 3.4 No Other Agreement. 4 3.5 No Discrimination 4 3.6 Recognition and Rights of Stewards 4 3.7 Bulletin Boards. 5 3.8 Nion Insignia 5 3.9 Right to Refuse to Cross Picket Lines 5 3.10 Time Off for Union Business 5 3.11 Union Insignia 5 3.12 Notification of Staff Representatives 6 3.13 Shop Steward Representatives 6 3.13 Shop Steward Representation 7 4.1 Union Membership 7 5.2 Remitting of Dues and Employee Information 7 5.3 Exclusive Membership Rights 8 ARTICLE 5 - UNION DUES 7 5.3 Exclusive Membership Rights </td <td>1.4</td> <td>Human Rights Code</td> <td>3</td>	1.4	Human Rights Code	3
2.2 Exercise of Management Rights 3 ARTICLE 3 - UNION RECOGNITION AND RIGHTS 3 3.1 Bargaining Unit Defined 3 3.2 Bargaining Unit Recognition 4 3.3 Correspondence 4 3.4 No Other Agreement 4 3.5 No Discrimination 4 3.6 Recognition and Rights of Stewards 4 3.7 Bulletin Boards 5 3.8 Union Insignia 5 3.9 Right to Refuse to Cross Picket Lines 5 3.10 Time Off for Union Business 5 3.11 Union Shop Card 6 3.12 Notification of Staff Representatives 6 3.13 Shop Steward Representation 6 3.14 Union Membership 7 ARTICLE 4 - UNION SECURITY 7 7 4.1 Union Membership 7 5.2 Remitting of Dues and Employee Information 7 5.3 Exclusive Membership Rights 8 ARTICLE 5 - UNION DUES 7 5.3 5.3 <t< td=""><td>ARTICLE 2 - S</td><td>TATEMENT OF MANAGEMENT RIGHTS</td><td>3</td></t<>	ARTICLE 2 - S	TATEMENT OF MANAGEMENT RIGHTS	3
2.2 Exercise of Management Rights 3 ARTICLE 3 - UNION RECOGNITION AND RIGHTS 3 3.1 Bargaining Unit Defined 3 3.2 Bargaining Unit Recognition 4 3.3 Correspondence 4 3.4 No Other Agreement 4 3.5 No Discrimination 4 3.6 Recognition and Rights of Stewards 4 3.7 Bulletin Boards 5 3.8 Union Insignia 5 3.9 Right to Refuse to Cross Picket Lines 5 3.10 Time Off for Union Business 5 3.11 Union Shop Card 6 3.12 Notification of Staff Representatives 6 3.13 Shop Steward Representation 6 3.14 Union Membership 7 ARTICLE 4 - UNION SECURITY 7 7 4.1 Union Membership 7 5.2 Remitting of Dues and Employee Information 7 5.3 Exclusive Membership Rights 8 ARTICLE 5 - UNION DUES 7 5.3 5.3 <t< td=""><td>2.1</td><td>Management Rights</td><td>3</td></t<>	2.1	Management Rights	3
3.1 Bargaining Unit Defined	2.2		
3.1 Bargaining Unit Defined		INION RECOGNITION AND RIGHTS	z
3.2 Bargaining Unit Recognition 4 3.3 Correspondence 4 3.4 No Other Agreement 4 3.5 No Discrimination 4 3.6 Recognition and Rights of Stewards 4 3.7 Bulletin Boards 5 3.8 Union Insignia 5 3.9 Right to Refuse to Cross Picket Lines 5 3.10 Time Off for Union Business 5 3.10 Time Off for Union Business 5 3.10 Time Off for Union Staff Representatives 6 3.13 Shop Steward Representation 6 ARTICLE 4 - UNION SECURITY 7 7 4.1 Union Membership 7 5.2 Remitting of Dues 7 5.3 Rewitting of Dues and Employee Information 7 5.3 Rewitting of Dues and Employee Information 7 5.3 Rewitting of Dues and Employee Information 7 5.4 Reduiting of Dues and Employee Information 7 5.3 Exclusive Membership Rights 8 7.1 7.1 Repr			
3.3 Correspondence 4 3.4 No Other Agreement 4 3.5 No Discrimination 4 3.6 Recognition and Rights of Stewards 4 3.6 Recognition and Rights of Stewards 5 3.8 Union Insignia 5 3.9 Right to Refuse to Cross Picket Lines 5 3.10 Time Off for Union Business. 5 3.11 Union Shop Card 6 3.12 Notification of Staff Representatives 6 3.13 Shop Steward Representation. 7 ARTICLE 4 - UNION SECURITY. 7 7 4.1 Union Membership 7 5.2 Remitting of Dues and Employee Information 7 5.3 Exclusive Membership Rights 8 ARTICLE 5 - UNION /EMPLOYEES WITH AGREEMENT 8 6.1 7.2 Remitting of Dues and Employee Information 7 5.3 Exclusive Membership Rights 8 ARTICLE 7 - UNION/EMPLOYER RELATIONS 8 7.1 Representation 8 7.2 Union Regresentation 9			
3.4 No Other Agreement 4 3.5 No Discrimination 4 3.6 Recognition and Rights of Stewards 4 3.7 Bulletin Boards 5 3.8 Union Insignia 5 3.9 Right to Refuse to Cross Picket Lines 5 3.10 Time Off for Union Business 5 3.11 Union Shop Card 6 3.12 Notification of Staff Representatives 6 3.13 Shop Steward Representation 7 4.1 Union Membership 7 4.1 Union Membership 7 5.2 Remitting of Dues and Employee Information 7 5.3 Exclusive Membership Rights 8 ARTICLE 5 - UNION DUES 7 5.2 Remitting of Dues and Employee Information 7 5.3 Exclusive Membership Rights 8 8 8 1 ARTICLE 7 - UNION/EMPLOYEES WITH AGREEMENT 8 8 1 1 6.1 Process 8 7.1 Representation 8 7.1 Representation 8 7.1 8 </td <td></td> <td></td> <td></td>			
3.5 No Discrimination 4 3.6 Recognition and Rights of Stewards 4 3.7 Bulletin Boards 5 3.8 Union Insignia 5 3.9 Right to Refuse to Cross Picket Lines 5 3.10 Time Off for Union Business 5 3.11 Union Shop Card 6 3.12 Notification of Staff Representatives 6 3.13 Shop Steward Representation 6 ARTICLE 4 - UNION SECURITY 7 4.1 4.1 Union Membership 7 5.2 Remitting of Dues and Employee Information 7 5.3 Exclusive Membership Rights 8 ARTICLE 5 - UNION DUES 7 5.2 Remitting of Dues and Employee Information 7 5.3 Exclusive Membership Rights 8 8 7.1 Process 8 ARTICLE 6 - ACQUAINTING NEW EMPLOYEES WITH AGREEMENT 8 7.1 Representation 7 5.3 Union Representation 7 9 7.2 Union Bargaining Committee 8 7.4 Technical Information		•	
3.7 Bulletin Boards 5 3.8 Union Insignia 5 3.9 Right to Refuse to Cross Picket Lines 5 3.10 Time Off for Union Business 5 3.11 Union Shop Card 6 3.12 Notification of Staff Representatives 6 3.13 Shop Steward Representation 6 ARTICLE 4 - UNION SECURITY 7 4.1 Union Membership 7 A.1 Union fo Dues 7 5.1 Deduction of Dues 7 5.2 Remitting of Dues and Employee Information 7 5.3 Exclusive Membership Rights 8 ARTICLE 6 - ACQUAINTING NEW EMPLOYEES WITH AGREEMENT 8 6.1 Process 8 ARTICLE 7 - UNION/EMPLOYER RELATIONS 8 7.1 Representation 8 7.2 Union Ragaining Committee 8 7.3 Union Representatives 8 7.4 Technical Information 9 7.5 Joint Union/Employer Committee 9 7.5 Joint Union/Employer Committee	3.5		
3.8 Union Insignia 5 3.9 Right to Refuse to Cross Picket Lines 5 3.10 Time Off for Union Business 5 3.11 Union Shop Card 6 3.12 Notification of Staff Representatives 6 3.13 Shop Steward Representation 6 ARTICLE 4 - UNION SECURITY. 7 4.1 Union Membership 7 4.1 Union Membership 7 5.1 Deduction of Dues 7 5.2 Remitting of Dues and Employee Information 7 5.3 Exclusive Membership Rights 8 ARTICLE 6 - ACQUAINTING NEW EMPLOYEES WITH AGREEMENT 8 6.1 Process 8 7.1 Representation 8 7.2 Union Bargaining Committee 8 7.3 Union Representatives 8 7.4 Technical Information 9 7.5 Joint Union/Employer Committee 9 7.4 Technical Information 9 7.5 Joint Union/Employer Committee 9 7.4 Technical Infor	3.6	Recognition and Rights of Stewards	4
3.9 Right to Refuse to Cross Picket Lines 5 3.10 Time Off for Union Business 5 3.11 Union Shop Card 6 3.12 Notification of Staff Representatives 6 3.13 Shop Steward Representation 6 ARTICLE 4 - UNION SECURITY 7 4.1 Union Membership 7 ARTICLE 5 - UNION DUES 7 5.2 Remitting of Dues and Employee Information 7 5.3 Exclusive Membership Rights 8 ARTICLE 6 - ACQUAINTING NEW EMPLOYEES WITH AGREEMENT 8 6.1 Process 8 7.1 Representation 8 7.2 Union Bargaining Committee 8 7.3 Joint Union/Employer Committee 9 7.4 Technical Information 9 7.5 Joint Union/Employer Committee 9 7.4 Technical Information 9 7.5 Joint Union/Employer Committee 9 7.4 Technical Information 9 7.5 Joint Union/Employer Committee 9 7.6	3.7	Bulletin Boards	5
3.10 Time Off for Union Business 5 3.11 Union Shop Card 6 3.12 Notification of Staff Representatives 6 3.13 Shop Steward Representation 6 ARTICLE 4 - UNION SECURITY 7 4.1 Union Membership 7 ARTICLE 5 - UNION DUES 7 5.1 Deduction of Dues 7 5.2 Remitting of Dues and Employee Information 7 5.3 Exclusive Membership Rights 8 ARTICLE 6 - ACQUAINTING NEW EMPLOYEES WITH AGREEMENT 8 6.1 Process 8 7.1 Representation 8 7.1 Representation 8 7.2 Union Bargaining Committee 8 7.3 Union Representatives 8 7.4 Technical Information 9 7.5 Joint Union/Employer Committee 9 7.5 Joint Union/Employer Committee 9 7.5 Joint Union/Employer Committee 9 8.1 Contents 10 8.2 Right to Review 10	3.8	Union Insignia	5
3.11 Union Shop Card 6 3.12 Notification of Staff Representatives 6 3.13 Shop Steward Representation 6 ARTICLE 4 - UNION SECURITY 7 4.1 Union Membership 7 ARTICLE 5 - UNION DUES 7 5.1 Deduction of Dues 7 5.2 Remitting of Dues and Employee Information 7 5.3 Exclusive Membership Rights 8 ARTICLE 6 - ACQUAINTING NEW EMPLOYEES WITH AGREEMENT 8 6.1 Process 8 ARTICLE 7 - UNION/EMPLOYER RELATIONS 8 7.1 Representation 8 7.2 Union Bargaining Committee 8 7.3 Union Representatives 8 7.4 Technical Information 9 7.5 Joint Union/Employer Committee 9 ARTICLE 8 - PERSONNEL FILE 10 8.1 Contents 10 8.2 Right to Review 10 9.1 Grievance 10 9.2 Informal Discussion - Optional Stage 10	3.9	Right to Refuse to Cross Picket Lines	5
3.12 Notification of Staff Representatives 6 3.13 Shop Steward Representation 6 ARTICLE 4 - UNION SECURITY 7 4.1 Union Membership 7 ARTICLE 5 - UNION DUES 7 5.1 Deduction of Dues and Employee Information 7 5.2 Remitting of Dues and Employee Information 7 5.3 Exclusive Membership Rights 8 ARTICLE 6 - ACQUAINTING NEW EMPLOYEES WITH AGREEMENT 8 6.1 Process 8 7.1 Representation 8 7.2 Union Bargaining Committee 8 7.3 Union Representatives 8 7.4 Technical Information 9 7.5 Joint Union/Employer Committee 9 ARTICLE 8 - PERSONNEL FILE 10 8.1 Contents 10 8.2 Right to Review 10 8.3 Contents 10 8.4 Contents 10 8.2 Right to Review 10 8.3 Contents 10 8.4 Con	3.10	Time Off for Union Business	5
3.13 Shop Steward Representation 6 ARTICLE 4 - UNION SECURITY 7 4.1 Union Membership 7 ARTICLE 5 - UNION DUES 7 5.1 Deduction of Dues 7 5.2 Remitting of Dues and Employee Information 7 5.3 Exclusive Membership Rights 8 ARTICLE 6 - ACQUAINTING NEW EMPLOYEES WITH AGREEMENT 8 6.1 Process 8 7.1 Representation 8 7.2 Union Bargaining Committee 8 7.3 Union Representatives 8 7.4 Technical Information 9 7.5 Joint Union/Employer Committee 9 ARTICLE 8 - PERSONNEL FILE 10 8.1 Contents 10 8.2 Right to Review 10 8.3 Contents 10 8.4 Contents 10 8.1 Contents 10 8.2 Right to Review 10 8.3 Ontents 10 8.4 Contents 10 <	3.11	•	
ARTICLE 4 - UNION SECURITY	3.12	Notification of Staff Representatives	6
4.1 Union Membership	3.13	Shop Steward Representation	6
ARTICLE 5 - UNION DUES 7 5.1 Deduction of Dues 7 5.2 Remitting of Dues and Employee Information 7 5.3 Exclusive Membership Rights 8 ARTICLE 6 - ACQUAINTING NEW EMPLOYEES WITH AGREEMENT 8 6.1 Process 8 ARTICLE 7 - UNION/EMPLOYER RELATIONS 8 7.1 Representation 8 7.2 Union Bargaining Committee 8 7.3 Union Representatives 8 7.4 Technical Information 9 7.5 Joint Union/Employer Committee 9 ARTICLE 8 - PERSONNEL FILE 10 8.1 Contents 10 8.2 Right to Review 10 9.1 Grievance 10 9.1 Grievance 10 9.2 Informal Discussion - Optional Stage 10	ARTICLE 4 - U	JNION SECURITY	7
5.1 Deduction of Dues 7 5.2 Remitting of Dues and Employee Information 7 5.3 Exclusive Membership Rights 8 ARTICLE 6 - ACQUAINTING NEW EMPLOYEES WITH AGREEMENT 8 6.1 Process 8 ARTICLE 7 - UNION/EMPLOYER RELATIONS 8 7.1 Representation 8 7.2 Union Bargaining Committee 8 7.3 Union Representatives 8 7.4 Technical Information 9 7.5 Joint Union/Employer Committee 9 ARTICLE 8 - PERSONNEL FILE 10 8.1 Contents 10 8.2 Right to Review 10 9.1 Grievance 10 9.1 Grievance 10 9.2 Informal Discussion - Optional Stage 10	4.1	Union Membership	7
5.2 Remitting of Dues and Employee Information 7 5.3 Exclusive Membership Rights 8 ARTICLE 6 - ACQUAINTING NEW EMPLOYEES WITH AGREEMENT 8 6.1 Process 8 6.1 Process 8 7.1 Representation 8 7.2 Union Bargaining Committee 8 7.3 Union Representatives 8 7.4 Technical Information 9 7.5 Joint Union/Employer Committee 9 ARTICLE 8 - PERSONNEL FILE 10 8.1 Contents 10 8.2 Right to Review 10 9.1 Grievance 10 9.1 Grievance 10 9.2 Informal Discussion - Optional Stage 10	ARTICLE 5 - U	JNION DUES	7
5.3 Exclusive Membership Rights 8 ARTICLE 6 - ACQUAINTING NEW EMPLOYEES WITH AGREEMENT 8 6.1 Process 8 ARTICLE 7 - UNION/EMPLOYER RELATIONS 8 7.1 Representation 8 7.2 Union Bargaining Committee 8 7.3 Union Representatives 8 7.4 Technical Information 9 7.5 Joint Union/Employer Committee 9 ARTICLE 8 - PERSONNEL FILE 10 8.1 Contents 10 8.2 Right to Review 10 9.1 Grievance 10 9.2 Informal Discussion - Optional Stage 10	5.1	Deduction of Dues	7
ARTICLE 6 - ACQUAINTING NEW EMPLOYEES WITH AGREEMENT 8 6.1 Process 8 ARTICLE 7 - UNION/EMPLOYER RELATIONS 8 7.1 Representation 8 7.2 Union Bargaining Committee 8 7.3 Union Representatives 8 7.4 Technical Information 9 7.5 Joint Union/Employer Committee 9 ARTICLE 8 - PERSONNEL FILE 10 8.1 Contents 10 8.2 Right to Review 10 9.1 Grievance 10 9.2 Informal Discussion - Optional Stage 10	5.2	Remitting of Dues and Employee Information	7
6.1Process8ARTICLE 7 - UNION/EMPLOYER RELATIONS87.1Representation87.17.2Union Bargaining Committee87.37.3Union Representatives7.4Technical Information97.57.5Joint Union/Employer Committee99ARTICLE 8 - PERSONNEL FILE108.1Contents8.2Right to Review109.1Grievance9.2Informal Discussion - Optional Stage10	5.3	Exclusive Membership Rights	8
6.1Process8ARTICLE 7 - UNION/EMPLOYER RELATIONS87.1Representation87.17.2Union Bargaining Committee87.37.3Union Representatives7.4Technical Information97.57.5Joint Union/Employer Committee99ARTICLE 8 - PERSONNEL FILE108.1Contents8.2Right to Review109.1Grievance9.2Informal Discussion - Optional Stage10	ARTICLE 6 - A	ACQUAINTING NEW EMPLOYEES WITH AGREEMENT	8
7.1 Representation 8 7.2 Union Bargaining Committee 8 7.3 Union Representatives 8 7.4 Technical Information 9 7.5 Joint Union/Employer Committee 9 ARTICLE 8 - PERSONNEL FILE 10 8.1 Contents 10 8.2 Right to Review 10 9.1 Grievance 10 9.2 Informal Discussion - Optional Stage 10			
7.1 Representation 8 7.2 Union Bargaining Committee 8 7.3 Union Representatives 8 7.4 Technical Information 9 7.5 Joint Union/Employer Committee 9 ARTICLE 8 - PERSONNEL FILE 10 8.1 Contents 10 8.2 Right to Review 10 9.1 Grievance 10 9.2 Informal Discussion - Optional Stage 10	ΔΑΤΙCΙ Ε 7 - Ι	INION/EMPLOYER RELATIONS	8
7.2Union Bargaining Committee87.3Union Representatives87.4Technical Information97.5Joint Union/Employer Committee9 ARTICLE 8 - PERSONNEL FILE 108.1Contents108.2Right to Review10 ARTICLE 9 - GRIEVANCE PROCEDURE 109.1Grievance109.2Informal Discussion - Optional Stage10			-
7.3 Union Representatives 8 7.4 Technical Information 9 7.5 Joint Union/Employer Committee 9 ARTICLE 8 - PERSONNEL FILE 10 8.1 Contents 10 8.2 Right to Review 10 ARTICLE 9 - GRIEVANCE PROCEDURE 10 9.1 Grievance 10 9.2 Informal Discussion - Optional Stage 10		•	
7.4 Technical Information			
ARTICLE 8 - PERSONNEL FILE	7.4	•	
8.1 Contents 10 8.2 Right to Review 10 ARTICLE 9 - GRIEVANCE PROCEDURE 10 9.1 Grievance 10 9.2 Informal Discussion - Optional Stage 10	7.5	Joint Union/Employer Committee	9
8.1 Contents 10 8.2 Right to Review 10 ARTICLE 9 - GRIEVANCE PROCEDURE 10 9.1 Grievance 10 9.2 Informal Discussion - Optional Stage 10	ARTICLE 8 - F	PERSONNEL FILE	0
8.2Right to Review10ARTICLE 9 - GRIEVANCE PROCEDURE109.1Grievance109.2Informal Discussion - Optional Stage10			
ARTICLE 9 - GRIEVANCE PROCEDURE 10 9.1 Grievance 10 9.2 Informal Discussion - Optional Stage 10	_		
9.1 Grievance	ΔΑΤΙΓΙΕΟ.		
9.2 Informal Discussion - Optional Stage10			

	9.4	Time Limits to Present Grievance at Step 2	11
	9.5	Step 2 - Procedure	11
	9.6	Time Limit to Reply at Step 2	11
	9.7	Step 3 - Procedure	11
	9.8	Time Limit to Reply at Step 3	11
	9.9	Failure to Act	11
	9.10	Time Limit to Submit to Arbitration	12
	9.11	Amending of Time Limits	12
	9.12	Dismissal or Suspension Grievance	12
	9.13	Deviation From Grievance Procedure	12
	9.14	Policy Grievance	12
	9.15	Technical Objections to Grievance	12
	9.16	Troubleshooter Provision	12
ARTICL	.E 10 - AF	RBITRATION	13
	10.1	Notification	13
	10.2	Arbitration Referral	13
	10.3	Appointment of the Arbitrator	13
	10.4	List of Arbitrators	13
	10.5	Arbitration Procedure	13
	10.6	Decision of Arbitrator	14
	10.7	Disagreement of Decision	14
	10.8	Expenses of Arbitrator	14
	10.9	Amending Time Limits	14
	10.10	Witnesses	14
	10.11	Expedited Arbitration	14
	10.12	Troubleshooter Provision	15
ARTICL	E 11 - DI	SCIPLINARY ACTION	15
	11.1	Procedure	15
	11.2	Right to Have Steward Present	15
	11.3	Burden of Proof	
	11.4	Letter of Reprimand	16
	11.5	Suspension Without Pay	16
	11.6	Dismissal	16
	11.7	Dismissal and Suspension Grievance	16
	11.8	Unjust Suspension or Discharge	16
ARTICL	E 12 - SE	NIORITY	17
	12.1	Seniority Defined	17
	12.2	Seniority List	17
	12.3	Loss of Seniority	
	12.4	Re-Employment	
	12.5	Bridging of Service	
	12.6	Seniority on Demotion	
ARTICL	E 13 - LA	YOFF AND RECALL	18
	13.1	Notice to Union	18
	13.2	Layoff and Recall	18
	13.3	Layoff Notice	19

	13.4	Layoff Options	.20
	13.5	Salary Protection	.20
	13.6	Retraining and Adjustment Period	.20
	13.7	Recall	.20
	13.8	Receipt of Due Monies	.21
	13.9	Termination of Benefits	.21
	13.10	Notice to Employee	.21
	13.11	Volunteers	
ARTICL	E 14 - SE	VERANCE PAY	.22
	14.1	Entitlement	.22
ARTICL	E 15 - HO	DURS OF WORK	
	15.1	Hours of Work	.22
	15.2	Afternoon Shift	.23
	15.3	Flex Days and Compressed Time	.23
	15.4	Shift Adjustments	.25
	15.5	Coffee Breaks	.25
	15.6	Meal Breaks	.25
	15.7	Hours of Work Record	.25
	15.8	Making Up Paid but Unworked Hours	.26
	15.9	Office Closure	.26
	15.10	Continuous Part-Time Work	.26
	15.11	Hours of Work - Work Schedules	.27
ARTICL	E 16 - O\	/ERTIME	. 27
	16.1	Definitions	.27
	16.2	Overtime Entitlement	.28
	16.3	Recording of Overtime	.28
	16.4	Sharing of Overtime	.28
	16.5	Overtime Compensation	.28
	16.6	Overtime Meal Allowance	.28
	16.7	Overtime Due to Training New Employees	.29
ARTICL	E 17 - ST	ATUTORY HOLIDAYS	. 29
	17.1	Paid Holidays	.29
	17.2	Holidays Falling on a Saturday or Sunday	.29
	17.3	Holiday Coinciding with a Day of Vacation	
ARTICL	E 18 - Al	NUAL VACATION	. 30
	18.1	Earning of Annual Vacation	.30
	18.2	Vacation - Entitlement	.30
	18.3	Selection of Vacation Dates	.30
	18.4	New Employees	.31
	18.5	Vacation Pay	
	18.6	Approved Leave of Absence With Pay During Vacations	
	18.7	Callback From Vacation	
	18.8	Vacation Carryover	
	18.9	Vacation Accrual	
	18.10	Loss of Accrual	
	18.11	Payout of Accrued Vacation Leave	
		•	

	18.12	Repayment of Vacation Credits Used but Unearned	32
ARTICI	LE 19 - SI	CK LEAVE	. 33
	19.1	Sick Leave Entitlement	33
	19.2	Use of Paid Sick Leave	33
	19.3	Unpaid Sick Leave	33
	19.4	Loss of Accrual	34
	19.5	Advising Employer	35
	19.6	Deduction of Sick Leave	35
	19.7	Repayment of Sick Leave Credits Used But Unearned	35
	19.8	Extended Absence for Medical Reasons	
	19.9	Scheduling of Medical/Dental Appointments	35
	19.10	Employer Share of Benefits	
ARTICI	LE 20 - O	THER LEAVE	. 36
	20.1	Bereavement Leave	
	20.2	Special Leave	
	20.3	Other Observances	
	20.3	Compassionate Leave	
	20.5	Critical Illness or Injury Leave	
	20.6	Compassionate Care Leave	
	20.7	Leave Respecting the Disappearance of a Child	
	20.8	Leave Respecting the Death of a Child	
	20.9	Family Responsibility Leave	
	20.10	Leave for Domestic or Sexual Violence	
	20.10	Donor Leave	
	20.12	Public Duties	
	20.12	Leave for Court Appearances	
	20.13	Elections	
	20.14	Absence Without Leave (Abandonment of Position)	
	20.15	Leave Without Pay	
	20.10	Educational Leave	
	20.17	Canadian Armed Forces	
	20.18	Planned Leave	
ARTICI	LE 21 - M 21.1	ATERNITY AND PARENTAL LEAVE	
	21.1	Eligibility and Notification	
		Maternity Leave Parental Leave	
	21.3		
	21.4	Return from Leave	
	21.5	Health and Welfare Benefits	
	21.6	Maternity and Parental Leave Allowances	
	21.7	Additional Benefits	
	21.8	Seniority Accrual	43
ARTICI		AFETY AND HEALTH	
	22.1	Joint Occupational Health and Safety Committee	
	22.2	Advising Employer	
	22.3	Video Display Terminals	
	22.4	Equipment Standards	44

	22.5	Personal Safety	44
	22.6	Unsafe Work Conditions	44
	22.7	Injury Pay Provision	45
	22.8	Transportation of Accident Victims	45
	22.9	Supply and Maintenance of Equipment	45
	22.10	Smoking Policy	45
	22.11	Workplace Violence	45
	22.12	Occupational First Aid	45
ARTICL	.E 23 - H/	ARASSMENT AND BULLYING	46
	23.1	Harassment and Bullying-Free Worksite	46
	23.2	Prohibition of Harassment and Bullying	46
	23.3	Disciplinary Action	46
	23.4	Limitation of Definition	46
	23.5	Response to Harassment or Bullying	47
	23.6	Confidential Investigation	47
	23.7	Sexual Harassment	47
	23.8	Complaint Procedure for Sexual Harassment	47
ARTICL	.E 24 - TE	CHNOLOGICAL CHANGE	48
	24.1	Process	48
ARTICL	.E 25 - RI	ECRUITMENT AND SELECTION	48
	25.1	Recruitment and Selection	48
	25.2	Postings	49
	25.3	Promotion While on Probation	50
	25.4	Accuracy of Information	50
	25.5	Job Sharing	50
	25.6	Job Exchanges	52
	25.7	Lateral Transfer Policy	52
	25.8	Unable to Place Into Vacancy Because of Operational Requirements	53
ARTICL	.E 26 - PF	ROMOTION	53
	26.1	Interview Expenses	53
	26.2	Transfers	53
	26.3	Probation on Promotion	53
	26.4	Notification to Employee and Union	54
	26.5	Salary on Promotion	54
	26.6	Right to Grieve	54
	26.7	Acting Prior to Promotion	54
ARTICL	.E 27 - PF	ROBATION	55
	27.1	Probationary Period	55
	27.2	Right to Grieve	55
ARTICL	.E 28 - PE	RFORMANCE APPRAISAL	55
	28.1	Probationary Employees	
	28.2	Non-Probationary Employees	56
	28.3	Procedure	
			56

	ROFESSIONAL CONDUCT/CONFIDENTIALITY	
29.1	Conflict of Interest	56
29.2	Personal Use of Employer's Resources	57
29.3	Oath of Confidentiality	57
29.4	Inappropriate Use of Information	57
29.5	Due Care	57
ARTICLE 30 - P/	AYMENT OF WAGES AND ALLOWANCES	
30.1	Equal Pay	
30.2	Paydays	57
30.3	Rates of Pay	57
30.4	Substitution Pay	58
30.5	Pay on Temporary Assignment	58
30.6	Salary Protection	59
30.7	Vehicle Allowance	59
30.8	Meal Allowance	59
30.9	Accommodation and Other Expenses	59
30.10	Personal Charge Accounts	60
30.11	Travel Advance	60
30.12	Expense Form	60
30.13	Travel Accounts	60
30.14	Transportation for Employees	60
30.15	Training Expenses	60
30.16	Relocation Expenses	60
30.17	Training Allowance	61
ARTICLE 31 - H	EALTH AND WELFARE BENEFITS	62
31.1	Benefits Committee	62
31.2	Commencement of Benefits	62
31.3		
	Group Plan	62
31.4	Group Plan Benefits Review	
	•	63
31.4	Benefits Review	63 63
31.4 31.5	Benefits Review Cost of Premiums	63 63 63
31.4 31.5 31.6	Benefits Review Cost of Premiums Basic Medical Insurance	
31.4 31.5 31.6 31.7	Benefits Review Cost of Premiums Basic Medical Insurance Group Plan Coverage	
31.4 31.5 31.6 31.7 31.8	Benefits Review Cost of Premiums Basic Medical Insurance Group Plan Coverage Medical Examination	
31.4 31.5 31.6 31.7 31.8 31.9	Benefits Review Cost of Premiums Basic Medical Insurance Group Plan Coverage Medical Examination Legislative Changes	
31.4 31.5 31.6 31.7 31.8 31.9 31.10	Benefits Review Cost of Premiums Basic Medical Insurance Group Plan Coverage Medical Examination Legislative Changes Health and Welfare Plan	
31.4 31.5 31.6 31.7 31.8 31.9 31.10 31.11	Benefits Review Cost of Premiums Basic Medical Insurance Group Plan Coverage Medical Examination Legislative Changes Health and Welfare Plan Coverage on Leave of Absence Without Pay	
31.4 31.5 31.6 31.7 31.8 31.9 31.10 31.11 31.12	Benefits Review Cost of Premiums Basic Medical Insurance Group Plan Coverage Medical Examination Legislative Changes Health and Welfare Plan Coverage on Leave of Absence Without Pay Coverage Sick Leave.	
31.4 31.5 31.6 31.7 31.8 31.9 31.10 31.11 31.12 31.13	Benefits Review Cost of Premiums Basic Medical Insurance Group Plan Coverage Medical Examination Legislative Changes Health and Welfare Plan Coverage on Leave of Absence Without Pay Coverage Sick Leave Benefit Allowance	63 63 63 63 63 63 63 63 63 63 64 64 64 64 64
31.4 31.5 31.6 31.7 31.8 31.9 31.10 31.11 31.12 31.13 31.14 31.15	Benefits Review Cost of Premiums Basic Medical Insurance Group Plan Coverage Medical Examination Legislative Changes Health and Welfare Plan Coverage on Leave of Absence Without Pay Coverage Sick Leave Benefit Allowance Retention Accounting	63 63 63 63 63 63 63 63 63 63 64 64 64 64 64 64 64
31.4 31.5 31.6 31.7 31.8 31.9 31.10 31.11 31.12 31.13 31.14 31.15 ARTICLE 32 - PI	Benefits Review Cost of Premiums Basic Medical Insurance Group Plan Coverage Medical Examination Legislative Changes Health and Welfare Plan Coverage on Leave of Absence Without Pay Coverage Sick Leave Benefit Allowance Retention Accounting Employees Aged 65 and Over	63 63 63 63 63 63 63 63 63 64 64 64 64 64 64 64 64 64 64 64 64 64
31.4 31.5 31.6 31.7 31.8 31.9 31.10 31.11 31.12 31.13 31.14 31.15 ARTICLE 32 - PI ARTICLE 33 - G	Benefits Review Cost of Premiums Basic Medical Insurance Group Plan Coverage Medical Examination Legislative Changes Health and Welfare Plan Coverage on Leave of Absence Without Pay Coverage Sick Leave Benefit Allowance Retention Accounting Employees Aged 65 and Over ENSION PLAN	
31.4 31.5 31.6 31.7 31.8 31.9 31.10 31.11 31.12 31.13 31.14 31.15 ARTICLE 32 - PI ARTICLE 33 - G 33.1	Benefits Review Cost of Premiums Basic Medical Insurance Group Plan Coverage Medical Examination Legislative Changes Health and Welfare Plan Coverage on Leave of Absence Without Pay Coverage Sick Leave Benefit Allowance Retention Accounting Employees Aged 65 and Over ENSION PLAN Parking	
31.4 31.5 31.6 31.7 31.8 31.9 31.10 31.11 31.12 31.13 31.14 31.15 ARTICLE 32 - PI ARTICLE 33 - G	Benefits Review Cost of Premiums Basic Medical Insurance Group Plan Coverage Medical Examination Legislative Changes Health and Welfare Plan Coverage on Leave of Absence Without Pay Coverage Sick Leave Benefit Allowance Retention Accounting Employees Aged 65 and Over ENSION PLAN	

33.5	Employee Workload	65
ARTICLE 34 -	CONTRACTING OUT	66
ARTICLE 35 -	WORK EXPERIENCE PROGRAMS	
35.1	Work Experience Programs	66
ARTICLE 36 -	CAREER DEVELOPMENT	66
36.1	Purpose	66
36.2	Scope and Procedures	66
ARTICLE 37 -	JOB EVALUATION	67
37.1	Consultation	67
37.2	Implementation Procedure	67
37.3	Documents for Committee	67
37.4	Rates of Pay	67
37.5	Job Evaluation Consultants	67
37.6	Classification Review	67
37.7	Salary Protection	68
37.8	Employee Request for a Job Description	68
37.9	Labour Market Review	68
ARTICLE 38 -	TEMPORARY AND CASUAL EMPLOYEES	
38.1		
38.2	Remuneration	69
38.3	Benefits	69
38.4	Probation	69
38.5		
38.6	Performance Review	69
38.7	Termination	70
ARTICLE 39 -	ACTING EMPLOYEES	70
39.1	Definition	70
39.2	Remuneration	70
39.3	Benefits and Entitlements	70
39.4	Probation	70
39.5	Seniority	70
39.6	Right to Position	70
ARTICLE 40 -	TERM OF AGREEMENT	70
40.1	Duration	70
40.2	8	
40.3	5 5	
40.4	5 5	
40.5	6	
40.6	5	
40.7	Essential Services	71
APPENDIX I -	Salary Grid	73
APPENDIX II	- Geographic Location	80
APPENDIX III	- Low Wage Redress (LWR) - Annual Wage Gap	

MEMORANDUM OF AGREEMENT #1 - Personal Growth Allowance	. 80
MEMORANDUM OF AGREEMENT #2 - Cost of Living Adjustments (COLA)	. 81
MEMORANDUM OF AGREEMENT #3 - Low Wage Redress (LWR)	. 81
MEMORANDUM OF AGREEMENT #4 - Indemnity	. 81
MEMORANDUM OF AGREEMENT #5 - Job Evaluation Process	. 82
MEMORANDUM OF AGREEMENT #6 - Compressed Time - Payment Services	. 84
MEMORANDUM OF AGREEMENT #7 - Workplace Mental Health	. 85
MEMORANDUM OF AGREEMENT #8 - Telework	. 85
MEMORANDUM OF UNDERSTANDING - Clause 31.9 (Legislative Changes)	. 85
LETTER OF UNDERSTANDING #1 - Service Improvement Fund - Hard to Recruit or Retain Staff Positions	. 85
LETTER OF UNDERSTANDING #2 - Public Sector General Wage Increase	. 86
LETTER OF UNDERSTANDING #3 - Joint Training	. 87
LETTER OF UNDERSTANDING #4 - Low Wage Redress	. 87
LETTER OF UNDERSTANDING #5	. 87

For the purpose of this agreement:

"Acting Appointment" - an acting appointment exists when a continuous employee is temporarily performing the duties of another position. The following clauses apply specifically to acting appointments: 25.1 (Recruitment and Selection), 25.2 (Postings), 26.5 (Salary on Promotion), 26.7 (Acting Prior to Promotion), 30.5 (Substitution Pay).

"*Bargaining Unit*" - is the unit for collective bargaining for which the B.C. General Employees' Union was certified by the Industrial Relations Council of BC on June 4, 1990.

"Calendar Year" - is the period January 1st to December 31st.

"*Child*" - wherever the word "*child*" is used in this agreement, it shall be deemed to include a child in care of a director designated under the *Child, Family and Community Service Act* or a child of a spouse, including the child of a common-law spouse, or a child to whom the employee stands in loco parentis.

"Classification" - an employee's classification is their job title.

"Component" - means a classification as defined by those listed Appendix 1 - Salary Grid

"*Day of Rest*" - in relation to an employee, means a day other than a holiday on which an employee is not ordinarily required to perform the duties of the position. This does not include employees on a leave of absence.

"*Demotion*" - means a change from an employee's position to one with a lower maximum salary.

"Employee" - means a member of the bargaining unit and includes:

(a) Continuous employee means an employee who is employed for work which is of a continuous full-time or continuous part-time nature;

(b) Temporary employee is an employee hired from outside the organization for a specified or indeterminate period between 60 and 250 continuous days of work;

(c) Casual employee is one who is hired for short-term specific tasks of less than 60 continuous working days duration.

"*Fiscal Year*" - means from April 1st of any year to March 31st of the following year.

"*Lateral Transfer*" - refers to the movement of an employee from one position to another which does not constitute a demotion or promotion.

"*Layoff*" - is a temporary or permanent cessation of employment as per Article 13 (Layoff and Recall) and 24 (Technological Change).

"Leave of Absence with Pay" - means to be absent from duty with permission and with pay.

"*Leave of Absence without Pay*" - means to be absent from duty with permission but without pay.

"*Operational Requirements*" - means the areas that the Employer must take into consideration to ensure that it meets its obligations in an efficient and effective manner. These include such things as adequate staff coverage over the entire working day. In making decisions with reference to operational requirements, the Employer will consider the following:

- to what extent will it have a visible impact on client service;
- what kind of additional workload will it likely place on other staff and managers;
- does it affect our contractual obligations to the province in letter or spirit;
- could it negatively affect how the organization is seen by the province?

When an employee request is declined due to operational requirements, the Employer will tell the employee why the request could not be accommodated. The Employer's decision is final.

"*Pay*" - means rate of compensation for the job.

"Peak Periods" - means July, August, December and the first 10 days in January.

"*Promotion*" - means a change from an employee's position to one with a higher maximum salary level.

"*Recall*" - is a return to work subsequent to a layoff.

"*Resignation*" - means voluntary notice by the employee that the employee is terminating service on the date specified.

"*Rest Period*" - is a paid break.

"Spouse" - includes a common-law spouse and a partner in a same sex relationship.

"*Temporary Position*" - is created where there is either a time limited new position or a temporary vacancy in an existing position.

"*Termination*" - is the separation of an employee from the Employer for cause pursuant to Article 11 (Disciplinary Action) of this agreement.

"*Transfer*" - refers to the relocation of an employee from one geographic work location to another.

"*Workday*" - is a period of 24 consecutive hours commencing with the starting time of any shift. For the purpose of calculating compensatory overtime rates only, the time worked prior to, but adjoining to a shift, shall be deemed as time worked after a shift.

"*Work Unit*" - is defined as a regional office as a whole, not the components within the office or other offices that are in the same space.

ARTICLE 1 - INTRODUCTION

1.1 Intent of Agreement

(a) This collective agreement constitutes the terms and conditions of employment for bargaining unit employees of the BC Family Maintenance Agency (hereafter called the Employer). These terms and conditions apply, as written, to all non-excluded employees of the Employer who are not in another bargaining unit and remain in effect until altered by way of agreement between the Employer and the bargaining agent for the employees at the end of the term specified in Article 40 (Term of Agreement).

(b) Every employee shall be given a copy of the collective agreement at the time an offer of employment is made. Any laws of Canada and of British Columbia which set standards, wages and practices that are legally binding on the Employer, are deemed to be in effect even though they may not be outlined in the collective agreement. If there is any conflict between those laws and this collective agreement, the laws shall prevail.

(c) The parties to this agreement share a desire to improve the quality of the services provided by BC Family Maintenance Agency Ltd. (BCFMA) and the Family Maintenance Enforcement Program or any other services undertaken by the Employer.

(d) The parties recognize the benefit to be derived from a work environment free from harassment where the conduct and/or language of all employees meet the acceptable social standard of the workplace. In keeping with the spirit of this article:

(1) the Employer will make every reasonable effort to maintain such an environment; and

(2) all employees have an obligation to advise the Employer if there is a problem that affects the work or environment of the office.

1.2 Future Legislation

In the event that any future legislation renders null and void or materially alters any provision of this agreement, the remaining provisions shall remain in effect for the term of the agreement, and the parties hereto shall negotiate a mutually agreeable provision to be substituted for the provision so rendered null and void or materially altered.

1.3 Conflict with Regulations

In the event that there is a conflict between an article of this agreement and any regulation made by the Employer, this agreement shall take precedence over the said regulation.

1.4 Human Rights Code

The parties hereto subscribe to the principles of the *Human Rights Code* of British Columbia.

ARTICLE 2 - STATEMENT OF MANAGEMENT RIGHTS

2.1 Management Rights

The Union acknowledges that the organization of the work, and the management and directing of employees in the bargaining unit is retained by the Employer, except as this agreement otherwise specifies.

2.2 Exercise of Management Rights

In exercising its management rights, the Employer recognizes that it has a responsibility to involve employees in the organization and development of tasks, policies, procedures and standards. In practice, this is achieved through various staff committees (training, computer, forms, etc.), the circulation of policy and procedure drafts, special task definition workshops, etc.

ARTICLE 3 - UNION RECOGNITION AND RIGHTS

3.1 Bargaining Unit Defined

The bargaining unit shall comprise all employees included in the certificate issued by the Industrial Relations Council on June 4, 1990, except those in positions which it was mutually agreed would be excluded by the parties as managerial and/or confidential.

3.2 Bargaining Unit Recognition

The Employer recognizes the B.C. General Employees' Union as the exclusive bargaining agent for all employees to whom the certification, issued by the Industrial Relations Council on June 4, 1990, applies.

3.3 Correspondence

The Employer agrees that all correspondence between the Employer and the Union related to matters covered in this agreement shall be to the President of the Union or designate. The Employer agrees that a copy of any correspondence between the Employer or Employer's official and any employee in the bargaining unit covered by this agreement, pertaining to the interpretation or application of any clause in this agreement, shall be forwarded to the President of the Union or designate.

3.4 No Other Agreement

No employee covered by this agreement shall be required to make a written or verbal agreement with the Employer or its representatives which may conflict with the terms of this agreement.

3.5 No Discrimination

Subject to other articles contained in this agreement the Employer agrees that there shall be no discrimination, interference, restriction or coercion exercised or practised with respect to any employee in the matter of hiring, rates of pay, training, upgrading, promotion, transfer, layoff, recall, discipline, discharge or otherwise for reasons of creed, place of residence, race, colour, ancestry, place of origin, political belief, religion, marital status, family status, physical or mental disability, sex, sexual orientation, gender identity or expression, or age of that person or because that person has been convicted of a criminal or summary conviction offence that is unrelated to the employment or to the intended employment of that person, or for reason of membership or activity in the Union, or for any other reason.

3.6 Recognition and Rights of Stewards

The Employer recognizes the Union's right to select stewards to represent employees. The Employer and the Union will agree on the number of stewards taking into account both operation and geographical considerations. The Union agrees to provide the Employer with a list of the employees designated as stewards, and alternates.

The steward or alternate, shall obtain the permission of the immediate supervisor before leaving work to perform those duties as steward which are of an urgent nature. Such permission shall not be unreasonably withheld. On resuming normal duties, the steward shall notify the supervisor. Review of non-urgent complaints and other communications shall not be done on the Employer's time.

The duties of stewards shall include:

- (a) review of complaints of an urgent nature;
- (b) processing of grievances and assisting any employee;
- (c) supervision of ballot boxes and other related functions during ratification votes;

(d) carrying out duties within the realm of specifically designated safety responsibilities, these being recognized as complaints of an urgent nature which require immediate attention; and

(e) attending meetings called by the Employer.

3.7 Bulletin Boards

The Employer shall provide bulletin board facilities for the exclusive use of the Union, the sites to be determined by mutual agreement. The use of such bulletin board facilities shall be restricted to the business affairs of the Union.

3.8 Union Insignia

A union member shall have the right to wear or display the recognized insignia of the Union.

3.9 Right to Refuse to Cross Picket Lines

(a) All employees covered by this agreement shall have the right to refuse to cross a lawful picket line arising out of a dispute as defined in the *Labour Relations Code* of British Columbia. An employee failing to report for duty shall be considered to be absent without pay. The Union will make every reasonable effort to ensure that its members can gain access to the workplace in the event that a BCFMA employee's entry might be denied because of a labour dispute between another employer and its employees.

(b) Failure to cross a lawful picket line encountered in carrying out the Employer's business shall not be considered a violation of this agreement nor shall it be grounds for disciplinary action.

(c) The Employer agrees that it shall not request or require or direct employees to perform work in progress that is currently being carried out by those on a strike or locked out. This does not restrict the Employer from replacing a product or service from a supplier whose employees are on strike with another supplier.

3.10 Time Off for Union Business

(a) Without Pay

Leave of absence without pay but without loss of seniority shall be granted:

(1) to an elected or appointed representative of the Union to attend conventions of the Union and bodies to which the Union is affiliated;

(2) for elected or appointed representatives of the Union to attend to union business, which requires them to leave their premises of employment;

(3) for employees who are representatives of the Union on the Bargaining Committee for BCFMA employees to attend meetings of the Bargaining Committee, and to carry on negotiations with the Employer. The Employer will pay 50% of the facility expenses for joint bargaining sessions and will pay the usual Employer's share of the benefits for employees who are on the Bargaining Committee while they are absent from work for bargaining purposes.

(4) the maximum leave entitlement as per Clause 3.10(a)(1) and (2) combined shall be 30 days per fiscal year: additional time shall be granted by the Employer and shall not be unreasonably denied, except where operational requirements, as determined by the Employer, do not permit.

(5) for employees selected for a full-time position with the Union or any body to which the Union is affiliated for a period of one year. The leave shall be renewed upon request, subject to operational requirements, and such request shall not be unreasonably withheld;

(6) for an employee elected to the position of President or Treasurer of the B.C. General Employees' Union for a period of three years and the leave shall be renewed upon request.

(b) With Pay

Leave of absence with pay and without loss of seniority will be granted:

(1) to stewards or their alternates, to perform their duties pursuant to Clause 3.6 (Recognition and Rights of Stewards);

- (2) to employees called to appear as witnesses before an arbitration board.
- (3) to employees to attend joint union-employer committee meetings.
- (c) Local Union Meetings

The Employer agrees to allow employees to leave work at 4:00 p.m. four times a year for the purpose of attending union meetings without loss of pay. The Union agrees to notify the Employer of the dates of such meetings at least two weeks prior to the meeting. Minimal staff coverage shall be maintained. If the normal shift is not concluded by the end of the union meeting, the employee shall return to work for the balance of the shift.

(d) Procedure

Employees granted leave of absence pursuant to this article shall receive their current rate of pay while on leave of absence with pay. Leave of absence granted under this article shall include sufficient travel time but no additional time or compensation may be claimed for travel time or extended workday. To facilitate the administration of clause (a) above, when leave without pay is granted, the leave shall be given with pay and the Union shall reimburse the Employer for the appropriate salary, and other benefit costs, including travel time incurred.

3.11 Union Shop Card

The Union agrees to furnish the Employer with at least one union shop card for each of the Employer's places of operations covered by this agreement, to be displayed on the premises. Such card will remain the property of the Union and shall be surrendered upon demand.

3.12 Notification of Staff Representatives

The parties recognize the value of communication on an ongoing basis, and to this end, each party will keep the other informed at all times of its staff representatives responsible for labour relations matters.

3.13 Shop Steward Representation

The Union shall be entitled to have a shop steward and alternate shop steward in the following three work locations:

- Victoria Client Offices
- Lower Mainland Client Office (Burnaby)
- Northern and Interior Client Office (Kamloops)

In the event that new offices are opened which are not part of an existing facility, each office will be entitled to its own shop steward and alternate shop steward.

ARTICLE 4 - UNION SECURITY

4.1 Union Membership

(a) All employees in the bargaining unit who, on June 4, 1990, were members of the Union or thereafter became members of the Union shall, as a condition of continued employment, maintain such membership.

(b) All employees, whose positions are in the BCGEU bargaining unit, hired on or after June 4, 1990, shall, as a condition of continued employment, become members of the Union and maintain such membership.

ARTICLE 5 - UNION DUES

5.1 Deduction of Dues

(a) The Employer shall, as a condition of employment deduct from the monthly wages or salary of each employee in the bargaining unit the amount of the regular monthly dues payable to the Union by a member of the Union.

(b) The Employer shall deduct from the monthly wages or salary of any employee who is a member of the Union any assessments levied in accordance with the Union Constitution and/or Bylaws and owing by the employee to the Union. The Union shall provide and maintain a current list of union members to the Employer.

(c) The Employer shall start to deduct union dues, as per Clause 5.2 (Remitting of Dues and Employee Information), as of the date an employee first commences work in a bargaining unit position.

(d) Deductions shall be made semi-monthly and membership dues or payments in lieu thereof shall be considered as owing in the month for which they are so deducted.

(e) The Employer shall supply each employee, without charge, a receipt for income tax purposes in the amount of the deductions paid to the Union by the employee in the previous year. Such receipt shall be provided to the employee prior to March 1st of the succeeding year.

(f) An employee shall, as a condition of continued employment, complete an authorization form providing for the deduction from an employee's wages or salary the amount of regular monthly dues payable to the Union by a member of the Union.

5.2 Remitting of Dues and Employee Information

(a) All deductions shall be remitted to the President of the Union not later than 28 days after the end of the month for which the dues were deducted.

(b) The Employer will remit union dues by electronic funds transfer (EFT). With each remittance the Employer will send an email to the address specified by the Union including: EFT date and dollar amount; employer name; pay period type (e.g.: monthly, semi-monthly, biweekly, etc.); pay period number; pay period end date; and pay period pay date.

(c) The Employer will provide to the Union with every regular dues remittance the following member information in a mutually agreed electronic format: employee ID number, last name, first name, dues amount, gross wages for the period, job/position title, appointment status (e.g. regular or casual), work location name, work location address, work phone number, and member home email known to the Employer.

(d) The Employer will provide to the Union on a quarterly basis a report of employees who have ceased employment including each employee's name and the code used in Block 16 of their Record of Employment form.

(e) Before the Employer is obligated to deduct any amount under this article, the Union must advise the Employer in writing of the amount of its regular monthly dues or assessments. The amount so advised shall continue to be the amount to be deducted until changed by further written notice to the Employer signed by the President of the Union. Upon receipt of such notice, such changed amount shall be the amount deducted.

5.3 Exclusive Membership Rights

From the date of the signing of this agreement and for its duration, no employee organization other than the Union shall be permitted to have membership dues or other monies deducted by the Employer from the pay of the employees in the bargaining unit.

ARTICLE 6 - ACQUAINTING NEW EMPLOYEES WITH AGREEMENT

6.1 Process

The Employer agrees to acquaint new employees with the fact that a collective agreement is in effect and with the conditions of employment set out in the articles dealing with Union Security and Dues Check-off. A new employee shall be advised of the name and location of the steward. The Employer agrees that the union steward will be given an opportunity to meet with each new employee within regular working hours without loss of pay, for 30 minutes sometime during the first 30 days of employment at a time mutually convenient to all parties for the purpose of acquainting the new employee with the benefits and duties of union membership and the employee's responsibilities and obligations to the Employer and the Union.

ARTICLE 7 - UNION/EMPLOYER RELATIONS

7.1 Representation

No employee or group of employees shall undertake to represent the Union at meetings with the Employer without the proper authorization of the Union. To implement this, the Union shall supply the Employer with the names of its officers and similarly the Employer shall supply the Union with a list of its supervisory or other personnel with whom the Union may be required to transact business.

7.2 Union Bargaining Committee

A union bargaining committee shall be appointed by the Union and shall consist of three members of the Union, with one representative from each of the three geographic locations, together with the President of the Union or designate. The Union shall have the right at any time to have the assistance of members of the staff of the Union when negotiating with the Employer.

The Union and the Employer recognize the mutual value of ongoing joint discussions and negotiations in matters pertaining to working conditions employment services and labour/management relations. To this end the Union Bargaining Committee and the Employer agree that in the event both parties wish to call a meeting under this clause, the meeting shall be held at a time and place fixed by mutual agreement.

7.3 Union Representatives

The Employer agrees that access to its premises will be granted as mutually agreed to members of the staff of the Union when dealing or negotiating with the Employer as well as for the purpose of

investigating and assisting in the settlement of a grievance. Members of union staff shall notify the designated supervisory official in advance of their intention and their purpose for entering and shall not interfere with the operation of the department or section concerned. In order to facilitate orderly, as well as the confidential, investigation of grievances, the Employer will make available to the union representatives or stewards, temporary use of an office or similar facility, if such is available.

7.4 Technical Information

The Employer agrees to provide to the Union such information that is available relating to employees in the bargaining unit, as may be required by the Union for collective bargaining.

7.5 Joint Union/Employer Committee

(a) A joint union/employer committee shall be formed for the purpose of developing mutually acceptable recommendations with respect to issues of particular concern to employees and/or the Employer. The Committee will meet at the request of either party up to three times per year, at a mutually acceptable date, place and time and shall consist of one union representative from each local committee pursuant to Clause 7.5(c). The Employer shall appoint an equal number of representatives to the Committee. If either party wishes to discuss the issue of workload, they shall advise the other party and they shall select the closest possible mutually acceptable date to meet. Workload shall be a standing agenda item.

Committee meetings shall be held during work hours and travel expenses shall be paid at the usual rate by the Employer. Meetings shall be chaired on a rotating basis and minutes shall be taken by the provincial office Executive Assistant or Executive Administrator, who shall be present but shall not be considered a member of the Committee. Minutes will be circulated for approval in draft form. Any disagreement with respect to the minutes shall be noted in the minutes if any representative is not in agreement with the decision recorded.

(b) Committee members will be asked to submit agenda items to the meeting Chairperson no later than two weeks prior to the meeting. The proposed agenda shall be circulated one week prior to the meeting and shall be posted on the union notice board. Minutes of committee meetings shall be posted on the union notice board within 14 days.

It is intended that the Committee have as flexible a role as possible. Its general purpose is to foster open communication, conjoint planning and mutual problem solving. It is not, however, a bargaining committee.

Upon mutual agreement, ad hoc committees or working groups may be established with a defined mandate which will report back to the Committee. Recommendations of the Committee which pertain to this agreement will be sent to bargaining principals for approval.

(c) A local union/employer committee shall be formed at each worksite to address issues of particular concern. Issues which cannot be resolved shall be referred to the Committee referenced in 7.5(a). The Local Committee shall consist of two union representatives which shall be the steward and the bargaining representative or designate. The Employer shall appoint two excluded managers, one of whom shall be a Senior Manager. The Committees shall each meet at least two times a year and more frequently where mutually agreed.

ARTICLE 8 - PERSONNEL FILE

8.1 Contents

The Employer shall maintain an official Personnel file on every employee. This file shall contain copies of any documents or notes which pertain to the hiring, promotion, job performance, pay and benefits, leave records, disciplinary action, conclusion of employment, and any other relevant documents or notes.

Any adverse documents pertaining to job performance, or reports demonstrating specific performance issues (not dealt with under Article 11 [Disciplinary Action]), excluding performance appraisals, shall be removed from the employee's file after the expiration date of 12 months from the date it was issued, provided there has not been further job performance issues.

Except as specified above or in Article 11 (Disciplinary Action), all documents shall remain on file permanently. If any disciplinary action is reversed upon review, grievance or arbitration, all reference to the reversed disciplinary action shall be removed from the employee's Personnel file.

8.2 Right to Review

The employee has the right to review their personnel file upon advising their supervisor or in their absence another excluded supervisor that they wish to do so, providing that the frequency of these requests is reasonable. An excluded supervisor shall grant the employee's request within one working day, subject to the operational availability of an excluded supervisor, and shall be present in the room while the file is being reviewed. The employee may request the supervisor in attendance to make photocopies of any document(s) on the file and this shall be done within one working day. All requests to review the file and for copies of documents shall be noted by the attending supervisor on the file's running record. The employee may authorize that a union representative be given access to their personnel file.

ARTICLE 9 - GRIEVANCE PROCEDURE

9.1 Grievance

The Employer and the Union recognize that grievances may arise concerning:

- (a) differences between the parties respecting the interpretation, application, operation, or any alleged violation of a provision of this agreement; or
- (b) the dismissal, discipline or suspension of an employee bound by this agreement.

The procedure for resolving a grievance is defined by this article.

9.2 Informal Discussion - Optional Stage

At the initial stage of any dispute, every effort should be made to resolve outstanding disputes starting first with an informal discussion with the designated supervisor. If the dispute is not resolved orally, the aggrieved employee may move to Step 1 of the grievance procedure.

9.3 Step 1 - Grievance Procedure

The parties shall acknowledge that the matter is at Step 1 of the grievance procedure. Every effort shall be made to settle the dispute through formal discussion with the designated local supervisor. The aggrieved employee shall have the right to have the steward present at such a discussion. If the dispute is not resolved as a result of such a meeting, the aggrieved employee may file a written grievance through the union steward, as per Step 2 of the grievance procedure.

9.4 Time Limits to Present Grievance at Step 2

An employee who wishes to present a grievance at Step 2 of the grievance procedure, in the manner prescribed in Clause 9.5 (Step 2 - Procedure), must do so not later than 30 days after the date:

(a) on which the employee was notified orally or in writing of action or circumstances giving rise to the grievance; and

(b) on which the employee first became aware of the action or circumstances giving rise to the grievance.

9.5 Step 2 - Procedure

(a) Subject to the time limits of Clause 9.4 (Time Limits to Present Grievance at Step 2), the employee may present a grievance at this level by:

(1) recording this grievance on the appropriate grievance form setting out the nature of the grievance and the circumstances from which it arose;

(2) stating the article or articles of the agreement infringed upon or alleged to have been violated, and the remedy or correction sought; and

(3) transmitting this grievance to the designated local supervisor through the union steward.

(b) The local supervisor shall:

(1) forward the grievance to the appropriate representative of the Employer authorized to deal with grievances at Step 2;

(2) provide the employee with a receipt stating the date on which the grievance was received.

9.6 Time Limit to Reply at Step 2

The appropriate representative designated by the Employer to handle grievances at Step 2 shall reply in writing to an employee's grievance within 14 days of receiving the grievance at Step 2.

9.7 Step 3 - Procedure

The President of the Union or designate may present a grievance at Step 3 by submitting it to the CEO:

(a) within 14 days after the decision has been conveyed by the representative designated by the Employer to handle grievances at Step 2; or

(b) within 14 days after the Employer's reply was due.

9.8 Time Limit to Reply at Step 3

The representative designated by the Employer to handle grievances at Step 3 shall reply in writing to the grievance within 14 days of receipt of the grievance at Step 3.

9.9 Failure to Act

If the President of the Union, or designate, does not present a grievance to the next higher lever within the prescribed time limits, the grievance will be deemed to have been abandoned. However, the Union shall not be deemed to have prejudiced its position on any further grievance.

9.10 Time Limit to Submit to Arbitration

Failing satisfactory settlement at Step 3, and pursuant to Article 10 (Arbitration), the President or designate, may inform the Employer of the intention to submit the dispute to arbitration within:

- (a) 14 days after the Employer's decision has been received;
- (b) 14 days after the Employer's decision is due.

9.11 Amending of Time Limits

The time limits fixed in this grievance procedure may be altered by mutual consent of the parties but the same must be in writing. Where a grievance or a reply is presented by mail or facsimile, it shall be deemed to be presented on the day on which it is postmarked or facsimile stamped and it shall be deemed to be received on the day it was delivered to the appropriate office of the Employer or the Union.

9.12 Dismissal or Suspension Grievance

In the case of a dispute arising from an employee's dismissal or suspension, the grievance shall commence at Step 3 of the grievance procedure within 14 days of the date on which the suspension occurred, or within 14 days of the employee receiving notice of dismissal or notice of suspension.

9.13 Deviation From Grievance Procedure

The Employer agrees that after a grievance has been filed by the Union, the Employer's representative will not enter into discussion or negotiation with respect to the grievance, either directly or indirectly with the aggrieved employee without the consent of the Union. Such consent shall not be unreasonably withheld when the purpose is to enable the Employer to formally investigate the grievance. In the event of such a meeting the employee shall be entitled to union representation. In the event that after having initiated a grievance through the grievance procedure, an employee endeavours to pursue the same grievance through any other channel, then the Union agrees that pursuant to this article, the grievance shall be considered to have been abandoned.

9.14 Policy Grievance

Where either party to this agreement disputes the general application, interpretation or alleged violation of this agreement, the dispute shall be discussed within 30 days of either party requesting a meeting to resolve the issue. Where no satisfactory agreement is reached, either party may submit the dispute to arbitration, as set out in Article 10 (Arbitration)

9.15 Technical Objections to Grievance

It is the intention of both parties to this agreement that no grievance shall be defeated merely because of a technical error other than time limitations in processing the grievance through the grievance procedure. To this end, an arbitration board shall have the power to allow all necessary amendments to the grievance and the power to waive formal procedural irregularities in the processing of a grievance, in order to determine the real matter in dispute and to render a decision according to equitable principles and the justice of the case, provided that the Arbitrator's decision does not contravene any other article of this collective agreement.

9.16 Troubleshooter Provision

Prior to arbitration pursuant to Article 10 (Arbitration), the parties may by mutual agreement, refer a matter to a troubleshooter as follows:

Where a difference arises between the parties relating to the dismissal, discipline, or suspension of an employee, or to the interpretation, application, operation, or alleged violation of this agreement,

including any question as to whether a matter is arbitrable, during the terms of the collective agreement, Chris Sullivan, or a substitute agreed to by the parties, shall at the request of both parties:

- (a) investigate the difference;
- (b) define the issue in the difference; and
- (c) make written recommendations to resolve the difference

within 30 days of the date of receipt of the request and, for those 30 days from that date, time does not run in respect of the grievance procedure.

ARTICLE 10 - ARBITRATION

10.1 Notification

Where a difference arising between parties as per Article 9 (Grievance Procedure) leads to arbitration the aggrieved party shall notify the other party at the same time it files the arbitration.

10.2 Arbitration Referral

Where a difference arises between the parties relating to the dismissal, discipline or suspension of an employee, or to the interpretation, application, operation or alleged violation of this agreement, including any question as to whether a matter is arbitrable, during the term of the collective agreement, an arbitrator from the attached list or a substitute agreed to by the parties, shall at the request of either party:

- (a) investigate the difference;
- (b) define the issue in the difference; and

(c) make written recommendations to resolve the differences within five days of receipt of the request; and, for those five days from that date, time does not run in respect of the grievance procedure.

10.3 Appointment of the Arbitrator

When a party has requested that a grievance be submitted to arbitration, both parties shall meet within seven days to select a single arbitrator, from the agreed-upon list as per Clause 10.4 (List of Arbitrators).

10.4 List of Arbitrators

The parties agree that the following list of arbitrators will be utilized for both full or expedited (re: Clause 10.11 [Expedited Arbitration]) arbitrations:

- Vince Ready
- Judi Korbin
- Chris Sullivan
- Irene Holden

10.5 Arbitration Procedure

The Arbitrator shall determine the procedure in accordance with the *Labour Relations Code* and shall give full opportunity to all parties to present evidence and make representations. The Arbitrator shall hear and determine the difference or allegation and shall render a decision within five days of the first meeting as per Clause 10.2 (Arbitration Referral).

10.6 Decision of Arbitrator

The decision of the Arbitrator shall be final, binding and enforceable on the parties. The Arbitrator shall have the power to dispose of a dismissal or discipline grievance by any arrangement which they deem just and equitable. However, the Arbitrator shall not have the power to change this agreement or to alter, modify or amend any of its provisions.

10.7 Disagreement of Decision

Should the parties disagree as to the meaning of the Arbitrator's decision, either party may apply to the Arbitrator to clarify the decision, which the Arbitrator shall make every effort to do within seven days.

10.8 Expenses of Arbitrator

Each party shall pay the same proportion of the total fees and expenses of the Arbitrator.

10.9 Amending Time Limits

The time limits fixed in the arbitration procedure may be altered by mutual consent of the parties but the same must be in writing.

10.10 Witnesses

At any stage of the grievance or arbitration procedure, the parties may have the assistance of the employee(s) and managers concerned as witnesses and any other witnesses. All reasonable arrangements will be made to permit the concerned parties or the arbitrator(s) to have access to the Employer's premises to view any working conditions which may be relevant to the settlement of the grievance.

10.11 Expedited Arbitration

(a) The parties shall meet every four months or as often as required to review outstanding grievances filed at arbitration to determine by mutual agreement those grievances suitable for this process and shall set dates and locations for hearings of groups of grievances considered suitable for expedited arbitration.

(b) All grievances shall be considered suitable for and resolved by expedited arbitration except grievances in the nature of:

- (1) dismissals;
- (2) rejection on probation;
- (3) suspensions in excess of 20 workdays;
- (4) policy grievances;
- (5) grievances requiring substantial interpretation of a provision of this agreement;
- (6) grievances requiring presentation of extrinsic evidence;
- (7) grievances where a party intends to raise a preliminary objection;
- (8) demotions; or
- (9) grievances that are related to a matter already filed at formal arbitration.

By mutual agreement, a grievance falling into any of these categories may be placed into the expedited arbitration process.

(c) The parties shall mutually agree upon single arbitrators who shall be appointed to hear and resolve groups of grievances.

(d) The Arbitrator shall hear the grievances and shall render a decision within two working days of such hearings. No written reasons for the decision shall be provided beyond that which the Arbitrator deems appropriate to convey a decision.

(e) Arbitration awards shall be of no precedential value and shall not thereafter be referred to by the parties in respect of any other matter.

(f) All settlements of expedited arbitration cases prior to hearing shall be without prejudice.

(g) A grievance determined by either party to fall within one of the categories listed in (b) above, may be removed from the expedited arbitration process and forwarded to a regular arbitration hearing, provided this occurs more than four weeks prior to the scheduled hearing date.

(h) The parties shall equally share the cost of the fees and expenses of the Arbitrator and hearing rooms.

10.12 Troubleshooter Provision

Where a difference arises between the parties relating to the dismissal, discipline, or suspension of an employee, or to the interpretation, application, operation, or alleged violation of this agreement, including any question as to whether a matter is arbitrable, during the terms of the collective agreement, Chris Sullivan, or a substitute agreed to by the parties, shall at the request of both parties:

- (a) investigate the difference;
- (b) define the issue in the difference; and
- (c) make written recommendations to resolve the difference

within 30 days of the date of receipt of the request and, for those 30 days from that date, time does not run in respect of the grievance procedure.

ARTICLE 11 - DISCIPLINARY ACTION

11.1 Procedure

The Employer may discipline for just cause, using the procedures set out below. An employee considered by the Union to be wrongfully or unjustly disciplined, suspended or dismissed shall be entitled to recourse under the grievance procedure, in accordance with Article 9 (Grievance Procedure) of this agreement.

11.2 Right to Have Steward Present

(a) Where a supervisor intends to interview an employee for disciplinary purposes or to obtain information about an incident which might lead to disciplinary action or where a probationary employee is advised of rejection on probation, the supervisor shall make every effort to notify the employee in advance of the purpose of the interview in order that the employee may contact the steward providing that this does not result in an undue delay of the appropriate action being taken.

(b) A steward shall have the right to consult with a staff representative of the Union and to have a local union representative present at any discussion with supervisory personnel which the steward is advised constitutes disciplinary action against the steward provided that this does not result in any undue delay of the appropriate action being taken.

11.3 Burden of Proof

In cases of discipline, suspension and dismissal, the burden of proof of just cause in any arbitration hearing or grievance procedure as outlined herein shall rest with the Employer. In proceedings pursuant to this

article in cases of discipline, suspension or dismissal, the Employer shall not produce evidence other than evidence in support of the allegations outlined in the written notice to the employee and the manner of discipline imposed.

11.4 Letter of Reprimand

When an employee has committed a breach of instructions, policy, and/or procedures, harassment (as defined in Article 23 [Harassment and Bullying]) or professional conduct (as defined in Article 29 [Professional Conduct/Confidentiality]) that employee may be given a formal letter of reprimand. If issued, a copy of the letter issued to the employee shall be called "*letter of reprimand*". The letter of reprimand will summarise the details of the incident(s), noting the reasons why the Employer is concerned. It will also specify any corrective action required by the employee and possible future consequences if such action is not taken. The letter will provide a space for the employee's signature along with an area for employee's comments.

A copy of this letter will be placed on the employee's personnel file along with any letter of response from the employee. This letter will remain on file until the next satisfactory performance appraisal or 12 months whichever is longer; if the performance appraisal is not produced within 10 days of its due date, the letter shall be withdrawn on its first anniversary date.

11.5 Suspension Without Pay

(a) The Employer may suspend an employee for just cause. Notice of suspension shall be in writing and shall set forth the reasons for the suspension.

(b) Where an employee has been suspended pending investigation of their conduct, the Employer will make every reasonable effort to complete its investigation and make a decision within 45 days of the commencement of the suspension.

(c) Notice of suspension, and any documentation relating to it, shall remain on the Personnel File for a period of 24 months or the next satisfactory performance appraisal, whichever is longer. If the performance appraisal is not produced within 10 days of its due date, the notice shall be withdrawn on its second anniversary date.

(d) No sick time, flex days, vacation days or seniority accrue to an employee who is on suspension, and no other forms of leave may be claimed during the suspension period.

11.6 Dismissal

The Employer may dismiss any employee for just cause. Notice of dismissal shall be in writing and shall set forth the reasons for dismissal.

11.7 Dismissal and Suspension Grievance

All letters of reprimand, dismissals and suspensions will be subject to formal grievance procedure under Article 9 (Grievance Procedure). A copy of the written notice of dismissal or suspension shall be forwarded to the President of the Union within five days of the action being taken.

11.8 Unjust Suspension or Discharge

An employee who is found to have been unjustly suspended or discharged, and providing that no lesser form of discipline is substituted by the Employer (prior to arbitration) or the Arbitrator (after submitted to arbitration), shall be immediately reinstated in their former position without loss of seniority and shall be compensated for all time lost in an amount equal to normal earnings during the period of such

suspension or discharge. Any additional compensation which is considered just and equitable in the opinion of the parties or in the opinion of the Arbitrator may be made.

ARTICLE 12 - SENIORITY

12.1 Seniority Defined

(a) "*Service seniority*" means a continuous employee's length of service with the Employer. Employees shall be credited with service seniority back to the date they commenced employment with the Employer. Part-time employees shall earn seniority on a prorated basis for all purposes.

(b) "*Position seniority*" means a continuous employee's length of service in the present position.

(c) Unless otherwise provided in this agreement, the term "*seniority*" shall include both position and service seniority.

(d) When two or more employees have the same service seniority date, the tie shall be broken by chance. The method shall be as mutually agreed between the parties.

12.2 Seniority List

The Employer shall maintain a service seniority list showing the date each employee commenced employment with the Employer. An up-to-date seniority list shall be sent to the President of the Union on or before March 31st of each year.

12.3 Loss of Seniority

(a) An employee on leave of absence without pay, other than leave of absence for an elected or appointed position in the Union, or leave granted under Article 21 (Maternity and Parental Leave) or any employment-protected leave under Part 6 of the *Employment Standards Act*, shall not accrue seniority for leave periods over 30 consecutive working days in a 12 month period.

(b) Employees on short-term, long-term or WCB disability benefits will continue to accrue service seniority without benefits. This is seniority only for the purpose of calculating length of employment in the event of determining who shall be laid off. Severance pay shall not be paid for the length of time someone was away receiving disability benefits, in the event of layoff.

(c) A continuous employee who is on leave of absence without pay in an elected or appointed position of the Union shall continue to accrue service seniority without benefits during the leave period, provided that, upon returning, the employee shall accept the first available position in their original classification at the work location nearest their residence.

(d) An employee shall lose their seniority as a continuous employee in the event that:

(1) they are discharged for just cause;

(2) subject to Clause 12.4 (Re-Employment), they voluntarily terminate their employment or abandon their position;

(3) they are on layoff for more than one year.

12.4 Re-Employment

(a) An employee who and is re-employed within 120 working days of their termination date is deemed to have been reinstated and shall retain rights in relation to salary grid position, to service seniority and other benefits, as they were as of the date they left. This reinstatement shall only continue

in effect providing all pension contributions which were paid out after the date of resignation, are returned to the employee's pension account within 30 days of recommencing employment. If the pension contributions are not returned in full within 30 days, the employee shall give up all the reinstated entitlements; the employee's seniority, salary and other benefits shall be those of a brand new employee as of the 1st day after the unpaid pension funds should have been returned.

(b) An employee who leaves their employment pursuant to Clause 19.8 (Extended Absence for Medical Reasons) shall, if re-employed within one year of termination, be granted all previous seniority and length of service related benefits upon successful completion of the probationary period. The employee's length of service shall be used for the purpose of determining their position on the salary grid, providing the position they have been re-employed in does not constitute a promotion.

12.5 Bridging of Service

If an employee resigns as a result of a decision to raise a dependent child(ren) and is re employed, the employee shall be credited with length of service accumulated at the time of termination for the purposes of benefits based on service seniority. The following conditions shall apply:

(a) The employee must have been an employee with at least three years of service seniority at the time of termination;

- (b) The resignation must indicate that the reason for termination was to raise a dependent child(ren);
- (c) The break in service shall be for not longer than six years; and

(d) The previous length of service shall not be reinstated until successful completion of the probationary period on re-employment.

12.6 Seniority on Demotion

An employee who is demoted through no fault of their own or who takes a voluntary demotion shall have position seniority adjusted to include all service previously held in the lower position together with all service in any higher positions.

ARTICLE 13 - LAYOFF AND RECALL

13.1 Notice to Union

Where the Employer plans to lay off employees or to reorganize the operation in a way which would result in redundancy, demotion or relocation, the Employer shall advise the Union of its intention prior to such layoffs or reorganization being implemented. A formal consultation meeting between the bargaining principals or their representatives shall be held within five working days at the request of either party.

13.2 Layoff and Recall

In the event of a layoff, the following shall apply to continuous employees;

(a) Where the employee's position is relocated, they shall be offered the position in the new location. An employee may decline an offer pursuant to this article.

(b) The Employer shall notify employees affected by Clause 13.1 (Notice to Union) 30 days prior to the effective date, along with an up-to-date service seniority list.

(c) Where a position has been identified by the Employer to be redundant, the employee shall be placed in a regular vacancy of the same or comparable position in the same geographic location

provided they possess the knowledge, skills and abilities to perform the job or could reasonably be expected to acquire such after a period of up to one month retraining/orientation.

- (d) If no appropriate vacancy is available, the following shall apply in sequential order:
 - (1) The employee shall be given the following options:

(i) Displace least senior employee in another position that does not constitute a promotion in the same work unit provided they possess the knowledge, skills and abilities to qualify for the job and demonstrates their ability to adequately perform the job in a one month orientation and training placement in that position, or

- (ii) be placed on the recall list for a period of 12 months, or
- (iii) receive severance as per Clause 14.1 (Entitlement).
- (2) (i) An employee who was previously an incumbent in the same or lower classification position shall have the option of displacing a less senior employee in that classification in the same work unit and geographic location provided they possess the knowledge, skills and abilities to perform the job after a period of familiarization up to one month of retraining/orientation.

(ii) For clarification purposes, an Enforcement Assistant shall be eligible to displace an Enquiry Representative of less service seniority providing they moved from an Enquiry Representative position to an Enforcement Assistant position prior to the Enquiry Representative position reclassification in June 2000.

(3) Employee(s) must notify the Employer within 10 working days of their choice in Clause 13.2(d)(1).

(e) Failing placement under (c) or (d) above, the employee who occupies a redundant position shall:

(1) receive layoff notice in accordance with Clause 13.3 (Layoff Notice);

(2) where another employee is displaced as a result of Clause 13.2(d) above, they shall receive layoff notice in accordance with Clause 13.3 (Layoff Notice).

(f) With respect to all sections of Article 13 (Layoff and Recall), the Employer shall determine if the employee is performing adequately based upon the employee's knowledge, skills, and ability. A person let go because they cannot perform in a new job shall be deemed to have been rejected on probation and deemed to have been laid off.

13.3 Layoff Notice

(a) A continuous employee who is issued layoff notice shall receive 30 working days' notice of layoff or pay in lieu for that period of notice which work is not made available. Such notice shall not be rescinded without mutual agreement with the employee once the employee has made their election. Upon receipt of such notice an employee shall elect from the options in Clause 13.4 (Layoff Options) within five working days.

(b) The continuous employee who is issued layoff notice resulting from actions taken per Clause 13.2 (Layoff and Recall)(d) shall elect one of the layoff options in Clause 13.4 (Layoff Options) below.

13.4 Layoff Options

(a) A continuous employee who is issued layoff notice shall elect one of the following options:

(1) To displace the temporary employee with the longest remaining appointment for which they are deemed to be qualified by the Employer. An employee who elects this option shall accrue seniority during their term on temporary appointment and shall be considered available for recall to continuous positions. Upon completion of the temporary assignment, they shall re-establish their right to a further 12 months of recall to either continuous or temporary positions.

(2) To be placed on the continuous and temporary recall lists for a period of 12 months. Should the employee not receive a written offer of continuous work during their recall period, they shall be entitled to receive severance pay pursuant to Clause 13.4(a)(3).

Written offers of employment shall be sent to the employee by double registered mail.

(3) Within 30 days of receipt of notice of layoff, or of refusing job offers in accordance with Clauses 13.2 (Layoff and Recall) and 13.4(a)(1), a continuous employee will be entitled to accept termination with no right of recall and severance pay based upon years of service, as per the chart in Article 14 (Severance Pay).

(b) Severance Pay Deferral

For the purposes of Clause 13.4(a)(3) a continuous employee shall be entitled to receive severance pay based upon years of service calculated at most recent salary rate for each completed year of service pursuant to the entitlement specified in Clause 13.4(a)(3).

(c) In the event that substantial and/or unanticipated difficulties in Employer funding arise, which alters the basis upon which Clause 13.4 (Layoff Options) was negotiated, the parties shall meet to discuss provision for pay out of severance monies to employees. In which case, the 12 month recall/qualifying period may be waived by mutual agreement.

13.5 Salary Protection

A regular employee who is assigned to a continuous vacancy, or who displaces a continuous employee which results in a demotion shall receive salary protection in accordance with Clause 30.7 (Salary Protection).

(a) However, the employee must accept subsequent job offers to higher rated positions for which they have the skills and abilities or forfeit the protection of Clause 30.7 (Salary Protection) after the expiration of 12 months from the date of demotion.

(b) After the expiration of 12 months and where the employee has not secured a higher rated position, the employee's salary rate shall become the top step of the classification occupied.

13.6 Retraining and Adjustment Period

Employees who assume a new position pursuant to this article will receive job orientation, including, where deemed appropriate, current in-service training, and shall be allowed 30 days to familiarize themselves with their new duties and to become proficient at them to the satisfaction of the Employer.

13.7 Recall

(a) Continuous employees who are laid off and opt to be on the recall list shall be placed on a recall list for 12 months.

(b) No new continuous employees shall be hired in the same position until employees on the recall list are recalled as specified in Clause 13.4(Layoff Options)(a). Employees on the recall list may be placed in an equivalent or lower vacant position providing they possess the knowledge, skills and abilities to perform the job. Employees on recall have the same rights with respect to competitions as other employees. If it is a different classification the employee shall be on probation as per Article 27 (Probation).

(c) A continuous employee who opts to remain on the recall list shall be recalled to a continuous or acting position in order of service seniority for positions at the same pay, or lower level provided they possess the knowledge, skills and abilities to perform the job or could reasonably be expected to acquire such after a period of up to one month retraining/orientation.

(d) The Employer will attempt to contact the employee for continuous and/or temporary assignment by telephone and registered letters for one working week.

(e) It is the responsibility of the laid off employee to ensure that the Employer is kept notified of their current mailing address, telephone number and availability.

(f) If the recalled employee with the highest service seniority is unavailable or refuses work, the employee with the next highest service seniority in the classification shall be recalled.

(g) A continuous employee shall remain on the recall list for 12 months from their date of layoff. Their service seniority shall be maintained during that period, however, seniority shall be accrued and wages and other benefits shall be paid only for periods worked.

(h) A continuous employee on the recall list who has not accepted full-time employment elsewhere may continue on the benefits plan for which they are eligible (basic medical plan, dental, extended health care, accidental death and dismemberment, and group life) by prepaying the monthly premiums for such coverage for a maximum of six months, subject to the approval of the Insurer. This does not include Short-Term Disability (STD) which ceases as of the day layoff commences. The Employer will pay its share of the premiums for 60 days, including the amount it would have paid for long-term disability, were it available.

(i) Employees who relocate pursuant to Article 13 (Layoff and Recall) shall be entitled to relocation expenses in accordance with Clause 30.17 (Relocation Expenses).

13.8 Receipt of Due Monies

An employee who is laid off shall receive, no later than the effective date of the layoff, net payout of all earned salary, vacation leave, and compensatory time off.

13.9 Termination of Benefits

When an employee is laid off, the employee's group benefits defied in Clauses 31.3 (Group Plan) and 31.5 (Cost of Premiums) are terminated on the last day of work (with the exception of MSP) unless the employee is receiving pay in lieu of notice, in which case the employee is entitled to be paid the Employer's share of the benefits premium they would otherwise have received during the notice period. This does not apply to an employee who is placed on the recall list defined in Clause 13.7 (Recall).

13.10 Notice to Employee

(a) If the employee is not given the opportunity by the Employer to work the full notice period they shall be paid in lieu of work for that part of the notice period for which work was not made available.

(b) Following such notice, employees so affected, who are seeking alternative employment, shall, subject to operational requirements, receive leave with pay to attend job interviews. The Employer agrees that such leave shall not be unreasonably withheld. The Employer may require reasonable proof that the employee has attended the job interview for which the time was taken off.

13.11 Volunteers

The Employer agrees not to assign work to volunteers, which would result in the layoff of an employee, or would allow the Employer to keep an existing position vacant.

ARTICLE 14 - SEVERANCE PAY

14.1 Entitlement

Continuous full-time employees who are laid off shall be entitled to severance pay as per Clause 13.4 (Layoff Options) at the rate noted in the following chart. Any part-time service for a continuous employee shall be prorated accordingly. No severance pay shall be paid to an employee who is dismissed, or who is rejected on probation.

Length of Service Seniority	Severance Amount
Less than six months	No severance pay
Over six months but less than two years	Two week's pay at current rate
Over two years	One week's pay at current rate for every complete year of service with the Employer, up to a maximum of 26 weeks

ARTICLE 15 - HOURS OF WORK

15.1 Hours of Work

The annual hours of work for a full-time employee is 1957.5 hours.

A daily shift consists of either 7.5 or 8 hours, depending on whether the employee is working under compressed time.

The employee shall designate their shift preference and that shift preference within their team coverage structure shall be agreed to by the Employer unless operational requirements do not permit the granting of that shift. If not all employees can be given their shift preference because of operational requirements, employees will be entitled to work a particular shift in order of position seniority. The Employer's approval of shifts are made using the following principles and guidelines:

If there is not adequate coverage on any shift, and agreement cannot be reached voluntarily on how to fill the shift, employees shall be assigned in reverse order of position seniority, except where the position in question is that of Enquiry Representative. For Enquiry Representatives the Employer may require the employees assigned to have a minimum of one year's experience in that position.

7.5 Hot	ur Shift	8 Hour Shift	
One hour lunch	30-minute lunch	One hour lunch	30-minute lunch
7:30 a.m 4:00 p.m.	8:00 a.m 4:00 p.m.	7:30 a.m 4:30 p.m.	7:30 a.m 4:00 p.m.
8:00 a.m 4:30 p.m.	8:30 a.m 4:30 p.m.	8:00 a.m 5:00 p.m.	8:00 a.m 4:30 p.m.
8:30 a.m 5:00 p.m.	9:00 a.m 5:00 p.m.	8:30 a.m 5:30 p.m.	8:30 a.m 5:00 p.m.

Once a shift has been selected and approved, it may be changed by mutual agreement subject to operational requirements. If the change cannot be accommodated because of operational requirements, and a change cannot be made by voluntary exchange of shifts, the employee cannot use their service

seniority to displace an existing employee of lower service seniority in the preferred shift until the following April 1st, provided notice of the desired shift change has been received from the employee by January 15th.

Note: Team Coverage Structure - the group of people who normally provide day to day coverage for each other.

15.2 Afternoon Shift

Any employee may be required to work one afternoon shift per week (ending no later than 8:00 p.m.) on a regular or as needed basis, provided that at least two weeks written notice has been provided to the employee by the Employer.

15.3 Flex Days and Compressed Time

An employee who has received unsatisfactory Performance Appraisal or a Letter of Reprimand may, at the Employer's reasonable discretion, not be eligible for compressed time for the next six months.

Payment Processing Clerks are not subject to this language (See Memorandum of Agreement #4 [Compressed Time - Payment Services]).

(a) Flex Days:

(1) All full-time continuous employees who have not elected, or are not eligible for compressed time, and temporary employees shall accrue five hours and 37½ minutes (three quarters of one workday) for each month of the year worked up to a maximum of 67½ hours (nine workdays) in any calendar year. Staff who elect or are required to work a seven and one-half hour shift take this time as per this article. Staff who work an eight-hour shift take this time as per Clause 15.3 (Compressed Time)(b).

(2) A flex day must normally be taken within one month of the month in which it accrues. An employee may take up to two flex days in July, August and December combined if desired, and the rest roughly evenly across remaining months. Employee preference will be accommodated on the basis of service seniority, and an employee may take all flex days on a Friday, Monday or day following a statutory holiday where that holiday falls on a Monday. Flex days cannot be banked.

(3) An employee shall, where requested by the Employer, because of inadequate staff coverage due to the absence of other staff, postpone their flex day and report to work as usual. If the employee was not given notice of such required postponement until the actual flex day, the employee shall have at least two hours to report to work. If this results in the employee working less than a full shift as per Clause 15.1 (Hours of Work), the employee shall still be paid for a full shift. If an employee cannot be reached for callback, it will be deemed that the employee was never called back.

(4) If a flex day is postponed at the Employer's request, the employee must choose another day within the following 60 days.

(5) A newly hired continuous or temporary full-time employee starts to accrue flex days in the first month which they work 10 full shifts. New employees on probation, and temporary employees on a term of less than six months, are eligible to take their flex days after five months but shall have the first five months accrual paid out in the last pay period of the month in which the flex day was earned.

(6) If an employee resigns or is terminated, no flex days shall accrue for that month, if they have worked less than 10 days. However, if a flex day has already been taken as scheduled for the

month in which the employee's resignation or termination takes effect, no charge shall be made against the employee for that time.

(7) Part-time and casual employees are not eligible for flex or compressed days.

(b) *Compressed Time:*

(1) Employees who are eligible to select the compressed time option work eight-hour daily shifts. New employees on probation (including temporary employees) are not eligible for compressed time until five months has elapsed.

(2) Employees who wish to work an eight-hour shift for the following year shall indicate so no later than December 1st of each year. The Employer shall then draw up a work schedule in consultation with the employee and/or work team based on the following:

(i) the employee will normally be scheduled for one working day off every two to three weeks, where possible, for all months, except an employee may take one day in each of July and August, up to a maximum of 20 days;

(ii) the day off will be scheduled across all working days of the week except the one day (either Tuesday or Thursday) designated as "*flex and compressed free*", and the compressed day shall not be banked. Prior to the posting of the approved vacation schedule, flex days can be exchanged within the same team. After the approved vacation schedule is posted, flex days can be exchanged within the same component in the same office or days can be changed for a similar day of the week within a 60-day period before or after the originally scheduled date if adequate coverage is maintained. In exceptional circumstances, exchanges and changes outside of these parameters may be approved by mutual agreement of the Employer and the employee;

(iii) the number of employees who can be absent on a compressed day on any given weekday shall be determined by the Employer within each office by team and subject to operational requirements;

(iv) Where an employee on a team of three or more employees requests at least three days of leave in a row, in the same week, a flex day held by another employee will not restrict the requesting employees' ability to take their desired leave subject to the following: One additional employee on the team can be off where their time off would normally exceed the maximum number of staff off as set by operational requirements. The maximum that operational requirements can be exceeded by is two days per week.

Where an employee on a team of two employees requests leave that would not meet operational requirements, it may be accommodated per (ii) above.

(v) if urgent operational requirements necessitate the Employer changing the scheduled day, the Employer shall re-schedule the compressed day in consultation with the employee and such day shall be taken within 60 days of the scheduled day;

(vi) An employee who is on a day off scheduled under this section shall, where requested by the Employer, because of inadequate staff coverage due to the unscheduled absence of other staff, postpone their day off and report to work as usual. If the employee was not given notice of such required postponement until the actual day off, the employee shall have up to two hours to report to work. If this results in the employee working up to two hours less than a full shift as per Clause 15.1 (Hours of Work), the employee shall still be paid for a full shift. If an employee cannot be reached for callback,

(vii) the schedule will be developed for the entire fiscal year, and shall rotate the days off through the four allowable days of the week so that over the course of a year, employees get approximately the same number of Mondays and Fridays off. The schedule will be distributed to all employees in advance of the vacation selection commencing.

(3) An employee who wishes to switch from compressed time shifts to regular shifts may do so but the change will not be effective until the following April.

(4) The Employer may remove any position from participating in the compressed time arrangements or may exclude a component from having compressed days off in July and August, because the overall job function is seriously and adversely affected. Prior to removing any position, the Employer will work with the employees to see if there is any way to modify the compressed time arrangements so that it might work.

If the modifications are not feasible, or do not alleviate the problem, the Employer will advise the Union of its intention to remove the position in question. The Union may request a meeting to explore other options, and the Employer agrees to end participation in compressed time only as a last resort.

15.4 Shift Adjustments

The Employer may require any employee to adjust their hours of work on certain occasions or to work overtime as per Article 16 (Overtime) to deal with emergencies or urgent situations, providing that at least 24 hours' notice has been provided to the employee.

15.5 Coffee Breaks

Employees are entitled to one 20-minute break for each one-half day (three hours and 45 minutes) worked. Breaks must be taken on the workday in which they are earned and cannot be banked or used to alter the normal length of the shift. Adequate coverage will be assured during breaks, and one break will be taken in the morning, and one break in the afternoon as scheduled by the Employer.

15.6 Meal Breaks

Employees are entitled to a one hour or a one-half hour unpaid meal break, depending on their selected shift schedule, for each full day worked to be taken at a time scheduled by the Employer in consultation with the employee.

Meal breaks cannot be banked or used to alter the normal length of the shift, with the exception that 30 minutes of a one-hour meal break can be used to make up paid but unworked hours.

Part-time employees who work more than five consecutive hours are entitled to a one-half hour meal break.

15.7 Hours of Work Record

All employees shall complete, in full, the Hours of Work Form and shall submit it to their supervisor no later than the end of the first working day of the following week. The employee shall only record hours worked in excess of the regular number of hours per day if those excess hours were given prior approval by the supervisor.

15.8 Making Up Paid but Unworked Hours

Where an employee must arrive at the workplace late or leave the workplace prior to completing the shift (whether or not the employee is able to return to the workplace before the end of the shift), due to reasonable circumstances, the employee shall have the following options:

- (a) to use earned vacation time or compensatory time off;
- (b) to make up the time by working all or part of the next flex day;

(c) to make up time by adding time to the normal shift, as mutually agreed between the employee and the Employer, 30 minutes must come from the meal break if the meal break is 60 minutes. Coffee breaks may not be shortened. The maximum amount of time allowed shall not exceed 40 hours in any fiscal year except pursuant to (f) below.

(d) to have the time not worked deducted from the next pay period provided the time to be adjusted exceeds one hour.

(e) There is one exception to the definition of reasonable circumstances outlined above. Each staff member will be permitted to make up 20 hours per year, at 30 minutes or more per request. These requests must be pre-authorized by the staff person's manager and will only be approved subject to operational requirements. Any time taken under this exception will be deducted from the 40 hours of make-up time allowed in (c) above. Staff who are on a performance management track or on an absentee management plan are not eligible to request this form of leave.

(f) The Employer shall consider and may grant requests to make up time in excess of the limit in (c) above in exceptional or unforeseeable circumstances or on compassionate grounds. Approval of such requests shall not be unreasonably withheld.

15.9 Office Closure

If weather or natural disaster make it necessary to close an office, the Employer shall do its best to advise every employee of that fact once the decision has been made. If an employee is unsure of the office status they should telephone a manager prior to leaving home. If employees are on any form of leave status as of the date(s) of closure, they shall not receive any form of credit for the closure.

15.10 Continuous Part-Time Work

(a) The Employer will seriously consider requests from employees for part-time work for any positions except those involving management of an ongoing caseload. The Employer will consider the following factors in making its decision:

- (1) the impact on coverage and ability to get the work done;
- (2) the impact on other employees;
- (3) additional cost.

The Employer's decision is final. Part-time work is not intended to substitute for job sharing.

(b) Part-time work shall not be for less than 17½ hours per week, and the schedule shall be determined by the Employer in consultation with the employee at the time the part-time position is approved. Changes to the schedule may be made by the Employer due to operational requirements, providing the employee is given 14 days' notice.

(c) All seniority entitlements and benefits are prorated for part-time employees. Part-time employees are not eligible for flex days or compressed time as per Clause 15.3 (Flex Days and Compressed Time).

(d) The Employer may end a part-time arrangement by giving the employee 90 days' notice. The employee then has the following choices:

(1) to convert to a full-time position if there is a vacancy, or if their total service seniority would allow them to displace another employee as per Clause 13.4 (Layoff Options); or

(2) to be laid off as per Article 13 (Layoff and Recall).

(e) If the employee requests to return to a full-time position, they shall do so in writing and shall be placed in the same position on a full-time basis as soon as one is available. The employee shall be given the shift and holiday schedule they would have been entitled to be based on their position and service seniority respectively, providing it does not require another employee's shift or holiday schedule to be changed during that fiscal year.

(f) The Employer will not normally approve a return to part-time hours where the employee has terminated a previous part-time arrangement.

15.11 Hours of Work - Work Schedules

If the union steward or bargaining committee member at the local level wishes to propose a change to the existing work schedules, in a given component or team, the steward or bargaining committee member shall provide the Employer with the proposal in writing.

Prior to initiating a meeting with the Employer, it is understood that the factors to be addressed in the Union's proposal would include, but not be limited to:

- (a) Any newly proposed work schedule shall meet the hours of operation requirements and shall consider unusual or seasonal demands and functionally linked work groups within the organization;
- (b) Proposed changes must not result in increased cost to the Employer.

The Employer will review any reasonable request, however, the Employer retains the right to deny the proposal based on operational and administrative requirements. The Employer will explain their reasons for such denial in writing.

This article is not intended to override or substitute for any other existing section of Article 15 (Hours of Work).

ARTICLE 16 - OVERTIME

16.1 Definitions

(a) "Overtime" means extra work time authorized in writing in advance by the Employer and performed by an employee in excess or outside of full-time, regularly scheduled hours of work, except where it is being worked to make up for unworked time previously paid for. The Employer may require an employee to work up to five overtime hours in a calendar month to deal with emergencies or urgent situations which are beyond the immediate control of the Employer. Overtime in other circumstances shall be by mutual consent.

An employee volunteering or required to work between 1:00 p.m. and 4:30 p.m. on December 24th or 31st, where such are normal working days, shall be paid at double-time and shall be given a one hour paid meal break from 11:45 a.m. to 12:45 p.m., subject to Clause 16.4 (Sharing of Overtime).

A part-time employee only receives overtime pay when they work more than the number of hours in a full-time shift.

- (b) "*Straight-time rate*" means the hourly rate of remuneration.
- (c) "*Time and one-half*" means one and one-half times the straight-time rate.
- (d) "*Double-time*" means twice the straight-time rate.

16.2 Overtime Entitlement

An employee shall be entitled to compensation at the applicable overtime rates for authorized overtime outside the full-time, regularly scheduled workday.

16.3 Recording of Overtime

Employees shall record starting and finishing times for overtime worked on a form determined by the Employer.

16.4 Sharing of Overtime

Where not caseload specific, overtime work shall be offered or allocated on an equitable basis within each position group.

16.5 **Overtime Compensation**

- (a) Overtime worked shall be compensated at the following rates:
 - (1) time and one-half for the first three hours of overtime on a regularly scheduled workday;
 - (2) double-time for hours worked in excess of (1); and

(3) time and one-half for all hours worked on a day of rest where the Employer/employee mutually agree (except for flex or compressed time in which case straight-time only shall be paid).

The computation of overtime in (1) and (2) is to be on a daily basis and not cumulative. All overtime which is paid out shall be paid by the next pay period following the one in which the overtime was earned.

(b) An employee who works on a designated holiday which is not a scheduled workday shall be paid at the rate of double-time for all hours worked; except for Christmas Day and New Year's Day when the rate shall be double-time and one-half for all hours worked.

(c) An employee on travel status who is required to travel on the Employer's business outside regular working hours shall be compensated at time and one-half for all hours travelled up to a maximum of four hours per round trip except as may be approved by the Employer in exceptional circumstances. There shall be no maximum limit to the number of overtime hours where the Employer directs the employee to drive. The Employer may determine the means and times of such travel.

(d) The employee may choose to receive cash for overtime or equivalent compensating time off in lieu. Time off shall be scheduled at a mutually agreeable time. The Employer will not unreasonably withhold approval for an employee to take time off pursuant to this article.

(e) Any overtime due at fiscal year-end shall either be paid in cash by the Employer within 30 days of fiscal year-end or time off which shall be scheduled at a mutually agreeable time within 60 days.

16.6 Overtime Meal Allowance

An employee who is required to work a minimum of two and one-half hours overtime before or after the scheduled hours of work shall be provided with a meal or shall be reimbursed at the rate as per Clause 30.8 (Meal Allowance) for that meal. A meal break of one-half hour with pay at the overtime rate

shall be given. This article shall not apply to an employee who is on travel status which entitles a claim for lodging and/or meals.

16.7 Overtime Due to Training New Employees

When staff are requested by a manager to spend time training new employees, the Employer recognizes that they may fall behind in their work. In such circumstances, the employee may request overtime, in writing, to catch up their work (except where it is an employee training someone else because of their upcoming planned leave as per Clause 20.20 [Planned Leave]). Overtime shall not be unreasonably denied and the supervisor will specify the maximum number of hours approved.

ARTICLE 17 - STATUTORY HOLIDAYS

17.1 Paid Holidays

(a) The Employer recognizes the following as paid holidays:

New Year's Day	Labour Day
Family Day	Day of Truth and Reconciliation
Good Friday	Thanksgiving Day
Easter Monday	Remembrance Day
Victoria Day	Christmas Day
Canada Day	Boxing Day
British Columbia Day	

(b) Any other holiday proclaimed as a holiday by the federal or provincial government for the locality in which an employee is working shall also be a paid holiday.

(c) Hours of work both on December 24th and December 31st shall be 8:30 a.m. to 1:00 p.m., with no lunch break, although employees present at work on those dates shall be paid for the entire shift.

(d) Notwithstanding (a) and (b) above, should the Government of the Province of British Columbia introduce a statutory holiday to honour Indigenous reconciliation (even if not titled "*Day of Truth and Reconciliation*") on a date other than September 30, employees shall be entitled to the new Provincial holiday but not the Federal holiday. For clarity, the Federal National Day of Truth and Reconciliation on September 30 would not constitute a paid holiday pursuant to (a) and (b) above.

(e) For the period between December 25th and December 30th of each year, any employee may choose any of the following shifts:

7.5 hour shift: 7:30 a.m. - 4:00 p.m. (with one hour lunch) 8:00 a.m. - 4:00 p.m. (with 30-minute lunch)

(f) Shift choices must be made in writing by December 15th of that year.

17.2 Holidays Falling on a Saturday or Sunday

For an employee whose workweek is from Monday to Friday, when any of the above-noted holidays fall on Saturday and/or Sunday, the following Monday (or Monday and Tuesday if applicable) shall be deemed to be the holiday for the purpose of this agreement.

17.3 Holiday Coinciding with a Day of Vacation

Where an employee is on vacation leave and a day of paid holiday falls within that period the paid holiday shall not count as a day of vacation.

ARTICLE 18 - ANNUAL VACATION

18.1 Earning of Annual Vacation

Vacation entitlement at the monthly rate of one-twelfth of the annual vacation leave entitlement in Clause 18.2 (Vacation - Entitlement), shall be accumulated as earned to the employee's credit. The period of calculation for annual vacation entitlement shall be April 1st - March 31st.

18.2 Vacation - Entitlement

A full-time continuous or temporary employee will have an annual vacation entitlement as follows:

Service Seniority	Work Hours Per Fiscal Year	
1 month to 36 months	127.5	
37 months to 60 months	150	
61 months to 144 months	172.5 (Plus 7.5 hours for every year worked in excess of five (5) years to a maximum of 225 hours)	
Effective April 1, 2012 - 157 months or more	232.5	

Part-time employees or employees who resign or are terminated before the end of any fiscal year shall accumulate holidays under this clause on a prorated basis.

In addition to the regular entitlements, at the employee's option, they shall have the right to choose one time payout or time off in the year in which they have completed their 10th, 15th and 20th year of service:

- In 10th year 15 hours
- In 15th year 22.5 hours
- In 20th year 40 hours

18.3 Selection of Vacation Dates

Vacation schedules, once approved by the Employer, shall not be changed, other than in an emergency for either party, except by mutual agreement. Employees with service seniority of less than three continuous months as of March 31st of any year will have all unused holiday credits automatically carried over to the new fiscal year.

(a) Peak Periods

Where more than one employee applies for holiday leave for the same period of time, leave shall be granted on the basis of service seniority. The number allowed off at any one time shall be subject to operational requirements.

The Employer shall circulate or post the vacation schedule by February 1st each year. In order of service seniority within the component of each team, each employee will have three calendar days to specify their preferred dates for vacation time before the list is to be given to the next senior employee. Vacation requests must be completed no later than March 15th each year. It is understood an employee who is absent when the list is circulated to them the Employer shall give reasonable time for the request to be completed. The Employer shall post the approved vacation schedule no later than March 31st each year.

Employees may request single vacation days, however the Employer may not grant the employee's request for further single days if it is in excess of two single days (a day is considered single even if it is

separated by a weekend, holiday or flex day from the next scheduled workday) during peak period time and/or subject to operational requirements.

Employees who do not respond within the three days shall not be entitled to exercise those rights over time selected by a less senior employee.

No more than two-thirds of an employee's annual vacation entitlement as per Clause 18.2 (Vacation Entitlement) will be given during July, August, December and the first 10 days of January, or by mutual agreement within a component/team and the Employer. Part A may be changed to suit the requests of the employees.

(b) Non-Peak Periods

An employee wishing to take vacation leave in non-peak periods shall apply in writing for the desired period of leave at least one week prior to the commencement of such leave. Employees' vacation requests will be considered and approved by seniority up to two-thirds of their total entitlement as long as the request is submitted to the Employer by the same deadlines noted in (a) above. The remaining one-third of vacation entitlement will be considered and approved by the order in which it was received by the Employer. No employee who has vacation approved will have that vacation displaced by another employee's request. The Employer will grant an employee's non-peak requested dates except where it will seriously affect operational requirements. The Employer's decision will be communicated within five working days of the request when it is for time other than peak periods.

18.4 New Employees

An employee is not entitled to take vacation leave during the first four months of employment unless special approval is given by the Employer.

18.5 Vacation Pay

Payment for vacations will be made at an employee's current pay as of the time the vacation leave is taken. If an employee who is in an acting position takes vacation time and is returning into the acting position, the rate of pay shall be at the acting rate. If an employee was in an acting position for more than 20 working days prior to taking vacation time, the vacation time earned while in the acting position shall be paid out at the acting pay rate.

18.6 Approved Leave of Absence With Pay During Vacations

(a) When an employee is absent on vacation leave and experiences a bereavement which would normally qualify that employee for bereavement leave, such bereavement leave shall be granted upon written request from the employee (subject to the limits in Clause 20.1 [Bereavement Leave]) and the days so granted shall replace an equal number of vacation days taken at the time of the bereavement.

(b) When an employee is absent on vacation leave and becomes ill to the point that they were hospitalized or specifically ordered on complete bed rest by a qualified medical doctor, sick leave for the time confined in hospital or to bed shall be granted (subject to the limits in Clause 19.1 [Sick Leave Entitlement]) upon a written request from the employee which is accompanied by adequate medical documentation. The days so granted as sick leave shall replace an equal number of vacation days taken at the time of the illness.

18.7 Callback From Vacation

(a) An employee who has commenced annual vacation shall not be called back to work, except in cases of extreme emergency.

(b) When, during any vacation period, an employee is recalled to duty the employee shall be reimbursed for all expenses incurred thereby in proceeding to the place of duty and in returning to the place from which the employee was recalled upon resumption of vacation, upon submission of receipts (except for meals) to the Employer.

(c) Time necessary for travel in returning to the place of duty and returning again to the place from which the employee was recalled shall not be counted against the employee's remaining vacation entitlement.

18.8 Vacation Carryover

An employee may carry over up to six days (45 hours if 7.5 hour shift, 48 hours if 8 hour shift) vacation leave per vacation year to the next vacation year but it must be used in that year. An employee who takes planned leave pursuant to Clause 20.20 (Planned Leave) and requests the planned leave at least six months prior to the commencement of the planned leave, may carry the aggregate of the vacation leave that accrues during this six-month period and the usual carryover of up to six days of vacation leave, into the planned leave period to be taken as vacation leave during the planned leave period.

A one-time carryover of an additional five days (37.5 hours if 7.5 hour shift, 40 hours if 8 hour shift) will be allowed every five years for employees who have at least five years' service seniority, providing the Employer is advised in writing of the intention to do so by September 30th of the year in which the vacation leave is earned.

18.9 Vacation Accrual

A continuous or temporary employee starts to accumulate vacation credits as of the date they commence employment with the Employer. Part-time employees accumulate holiday leave on a prorated basis. Casual employees hired for a period of less than 60 days shall not receive any vacation leave credits but shall receive 10% of their regular pay each pay period in lieu of benefits and vacation. An employee will be granted use of annual vacation accrual in advance of earning it except where the Employer has reasonable grounds to believe that the employee will not be able to accrue the amount requested. Non accrued vacation time may not be used in conjunction with, or instead of, paid or unpaid sick leave.

18.10 Loss of Accrual

Employees who are absent on any form of unpaid leave (this does not include long-term disability) in excess of 30 cumulative working days in any fiscal year (April 1st - March 31st) do not earn holiday credits for those time periods. Employees who have been suspended do not accrue holiday credits for the time they were on suspension. Employees granted leave as per Clause 3.10 (Time off for Union Business)(a)(5) do not earn holiday credits for any portion of the time they are on such leave.

18.11 Payout of Accrued Vacation Leave

All vacation leave is to be taken in the form of actual time off. Vacation leave credits will only be paid out if an employee ceases to work for the Employer. When an employee resigns, is laid off or is terminated the value of accumulated vacation leave credits shall be added to the final paycheque. The payout of any vacation time will be at the employee's salary as of the time they left the Employer. Employees on MAP leave may request payout of their accumulated vacation credits at any time during the leave.

18.12 Repayment of Vacation Credits Used but Unearned

Employees who have been advanced vacation or sick leave and subsequently leave the Employer prior to working long enough to have earned back the advanced credits, shall have the monetary equivalent of the time still owing to the Employer deducted from their final paycheque. If an employee's final

paycheque is not sufficient to reimburse the Employer for the holiday time owing, the Employer shall not release any other funds (this does not include pension money) which may be due to the employee until the employee has reimbursed the Employer.

ARTICLE 19 - SICK LEAVE

19.1 Sick Leave Entitlement

A full-time continuous or temporary employee shall earn paid sick leave at the rate of six and one quarter hours for each full month of service in which pay was received for at least 10 days (75 hours) up to a maximum of 75 hours per fiscal year. A part-time employee shall be entitled to paid sick leave credits on a prorated basis. Where an employee is absent from work because of illness or injury the employee shall be entitled to claim sick leave at the regular rate of pay for a maximum period equivalent to the accumulated paid sick leave credit. Employees shall be entitled to bank up to 150 hours of earned sick leave credits at any time. Any unused sick time in excess of 150 hours as of March 31st shall be converted to discretionary days at a ratio of two days (15 hours if seven and one-half hour shift, 16 hours if eight-hour shift) to one day to a maximum of three days. No partial day conversion will be granted. Discretionary days will be scheduled by mutual agreement for use in the next fiscal year. This means an employee may not have more than 225 paid sick hours in any fiscal year.

For compliance with the personal illness or injury leave provisions of the *Employment Standards Act*, a continuous or temporary employee may access five days of unaccumulated paid sick leave after 90 consecutive days of employment. Unaccumulated paid sick leave days will be offset against sick leave accumulated in the balance of the calendar years. Any negative balance as of December 31st will be reset to zero.

19.2 Use of Paid Sick Leave

Sick leave may be taken in the event of illness by the employee or an immediate family member of the employee which requires the employee's presence at home. The entitlement to use sick leave to care for an immediate family member applies only to paid sick leave. Employees who are absent for more than three consecutive days because of illness of themselves or a family member must submit on the fourth day at the Employer's request, a doctor's note with their application for sick leave. See Clause 20.1 (Bereavement Leave)(b) for a list of the family members who qualify under this article. The Employer shall reimburse the employee for the cost of such a note on presentation by the employee of a paid invoice from the doctor. If the employee does not provide this doctor's note and the Employer does not consider the employee's explanation for not having such a note to be reasonable, the sick leave applied for will not be granted as paid leave and the Employer may take disciplinary action. Upon return to work the employee must submit the sick leave form within two working days.

For illness of a chronic nature, the employee may submit a doctor's note stating the illness is chronic and the note will be kept on file for future use on subsequent absences for that illness for up to three years. This does not preclude the Employer from requesting updated information as necessary during the three-year period.

19.3 Unpaid Sick Leave

Any continuous employee (i.e., not a new employee on probation) who has used up all their sick leave credits as per Clause 19.1 (Sick Leave Entitlement), and is required to be absent from work due to illness or injury, is on a status of unpaid sick leave. The entitlement to use sick leave to care for an immediate family member does not apply to unpaid sick leave. The following are the conditions that apply to unpaid sick leave:

(a) Employees Not on STD/LTD or WCB Benefits

(1) The employee must produce a doctor's note where the cumulative absence in a fiscal year exceeds two unpaid sick leave days, except where the employee subsequently goes on Short-Term Disability (STD) or WCB benefits.

(2) Where there is a repeated pattern of using unpaid sick leave, and the employee does not subsequently qualify for STD/LTD or WCB benefits, the Employer may require the employee to have a medical assessment by the employee's own physician, at the Employer's expense, to determine that the employee is able to perform their duties.

(3) An employee taking an unpaid sick leave day may make the time up on their next scheduled flex or compressed day so that there is no financial penalty to the employee. This does not change the fact that the original day taken was unpaid sick leave.

This article does not restrict the normal management rights of the Employer with respect to managing absenteeism.

(b) Employees Who Have Qualified for STD or WCB Benefits

(1) The employee must contact the Employer after each visit to a doctor about the illness(es) for which the employee is absent, or every two weeks, whichever is sooner, to advise whether the employee is still unable to return to work and if an expected date of return to work is known;

(2) An employee who fails to report as per (1) shall be contacted in writing by the Employer reminding them of their obligation. If they still fail to report within 15 days, and are not reasonably prevented from reporting by the nature of their illness, they shall be deemed to have abandoned their position.

(3) When the employee has been absent from work for more than 60 continuous days, the employee must provide, in writing, at least 10 working days' notice to the Employer of their date of return; the Employer is under no obligation to permit the employee to return to their position prior to such notice being given;

(4) An employee must advise the Employer within five days if they have ceased to qualify for STD/WCB benefits and have not returned to work.

(c) Employees Who Have Qualified for LTD or WCB Benefits Greater than Four Months

(1) An employee must advise the Employer within five days of ceasing to qualify for LTD/WCB benefits if they have not returned to work;

(2) An employee who has qualified for LTD/WCB benefits must provide at least 30 days' notice they are fit to return to work except where there is an existing vacancy.

19.4 Loss of Accrual

Employees who are absent on unpaid leave in excess of 30 cumulative working days in any 12-month period and employees who are on short-term or long-term disability, do not earn sick leave credits for these periods.

This provision does not apply to maternity leave, parental leave, family responsibility leave, compassionate care leave, Reservists' leave, leave respecting disappearance of child, leave respecting death of child, leave respecting domestic or sexual violence or critical illness or injury leave.

19.5 Advising Employer

Employees who are unable to report in for work on any day because of illness, must phone or email their supervisor or their office manager by no later than their scheduled shift start time that same day to advise that they will not be coming in. The employee has the responsibility to advise their supervisor or colleague of any important matters which must be looked after that day. Disciplinary action may be taken against an employee who does not advise the Employer unless there is a reasonable explanation.

19.6 Deduction of Sick Leave

All absences on account of illness or injury on a normal working day (exclusive of designated paid holidays) shall be charged against an employee's sick leave credits, except as noted in Clause 22.7 (Injury Pay Provision).

19.7 Repayment of Sick Leave Credits Used But Unearned

Employees who have used all or part of their year's sick leave entitlement and subsequently leave the Employer prior to earning the sick leave credits used (as per Clause 19.1 [Sick Leave Entitlement]), shall have the equivalent monetary value of the days owing deducted from their final paycheque.

19.8 Extended Absence for Medical Reasons

The Employer acknowledges its obligation to accommodate a disabled employee up to the point of undue hardship and will take a worker-centred approach to such accommodation, including without limitation graduated return to work plans, modification of duties, and alternate duties. The Employer and the Union will work collaboratively to support disabled employees in their safe return to work.

If an employee is absent from work for medical/health reasons for an extended period and the employee has qualified for STD/LTD or WCB benefits, or any other income replacement disability benefit, the following shall apply:

(a) If the employee is absent for a cumulative period of four months in any 12-month period, the employee must provide the Employer with a medical statement indicating when the employee will be able to resume their normal duties.

(b) If the expected date of return is in the next 18 months, the Employer will keep the position so the employee may return prior to, or on the scheduled return date.

(c) If the employee's scheduled return date is past the 18-month period the Employer may fill the position on a permanent basis. The employee shall then be considered on unpaid sick leave and may return to work within a further 18 months (i.e. 36 months total), providing they are certified as fit to resume all duties and there is a vacancy in the same position and worksite. If no vacancy exists in the same position, the employee will be offered vacant positions for which they are qualified.

(d) An employee who is absent due to an injury suffered while on the Employer's business shall not have that time included in (a) to (c) above.

19.9 Scheduling of Medical/Dental Appointments

(a) Employees are expected to schedule appointments on a flex/compressed day off, or other planned day off.

(b) Where a medical or dental appointment cannot be reasonably obtained outside regular working hours, the employee shall be permitted to reasonable time off for medical and dental appointments for employees or for their dependent children, other dependent family members, or any person who is like

a close relative to the employee, whether or not they are related by blood, adoption, marriage or common-law partnership.

(c) Time away from work for such appointments may be made up by adding up to 60 minutes extra per day (30 minutes of which must come from the meal break if the meal break is one hour) during normal office hours (coffee breaks may not be used) up to a maximum of 24 hours per fiscal year, or may be deducted from sick leave, flextime, holiday allotment, or compensatory time owing, providing not all credits for the year have been used.

(d) An employee may always choose to have this paid but unworked time deducted from the next paycheque.

(e) Excluding emergencies, two working days' notice will be required for pre-set appointments.

19.10 Employer Share of Benefits

The Employer shall continue to pay its share of the benefit premiums defined in Clauses 31.2 (Commencement of Benefits) and 31.5 (Cost of Premiums) for a maximum of six months for employees who have qualified for Short Term Disability (STD), WCB or long-term disability benefits providing the employee reimburses the Employer for the employee's share of the benefits within 10 days of receiving an invoice from the Employer. The Employer shall issue the invoice upon payment of the employee's share of the benefits, and if the invoice is not paid by the employee within 10 days of the invoice being issued the benefits shall be terminated.

If the employee wishes to retain the benefits after six months, subject to the approval of the Insurer, the employee must pay 100% of the benefits premium within 30 days of receiving an invoice from the Employer. If the employee does not reimburse the Employer within this time period, the benefits shall be terminated.

ARTICLE 20 - OTHER LEAVE

20.1 Bereavement Leave

(a) The Employer shall grant up to five days paid leave (up to seven days if travel over 500 kilometres is required) in the event of the death of an immediate family member. Bereavement leave may be given verbally and later confirmed in writing. The Employer may extend the period of bereavement leave with or without pay.

(b) Immediate family for the purposes of this article and Clause 19.2 (Use of Paid Sick Leave) is defined as: any relative with whom the employee permanently resides; an employee's parent; former guardian; wife; husband; fiancé; common-law spouse; same sex partner; child; foster child in the care of the employee; brother; sister; father-in-law; mother-in-law; and stepparents.

(c) In the event of the death of an employee's grandparent, grandchild, son-in-law, daughter-in-law, brother-in-law, sister-in-law, aunt, uncle, niece or nephew, the employee shall be entitled to paid leave for up to three days in any fiscal year. The Employer may extend the period of bereavement leave with or without pay.

(d) Employees on any form of leave of absence without pay are not eligible for bereavement leave while they are on such status.

20.2 Special Leave

An employee not on leave of absence without pay shall be entitled to special leave at the regular rate of pay for the following up to a maximum of five days per fiscal year for all categories combined except the maximum number of days which may be taken under (d) and (e) is two days each:

(a)	marriage of the employeefour days;
(b)	attending wedding of employee's childone day;
(c)	birth or adoption of the employee's childone day;
(d)	serious household (employee's) or serious domestic emergencyone day;
(e)	moving of employee's household furniture and effects if employee changes permanent residencetwo days;
(f)	divorce hearing of employeeone day;
(g)	attending their formal hearing, or to take their oath to become a Canadian citizenone day;
(h)	to attend the funeral of a relative or close friend where bereavement leave is not eligible under 20.1one day;

Two weeks' notice is required for leave under Subsection (a), (b), (e), (f) and (g). The Employer may request information or documentation where appropriate to be satisfied that the leave is justified.

20.3 Other Observances

(a) Where established ethno-cultural or religious practices provide for ceremonial occasions, employees may request up to four days leave without pay per calendar year. Such leave shall not be unreasonably withheld.

(b) Employees shall provide the Employer with the dates for which the leave is requested and will provide a minimum of four weeks' notice.

20.4 Compassionate Leave

The Employer may grant special compassionate leave with or without pay for reasons other than those set out in Clause 20.2 (Special Leave). This leave is for purposes not contemplated or provided for by any other clause in the agreement. Permission is required from the Employer to be absent but only such explanation as is necessary to confirm for the Employer the appropriateness of the purpose shall need to be given and such explanation shall remain confidential.

20.5 Critical Illness or Injury Leave

An employee is entitled to a leave of absence without pay of up to

(a) 36 weeks to provide care or support to a family member who is under 19 years of age at the start of the leave, or

(b) 16 weeks to provide care or support to a family member who is 19 years of age or older,

if they are entitled to critical illness or injury leave under the *Employment Standards Act* and such leave will be in accordance with the *Employment Standards Act*. There will be no interruption in the accrual of seniority or eligibility for benefits.

20.6 Compassionate Care Leave

(a) In accordance with the *Employment Standards Act*, an employee will be granted a compassionate care leave of absence without pay for up to 27 weeks to care for a gravely ill family member. For the purpose of this article, *"family member"* includes immediate family as well as other relatives and individuals considered to be like family, whether or not related by marriage, common-law partnership, or any legal parent-child relationship. In order to be eligible for this leave, the employee must provide a medical certificate as proof that the ill family member needs care or support and is at risk of dying within 26 weeks.

(b) An employee who is granted a compassionate care leave of absence to care for a gravely ill family shall be entitled to the benefits as follows:

(1) The employee's benefit coverage will continue for the duration of the compassionate care leave, to a maximum of 27 weeks, and the premium payment shall be on the same basis as if the employee were not on leave.

(2) Where an employee elects to buy back pensionable service for part or all of the duration of the compassionate care leave, to a maximum of 27 weeks, the employer will pay the employer portion of the pension contribution in accordance with the pension plan rules.

(3) Compassionate care leave, up to a maximum of 27 weeks, shall be treated as continuous employment for the purposes of seniority accrual.

(4) An employee who returns to work following a leave granted under this provision shall be placed in the position the employee held prior to the leave, or in a comparable position.

(c) Additional Leave

Should an employee require additional time to care for a gravely ill family member, additional leave may be granted beyond the 27-week period specified. Such additional leave shall be in accordance with the *Employment Standards Act*.

20.7 Leave Respecting the Disappearance of a Child

An employee is entitled to a leave of absence without pay of up to 52 weeks if they are entitled to leave respecting the disappearance of a child under the *Employment Standards Act* and such leave will be in accordance with the *Employment Standards Act*. There will be no interruption in the accrual of seniority or eligibility for benefits.

20.8 Leave Respecting the Death of a Child

An employee is entitled to a leave of absence without pay of up to 104 weeks if they are entitled to leave respecting the death of a child under the *Employment Standards Act* and such leave will be in accordance with the *Employment Standards Act*. There will be no interruption in the accrual of seniority or eligibility for benefits.

20.9 Family Responsibility Leave

An employee is entitled to up to five days of unpaid leave during each year to meet responsibilities related to

- (a) the care, health or education of a child in their care, or
- (b) the care or health of any other member of the employee's immediate family.

There will be no interruption in the accrual of seniority or eligibility for benefits.

20.10 Leave for Domestic or Sexual Violence

(a) Where leave from work is required due to an employee, an employee's dependent child or another person as prescribed by law being a victim of domestic or sexual violence, the employee shall be granted leave, in each calendar year, as follows:

- (1) up to 10 days of unpaid leave to be taken intermittently or in one continuous period; and
- (2) up to 15 weeks of unpaid leave.

(b) Notwithstanding the above, the Employer will provide pay for five of the days referenced in (a) above. In the event the *Employment Standards Act* is changed regarding domestic violence leave to provide more than five days' paid leave, the Employer will provide such leave consistent with the *Act*.

20.11 Donor Leave

The Employer and the Union encourage employees to register as organ donors. An employee shall be granted the necessary leave of absence with pay for the purpose of donating bone marrow or an organ.

20.12 Public Duties

The Employer will grant, on written request, leave of absence without pay and without loss of seniority:

(a) for employees to seek election in a municipal or regional district, provincial, federal, or Indigenous election for a maximum period of 90 days;

(b) for employees elected to or appointed to a public office for a maximum period of five years.

20.13 Leave for Court Appearances

(a) The Employer shall grant paid leave to employees, other than employees on leave without pay, who serve as jurors or are subpoenaed as witnesses in a court action. The employee must produce an accounting of days and times the court is in session and, where reasonable, is expected to report to work for those portions of the day not occupied with court duties. Failure to report the times or to report to work will result in the employee not being eligible for pay while serving as a juror or witness.

(b) In cases where an employee's private affairs have occasioned a court appearance, such leave to attend court shall be granted without pay.

(c) An employee in receipt of regular earnings while attending court shall remit to the Employer all monies paid to them by the court, except travelling and meal allowances. The employee shall produce a statement of days/time the court was in session and an account of the monies paid.

(d) In the event an accused employee is jailed pending a court appearance, such leave of absence shall be without pay up to a maximum of three months, after which time the employee shall be dismissed.

20.14 Elections

Any employee eligible to vote in a federal, provincial, municipal or regional district, or Indigenous government election or referendum shall be entitled to reduce the workday, as specified by the Employer, by the time necessary to ensure the minimum number of hours to vote stipulated in legislation is met. That portion of the workday that may be required to provide such clear time shall be paid leave.

The Employer may grant, subject to operational requirements, leave of absence without pay to an employee to participate in an election campaign. The employee does not have to disclose to the Employer their political affiliation.

(a) In the case of a federal or provincial election, such leave shall not commence until the election writ is issued, and shall terminate no later than the day after the election.

(b) In the case of an Indigenous or municipal or regional district election, such leave shall not commence earlier than one month before the election and shall terminate no later than the day after the election.

20.15 Absence Without Leave (Abandonment of Position)

An employee who fails to report to work and does not request and/or is not eligible for authorized leave as per Articles 18 (Annual Vacation), 19 (Sick Leave), or 20 (Other Leave), shall be deemed to be absent without leave and may be terminated after five days absence except where there are reasonable circumstances which prevented the employee from reporting or requesting leave for which the employee is eligible.

20.16 Leave Without Pay

The Employer may, for an emergency or other exceptional circumstances, subject to operational requirements, grant an employee leave without pay for a period up to one year, which may be renewed by mutual agreement. All requests and approvals for such leave will be in writing. Employees absent under this paragraph shall be required to reimburse the Employer for the Employer's share of any available benefits premium for any period of such leave over 30 working days. Leave under this article will not be unreasonably withheld.

20.17 Educational Leave

Educational leave without pay may be granted by the Employer to regular employees requesting such leave, subject to operational requirements, in accordance with the following provision: the duration of educational leave granted to regular employees to take advanced or special training which will be of benefit to the Employer and employee may be for varying periods up to one year, which may be renewed by mutual agreement.

20.18 Canadian Armed Forces

Employees who participate in activities related to the Reserve Component of the Canadian Armed Forces may be granted leave of absence without pay, subject to operational requirements, for the purpose of fulfilling their commitments to the Armed Forces.

20.19 Planned Leave

Employees, upon completion of each period of five years of continuous service, at least two years of which are in their current position, may request unpaid leave for a fixed period of between four months and six months. In each office, no more than one employee in a position or 8% of the employees in a position, whichever is greater, may be on planned leave at the same time. When the number of employees requesting planned leave exceeds the number who may be on planned leave pursuant to this article, approval of the requests for planned leave will be subject to service seniority.

Application should normally be made at least three months prior to the requested commencement date.

Approval for planned leave and the commencement date of the planned leave shall be subject to operational requirements. The Employer shall communicate its decision regarding the request for unpaid leave within 30 days and once approval has been given it shall not be rescinded.

The Employer shall pay 25% of the total premiums for medical, extended health, dental and group life for the period an employee is on planned leave. The Employee shall pay 75% of the total premiums for these benefits prior to commencement of the planned leave.

Employees on planned leave do not accumulate seniority (except as per Clause 12.3 [Loss of Seniority][a]) or receive benefits other than medical, extended health, dental and group life benefits and shall return to their former position or to a position of equal rank or pay upon their return from planned leave.

Employees who do not return to work from a planned leave, or who terminate their employment within 180 days of return, shall reimburse the Employer the actual cost of the 25% benefit premium cost.

ARTICLE 21 - MATERNITY AND PARENTAL LEAVE

21.1 Eligibility and Notification

(a) Employees are eligible for unpaid leaves of absence from employment subject to the conditions in this article.

(b) An employee who intends to take a leave of absence under this article will give at least four weeks' notice in writing to the Employer unless there is a valid reason why such notice cannot be given and will inform the Employer in writing of the length of leave intended to be taken.

(c) The Employer may require reasonable evidence of the employee's entitlement to leave under this article.

21.2 Maternity Leave

(a) A pregnant employee is entitled to maternity leave without pay for 17 consecutive weeks.

(b) The period of maternity leave shall commence no earlier than 13 weeks prior to the expected birth date and no later than the actual birth date or date of the termination of the pregnancy. The commencement of leave may be deferred for any period approved in writing by a duly qualified medical practitioner.

(c) An employee who takes maternity leave is entitled to an additional 6 consecutive weeks of unpaid leave if, for reasons related to the birth or the termination of the pregnancy, the employee is unable to return to work at the conclusion of the maternity leave.

21.3 Parental Leave

(a) An employee is entitled to parental leave without pay as follows:

(1) for a parent who takes maternity leave, up to 61 consecutive weeks, which must begin within 78 weeks after the birth of the child;

(2) for a parent who does not take maternity leave, up to 62 consecutive weeks, which must begin within 78 weeks after the birth of the child; and

(3) for an adopting parent, up to 62 consecutive weeks, which must begin within 78 weeks after the child is placed with the parent.

(b) An employee who takes parental leave is entitled to an additional five consecutive weeks of unpaid leave if the child has a physical, psychological or emotional condition requiring an additional period of parental care at the conclusion of the parental leave.

21.4 Return from Leave

On return from leave an employee shall be placed in their former position or in a position of equal rank and salary providing that employee has not been laid off as per Article 13 (Layoff and Recall).

21.5 Health and Welfare Benefits

(a) The Employer shall maintain its portion of employee coverage for medical, extended health, long-term disability, dental and group life while the employee is on maternity or parental leave. The employee portion will be deducted from the maternity or parental leave allowance or accrued vacation time. If this is not enough to cover the benefit cost, the employee shall remit their monthly share of the benefits to the Employer within 30 days of receiving an invoice.

(b) An employee may add any vacation leave earned but not used as of the date of commencement of the maternity or parental leave, to extend the leave, or may use it at any time after their return to work, subject to Clause 18.3 (Selection of Vacation Dates).

21.6 Maternity and Parental Leave Allowances

Continuous employees who have completed six months continuous service and who qualify for maternity and/or parental leave and provide documented evidence they have applied for and will receive Employment Insurance Benefits as per the *Employment Insurance Act*, shall be eligible for the following additional allowance/benefits:

(a) *Maternity Leave* - 13% of weekly salary up to a maximum of \$130 per week (up to 15 weeks maximum).

(b) Parental and Adoption Leave - 13% of weekly salary up to a maximum of \$130 per week (up to 10 weeks maximum) where an employee has opted for standard (35-week) EI parental benefits, or 6.5% of weekly salary up to a maximum of \$65 per week (up to 20 weeks maximum) where an employee has opted for extended (61-week) EI parental benefits.

(c) The weekly allowance shall be paid based on the employee's rate of pay at the time they commenced the leave and shall be adjusted for step increases the employee would have been entitled to during the period they are on the leave. Where the employee was acting in a position for less than 90 days at the time the leave commenced, the rate shall be that of their regular position, not the acting position.

(d) Employees who are part-time and have variable hours shall be paid at the average of their weekly salary for the six months prior to the commencement of the leave.

(e) Where both parents are employees of the Employer, they may decide which employee shall be eligible for the 10-week parental leave allowance, but in no case may the allowance period for both parties exceed 25 weeks.

(f) Employees on maternity or parental leave shall continue to accrue vacation leave at their normal rate for the period of time they are on maternity or parental leave. The Employer will pay the accrued holiday time for the first six months of leave in 12 semi-monthly instalments, starting when the maternity or parental allowance ends. The remaining vacation time accrued after the first six months of maternity or parental leave shall be paid within 30 days of the employee's return to work. The employee may direct any portion of the payment, for accrued vacation leave, to be paid into their pension plan.

(g) If an employee fails to resume their employment with the Employer or fails to complete six months service upon their return, (except where due to layoff), the employee shall reimburse the Employer an amount equivalent to the allowance paid out on a prorated basis (i.e., one-sixth per month for each month less than six months).

(h) Important note for information only: The timing of an employee's leave(s) under this article can affect eligibility for related EI benefits and consequentially the allowances in this clause. The Employer and the Union strongly encourage an employee considering leave under this article to familiarize themselves with related EI benefit eligibility criteria from Employment and Social Development Canada.

21.7 Additional Benefits

The employee on maternity or parental leave may, on application, cash out some or all of their earned, unused vacation credits.

21.8 Seniority Accrual

An employee shall continue to accrue seniority for the entire duration of maternity or parental leave.

ARTICLE 22 - SAFETY AND HEALTH

22.1 Joint Occupational Health and Safety Committee

Each worksite shall have a joint occupational health and safety committee to consist of two employer representatives and two employee representatives, or a greater number of representatives by mutual agreement. In no case shall the employer representatives outnumber the worker representatives. The Committee shall meet as required by WCB Regulations or when either one of the parties deems it necessary, at the Employer's expense. In this capacity, the Employer and employee representative shall:

- ensure there are adequate procedures and training to identify potential violent clients, within statutory limits and knowledge;
- ensure that every employee has, and is briefed on, security and evacuation procedures;
- ensure that there are at least two fire drills and emergency incident drills each year;
- ensure that there is a safety audit performed annually and a furniture/equipment inspection performed quarterly;
- meet with employees individually or in groups to listen to any health or safety concerns and to ensure these are referred to the Senior Manager for response and, where appropriate, action;
- ensure employees are briefed on personal safety and that the worksite is reasonably secure;
- meet after any accident on the worksite to determine if follow up action is necessary.

The Union and the Employer agree that applicable regulations made pursuant to the *Workers Compensation Act* or any other statute of the Province of British Columbia shall be complied with.

22.2 Advising Employer

Any employee who is concerned that equipment or procedures pose an undue risk shall bring those concerns to the attention of their supervisor at the earliest available opportunity.

22.3 Video Display Terminals

(a) No employee shall work at a video display terminal, either reading or entering data, for more than 60 minutes continuously. After 60 continuous minutes, the employee shall spend at least 10 minutes on other tasks.

(b) Employees who spend a majority of time working at a VDT are encouraged to visit an eye care professional annually. The Employer will pay for any portion of the fee for this annual visit not covered by medical or extended health benefits (this fee may not include the annual health care deductible amount as per the Extended Health Benefits).

(c) Pregnant employees shall have the option not to continue to work with video display terminals which use cathode ray tubes. When a pregnant employee chooses not to work with such video display terminals, if another position at the same or lower level is vacant which does not require such use of VDT and for which they are qualified they shall be reassigned to such work and paid at the rate of pay for that position.

(d) Where work reassignment in (c) above is not available, a regular employee will be considered to be on leave of absence without pay until they qualify for maternity leave.

(e) Where employees are on leave of absence without pay pursuant to (d) above, and opt to maintain coverage for medical, dental, extended health, group life, and long-term disability plans, the Employer will continue to pay the Employer's share of the required premium, subject to the same conditions as per Clause 21.5 (Health and Welfare Benefits)(a).

22.4 Equipment Standards

The Employer shall ensure that new equipment shall:

- (a) have adjustable keyboard trays and screens;
- (b) meet required radiation emission standards established by the Ministry of Labour or Workers' Compensation Board.

The Employer shall provide sufficient number of standard earthquake preparedness kits in each office location.

The permanent Health and Safety Committee or the Joint Committee shall review and make recommendations to ensure that the lighting and other standards recommended by the Ministry of Labour, Occupational Environment Branch or the Workers' Compensation Board, are being met.

22.5 Personal Safety

If an employee believes they are physically at risk because of threats made by a debtor or creditor, an incident report shall be submitted immediately to the Enforcement Manager or, in their absence, to the Senior Manager. The employee shall be advised within 48 hours (sooner if the circumstances warrant it) of what action has or will be taken and what steps they should take, if any.

22.6 Unsafe Work Conditions

No employee shall be disciplined for refusal to work on a job which does not meet the applicable standards established pursuant to the *Workers Compensation Act* or other situation which a member of the Joint Health and Safety Committee reasonably believes is unsafe, or which an authorized staff person of the Workers' Compensation Board has stated is unsafe.

No employee shall be disciplined for refusing to work when confronted by a person who, in the employee's reasonable estimation, poses an immediate hazard to the employee's physical safety.

22.7 Injury Pay Provision

An employee who is injured on the job during working hours and is required to leave for treatment or is sent home for such injury shall receive payment for the remainder of the shift without deduction from sick leave.

22.8 Transportation of Accident Victims

Transportation to the nearest physician or hospital for employees requiring medical care as a result of an on-the-job accident shall be at the expense of the Employer.

22.9 Supply and Maintenance of Equipment

The Employer will furnish and maintain all equipment, furniture, machinery and supplies required by employees in the performance of their duties. Employees shall not suffer any loss in salary in the event that they cannot carry out their normal duties by reason of the Employer failing to furnish or properly maintain equipment, machinery or supplies. Where an employee suffers damage to their attire due to improperly maintained equipment, the Employer shall reimburse the employee for repair or, where repair is inappropriate, replacement of the clothing article, providing the Employer had been advised of the problem prior to the incident or could reasonably be expected to anticipate that such a problem was imminent.

22.10 Smoking Policy

Smoking is prohibited at all times in all parts of the Employer's office facilities. Employees who contravene this policy shall be disciplined accordingly.

22.11 Workplace Violence

(a) It is recognized that at certain worksites or in certain work situations employees may be at risk of verbal abuse from clients including threats of physical violence.

- (b) Where such potential exists:
 - (1) employees at those worksites or in those work situations shall receive training in the recognition and management of such incidents;
 - (2) reasonable physical and procedural measures to protect employees shall be implemented.

(c) The local Occupational Health and Safety Committee shall be consulted regarding the curriculum of training and the applicable physical and procedural measures referred to in (b) above.

(d) Employees shall be informed, based on the existing knowledge of the Employer, concerning the potential for violence or verbal abuse of a particular client, subject to statutory limitation.

(e) Immediate critical incident stress debriefing and post traumatic counselling shall be made available for employees who have suffered as a result of work related physical violence. Leave required to attend such debriefing or counselling sessions will be without loss of pay.

22.12 Occupational First Aid

(a) The Union and the Employer agree that first aid regulations made pursuant to the *Workers Compensation Act* which are applicable to this employer, shall be fully complied with.

(b) Where the Employer requires an employee to perform first aid duties in addition to the formal requirements of the job, the cost of obtaining and renewing the appropriate level of certification required under (a) shall be borne by the Employer, and leave to take the necessary courses shall be granted with pay.

ARTICLE 23 - HARASSMENT AND BULLYING

23.1 Harassment and Bullying-Free Worksite

The Union and the Employer recognize the right of the employees to work in an environment free from harassment and bullying, and the Employer undertakes to discipline any person employed by the Employer engaging in the harassment or bullying of another employee. No manager, employee or contractor shall harass or bully any other manager, employee, contractor, or any other person with whom that person has contact as a consequence of their work for the Employer.

Once the Employer becomes aware of the alleged harassment or bullying complaint, the Employer shall take all reasonable steps to ensure the situation is not aggravated and to relieve the employee's distress.

23.2 Prohibition of Harassment and Bullying

Harassment and bullying may be physical, verbal or sexual in nature and all forms are equally unacceptable. Harassment and bullying include any inappropriate conduct or comment by a person towards another person that the person knew or reasonably ought to have known would cause the other person to be humiliated or intimidated. Harassment and bullying may include, but is not limited to, threats or intimidation, physical assault, deliberate gestures, comments, questions and representations.

Inappropriate use of managerial/supervisory authority takes place when a person who supervises or is in a position of authority exercises that authority in a manner which serves no legitimate work purpose and which ought reasonably be known to be inappropriate.

Inappropriate use of managerial/supervisory authority does not include actions occasioned through the exercise in good faith, of the Employer's managerial/supervisory rights and responsibilities.

A manager, employee or contractor of the Employer agrees, as a condition of employment, to refrain from any form of harassment or bullying as described in this article.

23.3 Disciplinary Action

Any manager or employee deemed by the Employer to have been guilty of harassment or bullying is subject to disciplinary action, up to and including dismissal.

23.4 Limitation of Definition

(a) The use of professional, objective and non-demeaning language to explain, instruct, correct, counsel or supervise does not constitute harassment or bullying. The definition of harassment and bullying is not intended to restrict the Employer's managers and supervisors from exercising their normal supervisory responsibilities, including disciplining of persons reporting to them.

(b) Pressing debtors for payment does not constitute harassment or bullying for the purposes of this article, providing the debtor is treated objectively and courteously, is given factual and accurate information, and is not subjected to any demeaning personal remarks about their behaviour.

23.5 Response to Harassment or Bullying

Any employee who believes that they have been the victim of verbal, or physical harassment or bullying shall use the grievance process to seek redress. If the person against whom the employee is complaining reports to the CEO, then the grievance is to be filed directly with the CEO. If the person is complaining against the CEO, then the grievance is to be filed directly with a designated member of the Employer's Board of Directors.

23.6 Confidential Investigation

In the course of responding to a grievance alleging harassment or bullying, all steps will be taken to ensure that the matter is properly investigated and that all persons involved in the handling of a complaint shall hold in the strictest confidence all information of which they become aware; however, it is recognized that various officials of the constituent group(s) and the Employer will be made aware of all or part of the proceedings on a 'need to know' basis and, as may be necessary, to divulge any information as a result of any formal legal action which may subsequently result.

23.7 Sexual Harassment

(a) Sexual harassment shall be defined as any unwelcome comment or conduct of a sexual nature that may detrimentally affect the work environment or lead to adverse job-related consequences for the victim of the harassment. Sexual harassment may include, but is not limited to, unwanted physical contact, suggestive remarks or other verbal abuse with a sexual connotation, emphasizing sexuality, sexual identity or sexual orientation, compromising invitations, repeated or persistent leering at a person's body, demands for sexual favours, or sexual assault.

(b) Whenever possible, the employee alleging harassment has the right to discontinue contact with the alleged harasser, without incurring any penalty. In cases where harassment may result in the transfer of an employee, where possible, it shall be the alleged harasser who is transferred. The employee who is alleging harassment shall not be transferred against their will.

23.8 Complaint Procedure for Sexual Harassment

(a) The employee shall file a written complaint within 30 days of the alleged occurrence with the Senior Manager.

(b) The Senior Manager shall immediately advise the staff representative.

(c) If the complaint is against the Senior Manager, the complaint will be filed with the CEO. If the complaint is against the CEO, the complaint will be filed with a designated member of the Employer's Board of Directors.

(d) The investigation shall be concluded within seven days of the complaint being received.

(e) The Employer may request an extension for the investigation period from the staff representative or designate. The extension, if granted, shall not in any event, be longer than 14 days from the date of the written complaint.

(f) The complainant will be given the option of having the staff representative or designate present as an observer at the meeting(s) at which the complainant is present.

(g) The CEO or designate who has investigated the incident shall complete a written report within three working days of completion of the investigation.

(h) The staff representative and complainant shall be apprised of the action(s) to be taken.

(i) Where either party to the proceeding is not satisfied with the CEO's or designate's response, the complaint will, within 30 days, be referred to Clause 10.11 (Expedited Arbitration).

(j) Where the complaint is determined to be frivolous and/or vindictive in nature, the Employer will take appropriate disciplinary action.

ARTICLE 24 - TECHNOLOGICAL CHANGE

24.1 Process

The Employer will continue to develop its computer and telecommunications system to streamline the work. If this ongoing development makes a particular position with an incumbent redundant, the Employer shall:

(a) advise the Union at least 90 days prior to any action of the nature of the technological change, which employees will be affected, and how the Employer intends to deal with the employees affected;

(b) attempt to place the affected employee in another vacant position for which they are qualified; if the employee is not qualified but could reasonably be expected to be qualified within 60 days if training, study time or job experience were provided, the Employer shall make every effort to provide same;

(c) where such development described in (b) is required it shall be at the Employer's expense;

(d) if the employee has demonstrated that they can fulfil the duties of their new position within 60 days to the satisfaction of the Employer, the employee will be placed in that position permanently, except where the new position constitutes a promotion, in which case the normal probationary period shall apply;

(e) an employee placed into an available position as per this article shall not experience a reduction of salary and shall be paid as per Clause 13.5 (Salary Protection);

(f) if no vacant positions are available, the layoff and severance provisions as per Articles 13 (Layoff and Recall) and 14 (Severance Pay) will come into effect except that at least 90 days' notice of the layoff will be provided to the Union and the employee(s) affected.

ARTICLE 25 - RECRUITMENT AND SELECTION

25.1 Recruitment and Selection

Principle

The Employer wants to fill as many positions as possible through internal promotion and is committed to assisting employees who wish to prepare themselves for promotion.

(a) The Employer selects candidates for position vacancies, whether full-time or temporary, based on the knowledge, skills, ability, experience and education of the applicant. The Employer will accept an internal applicant's equivalent combination of education and experience in place of a formal educational credential. The successful candidate will be the person best qualified for the position. Wherever possible, preference in hiring shall be given to internal applicants. Where internal candidates are equally well qualified, service seniority shall apply.

(b) At least once each calendar year, the Employer will offer a 'promotion preparedness' training workshop for interested members of the bargaining unit provided that there are at least five employees at each location that have agreed to attend. The curriculum will include classroom training,

identification of skills and attributes which contribute to success in a position, suggested reading and practise exercises, and may include an opportunity to work with an individual holding the desired position for a time and on a schedule to be mutually agreed. Training may be offered outside of office hours, and will be without compensation. On the job training will be at the Employer's expense. Participation in this workshop is not a condition for promotion nor will it be a factor in selection or rejection.

(c) The Employer will use its best efforts to establish a 'relief pool' of persons who meet or have met the qualifications for bargaining unit positions and are available to perform temporary or casual work. Where a short-term vacancy is created, an available person from the relief pool shall be placed in the position, subject to the circumstances and availability of resources.

(d) When a vacancy arises which will increase the workload of existing employees, the Employer shall, within 14 days, advise employees of what it is doing to deal with the situation. If it is not possible or practical to backfill a position, the Employer shall consult with the affected employees to see what can be done to minimize the impact. Employees who are required to complete some additional work, due to a temporary vacancy, will receive ongoing, clear and reasonable direction from their supervisor regarding work priorities and expectations.

(e) An eligibility list by component shall be maintained by the Employer and those on the list shall be notified of their standing, including their position on the list and any subsequent change(s) to their position. The Employer shall post notification of the component(s) for which an eligibility list is being maintained. Where a dispute arises regarding the eligibility list, the Employer shall provide a union representative with access to the list. Subsequent vacancies for the same position and work location shall be filled from the eligibility list (except when a placement is made as per Clauses 39.6 [Right to Position] or 25.8 [Unable to Place into Vacancy Because of Operational Requirements]) without posting or holding a competition for a period of 12 months from competition end date. The start date of the eligibility list is the date the successful candidate commenced their duties after winning the competition. An offer shall be made from the eligibility list based on the ranking in the original competition. The Employer may hold a competition at any time while an eligibility list is still in effect, but no person shall be hired from any new eligibility list until the old list has expired. A list expires either one year from its start date or when every person on the list has been placed or has turned down a reporting date, whichever comes first. Any candidates remaining on the eligibility list at the time of expiration who have not rejected permanent employment, will move to a new list in the same relative order as the previous expired list. This is subject to the successful candidate's completion of at least 40 continuous working days in an acting or temporary capacity in the position for which they are qualified. All others working must apply.

25.2 Postings

(a) The Employer shall, at the time of recruitment for any full-time position or temporary bargaining unit position more than 90 days in duration (except when a position is filled pursuant to Articles 13 [Layoff and Recall] or 24 [Technological Change]) post a notice on the appropriate bulletin board at each of its offices, or on its web site, for at least one week prior to closing. The Employer will also email the posting to all employees at least one week prior to closing. Employees wishing to apply for another position shall submit the appropriate application form to the designated person. The Employer shall review all applications and shall interview all those persons who meet the qualifications for the new position. Qualifications shall include knowledge, education and training, skills and abilities and personal suitability. Unsuccessful internal applicants, upon written request within five working days of being notified that they would not be interviewed or that they were not the successful candidate, shall receive written reason(s) for unsuitability no later than 10 working days from date of request.

(b) Job posting notices shall contain the following information: job title, salary, skills and essential qualifications (as per the job description), if car use is required and whether the position is included in the bargaining unit.

(c) The Employer will make every reasonable effort, subject to its organizational and financial resources, to fill all temporary or permanent vacant positions as soon as possible.

25.3 Promotion While on Probation

No continuous employee can apply for a position at a higher pay level until the employee has completed five months of the probation period.

25.4 Accuracy of Information

The authenticity and accuracy of the information provided in the job application, attached resumes, letters of reference, degrees, diplomas or certificates, or any other document which is submitted by the employee as part of the selection process, is a condition of employment. Subsequent discovery that the information supplied is false and/or deliberately misleading shall constitute grounds for disciplinary action. If the false or misleading information would have meant the employee would not have been hired or promoted, the employee shall be dismissed.

25.5 Job Sharing

Job sharing arrangements may be entered into on a case-by-case basis where the terms and conditions are mutually acceptable to the Employer, the Union and the persons wishing to job share.

The parties understand that the two individuals will fill a single position, and that they are not considered to be filling two part-time positions. The position shared is that of the employee with the greater service seniority. The parties acknowledge that the details of a job sharing arrangement may require modification, and they are committed to a flexible approach to job sharing arrangements, as long as this does not compromise the ability of the Employer to meet its operational requirements.

(a) Instituting a job sharing unit must be on a voluntary basis by the present incumbent. It is understood that job sharing units will not necessarily be posted.

(b) The two employees with the same job classification will initiate a job sharing arrangement by presenting such a proposal to the Employer. The proposal shall be in a format set out by the Employer.

(c) The initial six months of a job sharing arrangement will be deemed to be a trial period. If the job sharing arrangement is unworkable and/or unacceptable to any of the parties, the arrangement shall be terminated at the end of the six month trial period. Both employees shall return to their previous positions.

(d) Once the initial six-month trial period is completed, the arrangement may be terminated upon 30 days' notice by mutual agreement of the two incumbents or by the Employer.

(e) (1) The two incumbents in the job sharing arrangement must share wages, benefits and conditions as provided in the collective agreement to a combined maximum equal to that which would be received if one employee occupied the position. The Employer will contribute towards the cost of benefits premiums an amount equal to what the Employer would have paid if the position was not shared. The two incumbents will be responsible for paying the balance of the total benefits premiums.

(2) The incumbents shall not be entitled to flex days under Clause 15.3 (Flex Days and Compressed Time). The Employer will adjust the salary of the position by the same number of

hours to which the position would have been entitled due to flex days, at the rate of pay in effect at the time the flex days were earned.

(3) Each of the incumbents shall accrue seniority for hours they worked during the term of the job sharing arrangement.

(4) Each of the incumbents shall be entitled to bereavement leave pursuant to Clause 20.1 (Bereavement Leave) as though they each occupied a full-time continuous position.

(5) In the event that either or both incumbents become entitled to severance pay pursuant to Article 14 (Severance Pay), each of them shall be paid one-half of the severance pay which they would have received if they were full-time employees.

(f) The scheduling of the job sharing unit shall be done at the local level and the number of hours must average to half-time on a monthly basis for both employees.

(g) (1) If the job sharing arrangement is terminated by the two incumbents at a time other than during or at the conclusion of the trial period, the incumbent with the greater service seniority will be given the opportunity to work full-time. This incumbent has one week in which to exercise their option under this provision. If they accept the full-time position, the incumbent with the lesser service seniority will be offered a vacant position in the same classification in the same geographical location if any such position is available. If no position is available or the incumbent with the lesser service seniority declines to accept the vacant position, they will be deemed to have resigned.

If the incumbent with the greater service seniority declines to accept the full-time position, they will be deemed to have resigned. The incumbent with the lesser service seniority will then be offered the full-time position. They have one week in which to exercise their option under this provision. If they decline to accept the full-time position, they will be deemed to have resigned.

(2) If the job sharing arrangement is terminated by the Employer at a time other than during or at the conclusion of the trial period, the incumbent with the greater service seniority will be given the opportunity to work full-time. This incumbent has one week in which to exercise their option under this provision. If they accepts the full-time position, the incumbent with the lesser service seniority will be offered a vacant position in the same classification in the same geographic location, if any such position is available. If no position is available the incumbent will be subject to Article 13 (Layoff and Recall) of the collective agreement (Layoff and Recall). If the incumbent with the lesser service seniority declines to accept the vacant position, they will be deemed to have resigned.

If the incumbent with the greater service seniority declines to accept the full-time position, they will be deemed to have resigned. The incumbent with the lesser service seniority will then be offered the full-time position. They have one week in which to exercise their option under this provision. If they decline to accept the full-time position, they will be deemed to have resigned.

Notwithstanding Article 13 (Layoff and Recall), all employees participating in a job sharing arrangement regardless of the years of service seniority, are entitled to 30 days' notice of layoff where a layoff notice is issued pursuant to 7(b).

(3) In the event that either incumbent in a job sharing arrangement is unable to perform the job for an extended period of time, as a result of a MAP or other leave, the remaining incumbent's hours will be increased to full-time unless another employee can fulfill the absence until such time as a return to work occurs. It is understood that an increase in hours to one partner shall not be

permanent unless agreed to by both partners. The Supervisor will give as much notice as possible to the partner before increasing a partner's hours of work.

(4) In the event that the position held by the two incumbents in a job sharing arrangement is made redundant, both incumbents will be subject to Article 13 (Layoff and Recall) of the collective agreement.

25.6 Job Exchanges

A job exchange with an employee in another agency, company or organization is permissible but is subject to case by case approval of the Employer, the employee concerned and the Union.

25.7 Lateral Transfer Policy

(a) An employee who has completed a minimum of four years of service in a particular position shall be eligible for a lateral transfer into the same position in another officer where there is a vacancy for the same position (continuous not temporary) under the following circumstances:

(1) The employee's last performance appraisal was wholly satisfactory

(2) The employee has no disciplinary action in progress and no letters of reprimand or suspension are on the personnel file.

(b) An employee who is eligible shall advise their Senior Manager in writing of their desire to be granted a lateral transfer. The Senior Manager shall confirm their eligibility and if all is in order, will advise their counterpart in the office the employee wished to move. Any existing eligibility list shall be amended so that the transferring employee's name is now at the head of the list. All internal employees whose names are on the list shall be given a copy.

(c) The lateral transfer cannot take place until there is a continuous position of the same job type in the office to which the employee wishes to move to. If the employee leaves their job in the original office prior to there being a vacancy for them in the new office, they shall be considered to be on leave without pay and may continue the benefits that they are still eligible for (this does not include Short-Term Disability (STD) or long-term disability which are not available to the employee in these circumstances) under the terms of our contract with the insurer, provided they pay 100% of the monthly premium cost (this includes what would normally be the Employer's share) within the 10 days of receipt of invoice. Employees who do not reimburse the Employer within 10 days of receipt of the invoice may have their benefits cancelled without further notice by the Employer. The maximum period of time for leave of absence without pay under this article is 12 months. If there has not been a vacancy during that time the employee will be deemed to have voluntarily resigned and shall not be eligible for any severance pay or other benefits.

(d) An employee who has been granted Leave Without Pay as per (c) above, may occupy a vacant temporary position in the new office for which they are qualified until such time as the temporary position expires or they are placed in a vacant position as per the eligibility list. However, the following shall apply in such cases of temporary placement:

(1) Their existing position and service seniority shall count only for their placement on the salary grid (if it is the same position as the one they had) and the entitlement for holiday time and benefits premiums proration. They may not use their existing service seniority for holiday selection or bumping while they are in the temporary position.

The Employer may grant, at its discretion, a lateral transfer to an employee for compelling (e) compassionate or medical grounds. The Employer shall only grant such lateral transfer where the reasons clearly indicate that this is beyond the employee's reasonable control and/or the consequences of not granting the transfer would result in undue hardship to the employee. If the Employer grants such lateral transfer, Clauses (b) - (d) shall apply.

25.8 **Unable to Place Into Vacancy Because of Operational Requirements**

Where an existing continuous employee who is on the eligibility list for a higher classification cannot be placed into the vacancy to which they would be entitled because they cannot be released from their current position due to operational requirements, the following procedure shall be followed:

The employee shall be placed into the new position to which they are entitled within five days of (a) the vacancy becoming available;

(b) The employee shall then be seconded back into their old position but shall continue to be paid at the rate of their higher position, and they shall earn position seniority on their higher position while they continue in their former position;

(c) The probationary period for the new position shall not start until the employee actually commences their duties in the new position.

ARTICLE 26 - PROMOTION

26.1 **Interview Expenses**

A continuous employee who applies for a posted position which would constitute a promotion, and who is not on leave of absence without pay, shall have their authorized expenses paid for attending a job interview outside the applicant's geographic location, as defined in Appendix II. Travel shall be in the form directed by the Employer.

26.2 Transfers

(2)

No employee shall be required to transfer from one geographic location to another as per Appendix II against their will.

26.3 **Probation on Promotion**

(a) If a continuous employee is promoted to a higher paying position they shall be placed on probation for a period of six months. Upon successful completion of probation the employee shall be confirmed in the position. In the event that the employee proves unsatisfactory in the new position during the probationary period, or if the employee requests in writing to return to their former position during the probationary period, the employee shall be returned to the former position or shall be given another position for which they are qualified in the same geographical location at an equivalent salary to the position they were promoted from, except where the employee was terminated for cause. There shall be no loss of service seniority; any position seniority earned prior to the promotion shall be restored upon such demotion. Any other employee promoted or transferred because of the rearrangement of positions shall also be returned to their former position without loss of seniority.

(b) Probation on promotion does not affect an employee's benefit entitlements. Service seniority is also not affected, just position seniority.

(c) Where an employee is absent in excess of 40 hours on sick leave during the period of probation, the period of probation shall be extended by an equivalent numbers of hours.

26.4 Notification to Employee and Union

The Employer shall advise all internal interviewees in a competition for a bargaining unit position of the name of the successful candidate within 10 days of that candidate being selected.

(a) The Employer is committed to providing feedback to internal candidates. Feedback will be provided in the spirit of supporting employee development, progress and growth. All candidates will be given the opportunity to receive feedback. The Employer shall provide the feedback within five days of identifying a successful candidate. On employee request within five days of the feedback session, the Employer will provide the reason the employee was not selected for the vacancy in writing within five days of the request. These timelines may be extended by mutual agreement. No appeal or grievance may be submitted if the employee has not requested written reasons as to why they were not selected. The Union shall be notified of all appointments, layoffs, transfers, recalls and terminations of employment.

(b) Any person who was placed on an eligibility list will receive verbal feedback, upon request. The Employer shall provide feedback within five days of the request.

26.5 Salary on Promotion

Where an employee is promoted to a position with a higher starting salary, the employee shall be placed at the first step of the new position which would result in a salary increase of at least 4%.

26.6 Right to Grieve

(a) An employee may grieve the decision of the Employer at Step 3 of the grievance procedure as per Article 9 (Grievance Procedure), within five working days of receiving the reasons they were unsuccessful, as per Clause 25.2 (Postings) and Clause 26.4 (Notification to Employee and Union). If the grievance is upheld by the Employer or an arbitrator, the only remedy is that the competition will be reposted.

(b) Where a grievance has been filed under this article, no permanent transfers or placements related to the position in question shall be made until the grievance is resolved.

(c) Time limits under this clause may be extended in writing by mutual agreement.

26.7 Acting Prior to Promotion

When an employee is confirmed in a position which constitutes a promotion, and that employee was acting in that position immediately prior to confirmation for a period over 10 days, the following shall apply:

(a) the time spent in the acting position shall be credited for purposes of position seniority date and placement on the salary grid;

(b) any time over two months spent in the acting position shall be credited towards the six month probationary period.

(c) where an employee returns to an acting position within 12 months, the employee shall be credited with the previous time spent in that position for the purposes of determining placement on the pay grid.

ARTICLE 27 - PROBATION

27.1 Probationary Period

(a) The probation period shall be six months for continuous full-time employees. Continuous part-time employees shall have a prorated probationary period equivalent to six months full-time work. Where an employee is absent in excess of 20 hours on sick leave during the period of probation, the period of probation shall be continued by an equivalent number of hours. The Employer may reject any probationary employee for just cause. Any rejection during probation shall not be considered a dismissal for the purpose of Article 11 (Disciplinary Action) of this agreement. The test of just cause for rejection shall be a test of suitability of the probationary employee for continued employment in the position to which the employee has been appointed. If a promoted continuous employee is rejected on probation, Clause 26.3 (Probation on Promotion) shall apply.

(b) At the end of three month's continuous employment, a probationary employee shall be given an interim evaluation which will provide written feedback from the employee's supervisor about the employee's work performance, attitude and work relationships.

(c) At the end of the six month probationary period, the employee shall be given verbal and/or written notice that they have:

(1) satisfactorily passed the probationary period and are now confirmed in the position as either a continuous employee, or, if a temporary employee for the balance of the temporary term, or

(2) had their employment terminated.

If the employee has not been given or sent such notice within 10 working days after the expiry of the probationary period, that employee shall be deemed to have successfully completed the probationary period.

(d) An employee who has been rejected on probation shall be paid within five working days for all outstanding wages, compensatory time, overtime and holiday time. Any keys and/or equipment issued to the terminated employee must be returned.

27.2 Right to Grieve

When an employee feels they have been aggrieved by the decision of the Employer to reject the employee during the probationary period, the employee may grieve the decision pursuant to the grievance procedure outlined in Article 9 (Grievance Procedure) of the agreement commencing at Step 3.

ARTICLE 28 - PERFORMANCE APPRAISAL

28.1 Probationary Employees

All employees who are on probation shall receive a performance appraisal at the conclusion of the third and fifth month of employment. The purpose of these appraisals is to assess the employee's progress in learning and performing their duties with a specific focus on the employee's suitability for permanent employment in the position occupied.

28.2 Non-Probationary Employees

Employees who are not on probation shall receive an annual performance appraisal. If an annual performance appraisal is not received within 30 working days of employee's annual appraisal date it shall be deemed completed for that year.

The Employer shall keep a record of the appraisal dates of all employees and each employee shall be made aware of their appraisal date.

To establish the appraisal date, the Employer shall use the anniversary of the last appraisal completed or the anniversary of the employee's position confirmation date, whichever is mutually agreed upon.

The purpose of the appraisal is to provide information to the employee about performance in carrying out job duties, relationships with supervisors and co-workers, and attitude. The appraisal will indicate strengths and will specify any areas which need improvement. Specific performance, behaviour or training objectives for the next evaluation period will be outlined and the extent to which they were met will be assessed in the next performance appraisal. The performance appraisal is not a disciplinary document although it may make reference to disciplinary action taken during the appraisal period. Subsequent disciplinary action may also refer to the performance appraisal when relevant.

28.3 Procedure

The supervisor shall give the employee a draft copy of the appraisal and shall discuss it with the employee. Changes may be made to the draft based on the discussion and that shall constitute the final version of the appraisal. If no changes are deemed necessary by the supervisor, the draft copy shall be the final version. If the employee is absent from work the appraisal shall be held until the employee returns to work except where the purpose of the appraisal is to advise the employee of their status at the conclusion of the probationary period.

28.4 Right to Review

When an employee receives a performance appraisal they will be given an opportunity to take it away overnight and review it and to indicate agreement or disagreement with the appraisal by signing the appropriate line. The signed appraisal must be received by the Employer within two workings days except where longer is agreed to by mutual consent of the supervisor and the employee. If the appraisal is not signed and returned within this period a note shall be made on the file copy of the appraisal. That appraisal shall be regarded as valid and shall be used as if it had been signed by the employee, except that the unsigned appraisal may not be grieved by the employee.

28.5 Right to Grieve

A performance appraisal may be grieved but only if the employee signs that they disagree with the appraisal and where work performance is assessed as less than satisfactory.

ARTICLE 29 - PROFESSIONAL CONDUCT/CONFIDENTIALITY

29.1 Conflict of Interest

Employees must advise their supervisor in writing using the standard form as soon as they are aware that their job responsibilities might involve them with a case to which they are, or were, a party or where either the creditor or debtor is personally known (this does not include one-time meetings) to that employee. Failure to disclose a potential conflict of interest may be disciplinable. The supervisor shall discuss the situation with the employee and if the supervisor believes that there is a potential conflict of

interest, the case will be assigned to another employee. If the supervisor and the employee cannot agree over whether the case should be reassigned, the Senior Manager shall make the final decision.

Once it has been determined that a conflict of interest exists, the employee may not access any information about that case or discuss it with any other co-worker who might be in a position to access information on the case, or influence the conduct of the case. If the employee is a recipient or payer, they shall communicate with the Program about their case in writing or by talking to a manager. Employees who violate these requirements may be disciplined and may be dismissed.

29.2 Personal Use of Employer's Resources

Any employee who uses resources of the Employer or access provided to the Employer by the provincial government for non-work related purposes without express permission is guilty of a serious breach of professional conduct, and will be disciplined and may be dismissed.

29.3 Oath of Confidentiality

- (a) Employees are required to swear an undertaking that all information about any aspect of any case will not be revealed to anyone not authorized in the Communications Policy to receive that information.
- (b) Any employee who violates their undertaking of confidentiality either intentionally or through gross irresponsibility may be subject to immediate dismissal.

29.4 Inappropriate Use of Information

Any employee who obtains information by unauthorized means or uses information about a creditor or debtor for any purpose other than the regular performance of their assigned responsibilities shall be immediately dismissed without further notice.

29.5 Due Care

All employees of the Employer shall exercise all reasonable care to ensure files and other documents or information within their control are secure.

ARTICLE 30 - PAYMENT OF WAGES AND ALLOWANCES

30.1 Equal Pay

The Employer shall not discriminate by employing a person of one sex, sexual orientation or gender identity or expression for any work at a rate of pay that is less than the rate of pay at which a person of any other sex, sexual orientation or gender identity or expression is employed for the same or substantially the same work.

30.2 Paydays

Employees shall be paid on the 15th and last day of each month. When a payday falls on a Saturday, Sunday or statutory holiday, the Employer agrees to deposit the employee's paycheque in the employee's bank account by 3:30 p.m. on the last scheduled shift prior to the payday.

30.3 Rates of Pay

(a) Employees shall be paid in accordance with rates of pay set out in Appendix I (Salary Grid) by direct deposit.

(b) Converting One Week of Vacation Entitlement

Each year prior to 5 p.m. January 15th any employee who has not already done so may elect to irrevocably convert annual vacation entitlement (Clause 18.2 [Vacation - Entitlement]) to salary subject to the following conditions:

(1) may only be converted in 37.5-hour blocks and the maximum total conversion may not exceed 37.5 hours;

(2) the conversion will be effective April 1st of the same calendar year of the notice deadline in (b) above;

(3) each 37.5-hour block of time converted will result in the addition of 2% rounded to the nearest dollar to the employee's base pay effective April 1^{st} of the same calendar year of the notice deadline in (b) above.

(c) Converting Additional Vacation Entitlement

Each year prior to 5 p.m. January 15th any employee who has already converted, or who is at the same time converting, annual vacation entitlement pursuant to (b) above and who has not already converted the maximum additional vacation entitlement under this (c) may elect to irrevocably convert additional annual vacation entitlement (Clause 18.2 [Vacation - Entitlement]) to salary subject to the following conditions:

(1) will only be converted in 7.5-hour blocks;

Service Seniority	Maximum Additional Conversion	Cumulative Maximum Additional Conversion	Total Hours Available For Conversion
37 months to 72 months	+ 22.5 hours	22.5 hours	60 Hours
73 months or more	+ 15 hours	37.5 hours	75 Hours

(2) the additional conversion will not exceed the following:

(3) the conversion will be effective April 1st of the same calendar year of the notice deadline in (b) above;

(4) each 7.5-hour block of time converted will result in the addition of 0.4% rounded to the nearest dollar to the employee's base pay effective April 1^{st} of the same calendar year of the notice deadline in (b) above.

30.4 Substitution Pay

When an employee is formally appointed to substitute in, or perform most of the principal duties of a higher paying position for more than five continuous days they shall receive the rate for the higher paying position (as per Clause 26.5 [Salary on Promotion]), effective the first day. Employees who go onto short-term disability leave, special leave or any other paid leave of absence (except as per Clause 18.5 [Vacation Pay]) from an acting position will be paid at the basic rate of pay they received prior to substituting in the higher position.

30.5 Pay on Temporary Assignment

An employee temporarily assigned by the Employer to a position with a rate of pay lower than their regular rate of pay shall maintain their regular rate of pay.

30.6 Salary Protection

Where an employee's salary is frozen because it has been protected as per Clauses 13.5 (Salary Protection) and 24.1 (Process), the employee shall not receive negotiated salary increases until the salary of their new classification equals or exceeds the salary which the employee is receiving.

That employee shall receive the full, negotiated salary increases for the new classification thereafter.

30.7 Vehicle Allowance

(a) Vehicle allowance for all distances required by the Employer to be travelled on the Employer's business in the employee's personal vehicle shall be paid as follows:

(1) At the maximum reasonable per kilometre employee allowance rate set by the Canada Revenue Agency.

(2) Mileage shall be measured from the office to the place the employee is conducting business; where travel is directly from home without first going to the office and such has been authorized by the Employer, door to door mileage is claimable but must be noted as such on the travel claim form.

(b) Ownership of a personal vehicle shall not be a condition of employment.

(c) When an employee is required to perform duties requiring vehicle transportation and the employee chooses not to use their own car, the transportation will be provided by the Employer at the Employer's expense.

30.8 Meal Allowance

(a) Meal allowances shall be paid when the employee is required to be more than 40 kilometres away from their assigned place of work at the rates noted in the chart below and under the following circumstances:

(1) *Breakfast* - When the employee left home before 7:00 a.m. or is away from home on travel status overnight;

(2) Lunch - When the employee is outside of the 40-kilometre mileage range between the hours of 12:00 p.m. and 1:30 p.m., or where the employee is participating in a committee meeting or training event where a majority of other participating employees are on travel status;

(3) *Dinner* - When the employee is away from home on travel status overnight, or arrives home after 6:30 p.m.

Meal	Rate	April 1, 2022	April 1, 2023	April 1, 2024
Breakfast	\$12.75	13.26	TBD	TBD
Lunch	\$14.65	15.24	TBD	TBD
Dinner	\$24.65	25.64	TBD	TBD

(b) Meal allowances may not be claimed for any meal which was provided at the Employer's expense or is included in the price of a ticket, or accommodation/conference package.

30.9 Accommodation and Other Expenses

(a) Accommodation will be provided at actual cost at hotels approved by the Employer and where possible, booked in advance by the Employer. Private accommodation will, where substituted for authorized overnight accommodation in a hotel, be reimbursed by a gratuity payment of \$53 per night.

Every employee shall have the right to a single room except where the travel is to participate in an overall Office, Program or Position Conference.

(b) Other legitimate travel costs shall be paid for at cost where noted on the expense form and authorized as allowed by the Employer.

30.10 Personal Charge Accounts

No employee shall be required to have or use any personal charge account for the purpose of carrying out the Employer's business.

30.11 Travel Advance

An employee required to travel more than 40 kilometres from the worksite on the Employer's business may request and shall be entitled to receive a travel advance sufficient to cover the estimated amount of claimable expenses. An employee who receives a travel advance shall submit a travel claim within five working days after the completion of the travel, noting all travel expenses and, if any balance is owing to the Employer, shall attach a cheque or cash.

30.12 Expense Form

Expenses, claims and accounting of all travel accompanied by appropriate receipts shall be completed by employees on a form provided by the Employer for this purpose within five days of completing the travel. Travel expenses shall be reimbursed by the Employer no later than 30 days from date the claim form is submitted by the employee.

30.13 Travel Accounts

Wherever reasonable and practicable, the Employer will arrange for business accounts to be opened with appropriate transportation companies, hotels and other agencies, to reduce the necessity of staff carrying with them large quantities of cash while travelling on employer business.

30.14 Transportation for Employees

The Employer agrees to pay the cost of taxi transportation of any employee required to travel to home from work between the hours of 8:30 p.m. and 7:00 a.m. Any employee who lives more than 10 km from the worksite and who does not have their own transportation is responsible for notifying the supervisor of this fact when asked to work late overtime so that overtime taxi transportation may be avoided.

30.15 Training Expenses

Where the Employer requires an employee to take a training program, the cost of training and normal meal, accommodation and travel expenses, if the employee is on approved travel status, shall be at the Employer's expense as per this article.

30.16 Relocation Expenses

(a) Employees who have to move from one geographic location to another after winning a competition or who take a lateral transfer shall be entitled to \$530 transfer expense providing such relocation requires a change of residence. Employees are eligible to receive this transfer expense only once in their employment period with the Employer.

(b) If an employee is required by the Employer to move from an office in one geographical area to one in another geographical area which is at least 40 kilometres apart, and such relocation requires a change of residence, the Employer shall pay the following relocation benefits:

(1) *Moving* - The costs of moving the personal household effects of the employee and their immediate family who normally reside with the employee up to a maximum of \$3180; this shall not cover any structures, vehicles, non-household machinery or equipment, boats, etc. The employee shall use movers approved by the Employer. The cost of packing and unpacking will be included. Employees who wish to undertake their own moving may negotiate an amount to be paid in lieu. Additional insurance coverage beyond the liability of the movers is the responsibility of the employee.

(2) *Real Estate and Legal Fees* - Any commission actually paid by a relocated employee who sells their primary residence which was occupied by the employee at the time the relocation offer was made, may be reimbursed up to a maximum level of \$4,245 Legal fees connected with the selling of the house and the conveyancing of a new house (but not mortgage fees) will be reimbursed at cost. Property transfer taxes or GST may not be claimed.

(3) *Relocation Allowance* - The employee may claim up to 10 days travel allowances meals, accommodation, as per this article, if temporary accommodation in a hotel is required prior to moving to the new residence, at the commencement of employment in the new location.

House-hunting Trip - Employees who are eligible for relocation expenses under Clause 30.16(b) will be granted up to three workdays prior to their move for the purpose of finding accommodation in their new location. Full travel allowances will be paid for both the employee and their spouse for up to three days for any house-hunting trip.

Relocation Leave - Employees required to relocate will be given three days paid leave at the time of the move only.

(c) Clause 30.16(b) does not apply if the relocation is due to the employee applying for a vacant position in another office or because the employee's position in one office was eliminated and the employee opted to take a vacant position in another office.

(d) The Employer may, at its discretion, provide financial assistance to an employee not eligible for the allowances specified under Clause 30.16(a) (b). The level of assistance to be provided is at the discretion of the Employer.

(e) An employee who receives relocation benefits as per Clause 30.16(a) (b) (d) and resigns within two years of the date of the move, shall reimburse the Employer upon receipt of an invoice within 28 days as follows:

(1) within 12 complete months of the move - 100% of all the expenses paid out to the employee under Clause 30.16(a)(b)(d).

(2) after 12 complete months of the move - 100% of all the expenses paid out to the employee under Clause 30.16(b)(d) less one-twelfth of those expenses for each full month worked in excess of 12 months since the move.

30.17 Training Allowance

The parties recognize that employees who are requested by the Employer to train new employees as part of those employee's orientation, and who do not need to use overtime as per Clause 16.7 (Overtime Due to Training New Employees), will be entitled to claim a trainer's allowance of \$4.25 per hour, subject to the following:

(a) The allowance will only be paid where the training is for a concurrent period of at least three hours in a one-day period.

(b) The trainer is not eligible for the allowance if they are training their replacement because the trainer has been seconded from a higher classification and is still being paid at the higher rate.

ARTICLE 31 - HEALTH AND WELFARE BENEFITS

31.1 Benefits Committee

(a) There shall be a benefits committee consisting of three members appointed by the Union from BCFMA employees and three members appointed by the Employer from management. The Committee shall meet at least once per year, at the Employer's expense, (except that no overtime or compensatory time off shall be paid where meeting hours and/or travel extend the normal workday), or more frequently by mutual agreement. Resource people may be assigned to the Committee by either party.

(b) The mandate of the Committee is as follows:

(1) to ensure the scope, features and cost of the benefits package are the best they can be given with the resources available;

(2) to review the performance of the existing carrier and all aspects of plan administration;

(3) to change the benefit details of the plan if the Committee believes it is in the best interest of plan participants and the Employer;

(4) to ensure that the carrier provides adequate information on performance and plan details;

- (5) to deal with other benefit related issues as they arise.
- (c) Decisions will be made on the basis of mutual agreement.

31.2 Commencement of Benefits

Continuous employees become eligible for all benefits after they have completed three months continuous service with the Employer (except standard medical which is covered from the first day of employment). Part-time employees shall be eligible for all benefits on the same basis as full-time employees providing they work a minimum of 18 ³/₄ hours per week, except where the employee is part of a job share when it shall be a minimum of 37¹/₂ hours every two weeks.

31.3 Group Plan

The Employer will maintain a group benefit contract for its employees which includes at least the following components:

- Life Insurance
- Accidental Death and Disability Insurance
- Extended Health
- Short-Term Disability
- Long-Term Disability
- Dental
- Employee Assistance Plan

The minimum level of coverage shall be in accordance with the contract between the Employer and the current carrier, a copy of which is filed with the Union. The Employer is not the insurer but has the obligation to pay the premiums as noted in Clause 31.4 (Benefits Review) and to insist that the insurer meet the obligations of the insurance contract.

31.4 Benefits Review

The Employer shall review the health and welfare benefits covered by the insurer's contract prior to July 1st of each year.

31.5 Cost of Premiums

The Employer and the employee shall split the overall cost of the premiums for the benefits noted in Clauses 31.3 (Group Plan) and 31.6 (Basic Medical Insurance) on an equal basis (except for the Employee Assistance Plan which is paid 100% by the Employer) if the employee has less than four years continuous full-time service. The ratio is 70% Employer, 30% employee for those employees with more than four years continuous full-time service (or equivalent for part-time employees, i.e., eight years half-time service). The premium cost for all benefits which may be added in the future shall be split on the same basis as described in this paragraph except if otherwise agreed.

31.6 Basic Medical Insurance

If the provincial government reinstates premiums for the Medical Services Plan (MSP) or equivalent basic medicare plan, continuous and temporary employees must be covered by the plan. This coverage must be through the Employer unless the employee provides proof the employee is already covered and does not wish to be covered under the Employer's group. The cost of premiums for those participating is set by the Province and shall be paid as per Clause 31.5 (Cost of Premiums) above.

31.7 Group Plan Coverage

All continuous and temporary employees must, after three months continuous service, participate in all the benefits noted in Clause 31.3 (Group Plan) subject to the restrictions noted in Clause 38.3(a). This shall not include Dental and Extended Health where the employee provides proof of similar coverage under another plan. An employee who is covered by another plan and elects to participate in the extended health and dental components of the Employer's plan shall not be permitted to later withdraw from these components of the plan.

31.8 Medical Examination

Where the Employer requires an employee to submit to a medical examination or medical interview, it shall be at the Employer's expense and on the Employer's time. This does not apply if it is the benefits insurer which is requiring the examination.

31.9 Legislative Changes

If the premiums paid by the Employer for any employee benefit covered by this agreement is reduced as a result of any legislative or other action, the amount of the saving shall be used to increase other benefits available to the employees, as may be mutually agreed between the parties.

*See Memorandum of Understanding (Clause 31.9 [Legislative Changes]) for clarity on the application of this clause.

31.10 Health and Welfare Plan

A copy of the benefit contracts with the carriers for the extended health care, dental and group life plans shall be sent to the President of the Union. A descriptive pamphlet will be given to every employee upon request.

31.11 Coverage on Leave of Absence Without Pay

Benefits as per Clause 31.3 (Group Plan) and 31.6 (Basic Medical Insurance) will be maintained upon request for employees absent on leave without pay in excess of 30 workdays in any calendar year providing the employee pays to the Employer 100% of the monthly premium within 30 days of billing by the Employer. The limits of the Insurer regarding the maximum time period for coverage shall apply.

31.12 Coverage Sick Leave

The Employer shall continue to pay its portion of the eligible benefits described in Clauses 31.3 (Group Plan) and 31.6 (Basic Medical Insurance), to a maximum of six continuous months, for employees on approved short-term disability and WCB (except where the premiums are paid by the Insurer) and sick leave providing the employee pays their share of the benefits within 30 days of billing by the Employer. The limits of the Insurer regarding the maximum time period for coverage shall apply.

31.13 Benefit Allowance

The Employer will contribute a monthly allowance of \$65 (this is taxable) to every full-time continuous employee.

This amount shall increase to \$150 as of the date of ratification and \$165 as of April 1, 2024.

Part-time, continuous employees shall receive this allowance on a prorated basis.

31.14 Retention Accounting

(a) On January 1, 1999, that portion of the Health and Welfare benefits covered by the benefits provider became subject to retention accounting. As part of the retention account, a Benefits Stabilization Fund was established from premiums paid by the Employer and employees.

(b) The Benefits Stabilization Fund shall be maintained at an adequate level to minimize the impact of significant yearly fluctuations in the premium rates.

(c) At the end of each benefit year (currently October 31st) any surplus in the retention account is repaid to BCFMA by the benefits provider. The surplus is split between BCFMA and the contributing employees in a proportion equal to the contributions paid by both parties. The employees' portion is divided amongst the employees as per the instructions of the bargaining Principals.

(d) In the event of wind up or transfer of the Benefits plan, the Benefits Stabilization Fund shall either be transferred to a new carrier and used for the same purpose, or shall be liquidated and paid out as per (c) above.

31.15 Employees Aged 65 and Over

Clauses 31.2 (Commencement of Benefits) through 31.14 (Retention Accounting) apply only to employees under the age of 70 with the exception of Long-term Disability coverage which remains terminated 120 days prior to age 65 and Life Insurance and Accidental Death & Dismemberment Insurance are reduced by 50% at age 65, consistent with the Employer's agreement with the benefits carrier.

ARTICLE 32 - PENSION PLAN

The Employer will continue its enrolment and employee participation in the Public Service Pension Plan. Eligibility and terms and conditions for the pension will be those contained in the Public Service Pension Plan Rules.

ARTICLE 33 - GENERAL CONDITIONS

33.1 Parking

The Employer agrees to pay parking costs on behalf of employees required to have their vehicle at work for use in the performance of their duties.

33.2 Political Activity

(a) Municipal and School Board Offices

Employees may seek election to municipal and school board offices provided that the duties of the municipal and school board office do not impinge on normal working hours as an employee of BC Family Maintenance Agency Ltd. (BCFMA) and providing that the exercise of those duties do not constitute what might reasonably be interpreted as a conflict of interest between the duties of the public office and the business of the Employer.

(b) Employees who are seeking public office or who are engaged in any form of political activity are bound by their oath of confidentiality and may not make public statements about the Employer or the Family Maintenance Enforcement Program which might reasonably be interpreted as jeopardizing the contractual relationship between BCFMA and the provincial government.

33.3 Copies of Agreement

The Union and the Employer desire all employees to be familiar with the provisions of this agreement and their rights and obligations under it. For this reason the Employer and the Union shall share the cost of printing this agreement on an equal basis. Each employee has a right to a copy of the collective agreement upon employment.

33.4 Personal Duties

The Employer and the Union agree that an employee will neither be required nor permitted to perform work not related to the business of the Employer. To this end, it is agreed that an employee will not be required or permitted to perform duties of a personal non-work related nature for supervisory personnel.

33.5 Employee Workload

Assignment of Work

(1) The parties agree that it is essential to ensure that all employees be advised of their job expectations, duties and responsibilities and that procedures are in place to meet work assignments.

(2) The Employer shall make every reasonable effort to ensure that the ongoing workload is manageable and equitably distributed.

(3) Where an employee is concerned that they cannot complete assignments and/or their work obligations, it is their responsibility to seek advice and direction from their supervisor. The supervisor will then provide direction to the employee, as necessary, on how to complete the assigned duties. This may include instructions on the priorities of the assigned duties.

(b) The short-term assignment of some of the duties normally performed by another employee in another position group for the purposes of providing coverage for those employees who are absent on flextime or various forms of leave, is a normal part of every employee's work responsibilities and is not grievable. This is not intended, however, to release the Employer from the obligation to pay substitution pay as per Clause 30.4 (Substitution Pay),

(c) It is not the Employer's intention to use this article to avoid regularizing position vacancies as soon as possible and practicable, but to deal with staffing or workload problems which are short-term or of an urgent nature. The Employer will consult with the affected employees to arrive at the best possible solution to the problem.

ARTICLE 34 - CONTRACTING OUT

The Employer agrees not to contract out any work presently performed by employees covered by this agreement which would result in the layoff of such employees.

ARTICLE 35 - WORK EXPERIENCE PROGRAMS

35.1 Work Experience Programs

The Employer may provide placements for those on work experience or co-op programs where the specific objective is to provide an opportunity for the participant to learn or upgrade their work skills for academic or job preparation purposes. Such placement is subject to the following conditions:

(a) The placement shall not result in the layoff of an existing employee or allow the Employer to keep an existing position vacant;

(b) The placement cannot exceed six months;

(c) The person in the placement shall become a member of the bargaining unit. The status of the participant shall be that of a "*casual employee*" for the entire duration of the placement. The maximum duration for any such placement will be four months except by mutual agreement of the parties.

(d) The Union shall be advised of the details of any such work experience placement before its commencement.

(e) The participant shall be paid at the Step 1 rate as specified in Appendix I (Salary Grid) of this agreement for the work they are performing, or where no existing position is appropriate, at a rate agreed by the parties.

ARTICLE 36 - CAREER DEVELOPMENT

36.1 Purpose

Both parties recognize that an improved service to the public will result if employees acquire knowledge and skills relating to their job functions.

36.2 Scope and Procedures

(a) The Employer will provide orientation and job-specific training, where the knowledge and skills are solely applicable to the Employer's operations, at the time they commence their employment. The purpose of this training is to provide employees with a basic understanding of their responsibilities, the procedures they are required to follow and the equipment they are required to use.

(b) Upon the completion of the probationary period, and in every performance appraisal prepared thereafter, the employee and their supervisor shall specifically identify training and development activities for the employee which will either assist in improving current work performance or which might prepare the employee for additional responsibilities or promotion.

(c) The Employer will attempt to assist employees to enhance their skills through providing time off and/or financial assistance for relevant courses and training programs, creating special assignments, and utilizing temporary secondment or acting appointments into higher level positions. The Employer will attempt to fill vacant positions through internal promotion wherever appropriate.

(d) Employees wishing time or financial assistance to participate in self-initiated training or development activity must apply in writing to their supervisor prior to the event. The Employer will decide what portion, if any, of the cost or time of the activity it will reimburse depending on the potential contribution of the proposed activity to the employee's current or future job performance.

(e) The Employer will canvass bargaining unit employees for interest in attending conferences, meetings, special assignments and the like.

The Employer will decide who attends or participates.

ARTICLE 37 - JOB EVALUATION

37.1 Consultation

The Employer will consult with the Union through the Joint Committee on any job evaluation scheme which is implemented for the purpose of reviewing the relative salaries of existing positions.

37.2 Implementation Procedure

If the Employer decides to implement any salary changes based on such job evaluation, the Employer will provide at least 30 days' notice of the proposed changes to the Union, and the bargaining Principals shall meet within 15 days to negotiate the details of any such changes. If the parties cannot agree if and how such changes shall be implemented, and the Employer still wishes to proceed with the issue, the matter shall go to binding arbitration.

37.3 Documents for Committee

The Committee shall be supplied with all the documentation, existing classifications and job descriptions as well as any other information relating to job evaluation, either existing or proposed.

37.4 Rates of Pay

When establishing a new type of position, the rate of pay shall be subject to negotiation between the Employer and Union. The new rate of pay shall be set in relation to other positions and shall be retroactive to the time the new position was established.

37.5 Job Evaluation Consultants

Nothing in this agreement shall be interpreted as barring either party to this agreement from engaging consultants and/or advisors as representatives of either party.

37.6 Classification Review

(a) If an employee believes that their job duties have been materially changed by the Employer from those in the job description, so that the employee is spending a substantial part of their workday in performing duties normally performed by a higher classification, the employee may, within 30 days of receiving the direct instruction, request that the matter be reviewed by the Employer.

(b) If as a result of the investigation the Employer agrees that the employee's duties have changed as per (a) above but that they were undertaken without the direct instructions (verbal or written) of the

Employer, the employee shall be directed to adhere to duties compatible with the employee's job description.

(c) If the Employer determines that the employee's job duties have changed as per (a) above and that such job duties were taken on at the direct instruction of the Employer, then the Employer shall attempt to resolve the issue.

(d) If the employee is not satisfied with the Employer's decision the employee may file a grievance.

- (e) Any arbitration decision under this article must address the criteria as per Clause 37.6(a).
- (f) The job descriptions for all positions define the substance and intent of each employee's work duties. A copy of each job description, and any changes thereto shall be sent to the Union.

37.7 Salary Protection

If an employee's position is reclassified, their salary shall be according to the new rate of pay at the appropriate position seniority step. If this would result in a reduction of salary for any employee, that employee's salary shall remain at their current level until such time as the new salary in the appropriate position seniority step at least equals their current salary.

37.8 Employee Request for a Job Description

Upon request from an employee, the Employer shall, within a reasonable time, supply the employee with a current job description. The parties recognize the value of consultation between the Employer and the employee affected in describing and changing job duties.

37.9 Labour Market Review

Where the Employer has concluded that it is experiencing significant problems in recruiting and/or retaining staff to a specific position, the Employer shall advise the Union that it intends to do a Labour Market Review and shall outline the process it is intending to use. The Employer shall consult with the Union and obtain input and advice.

When the Review is completed the Employer shall advise the Union of its findings and shall table any proposed change to the salary grid for the component in question for the Union's approval. If discussions are necessary they shall be conducted between the parties. If the parties cannot agree on the amount of the proposed change to the salary grid for the specific position in question, then no change shall be implemented.

This clause is not intended to serve as a mechanism for general wage review. It is restricted to those positions where the Employer believes that because of the salary level it is difficult to recruit or retain staff.

ARTICLE 38 - TEMPORARY AND CASUAL EMPLOYEES

38.1 Definition

A temporary employee is one who is employed on a defined term basis for the reasons noted below and is so informed at the time of hiring; for

- replacement of employees on vacation, sick, or other unpaid or paid leave;
- a position created to carry out special projects or work which is not continuous;
- short-term vacancy or workload relief.

The offer of employment shall note the last date of the temporary term but this date may be extended if the conditions which lead to the need for the temporary employee still exist.

A casual employee is one who is hired for short-term specific tasks of less than 60 continuous working days duration. If the task is extended beyond that period, the employee shall be considered a temporary employee as of the 60th day and is entitled to the benefits as per Clause 38.3 (Benefits) as of that date.

A casual employee does not accrue service or position seniority. Seniority, as per Clause 38.5 (Seniority), begins to accrue only on the casual employee's 60th continuous working day.

38.2 Remuneration

Where a temporary or casual employee is hired to perform a job which is classified under Appendix I, that employee shall be paid according to the salary grid outlined in Appendix I.

38.3 Benefits

(a) A temporary employee is entitled to all benefits on the same basis as continuous employees except for Short-Term Disability (STD), Long-Term Disability benefits, or any other benefits that are restricted to continuous employees and not available to temporary employees under the Employer's benefits contract with its insurer.

(b) A casual employee is not eligible for any benefits including leave, flex or compressed time, non statutorily required maternity leave benefits, layoff or severance pay benefits. The casual employee shall receive 10% of their salary in lieu of benefits and vacation.

38.4 Probation

Article 27 (Probation) applies for all temporary employees where the term is greater than six months.

38.5 Seniority

(a) A temporary employee accrues no service seniority and accrues position seniority for the purposes of placement on the salary grid only. If a temporary employee becomes a continuous employee during the temporary term within six months of the termination of temporary employment, the time spent as a temporary employee will be credited and (if applicable) position seniority upon successfully completing the probationary period as a continuous employee. However, the seniority earned during the period of temporary employment may not be credited for the purpose of exercising bumping rights as per Clause 13.2 (Layoff and Recall)(d)(1)(i) or (d)(1)(ii).

(b) If a temporary employee remains in service with the Employer for a period greater than 250 days of work within a 15-month period, that individual shall be considered a continuous employee and shall have their salary and entitlements dated from the first day of employment with the Employer. However, the seniority earned during the period of temporary employment may not be credited for the purpose of exercising bumping rights as per Clause 13.2 (Layoff and Recall)(d)(1)(i) or (d)(1)(ii).

If the purpose of the temporary appointment was for replacement of an employee absent on Maternity Leave as per Article 21 (Maternity and Parental Leave), the temporary employee has to have been with the Employer for 300 days of work within a 15-month period before they are considered a continuous employee.

38.6 Performance Review

Article 28 (Performance Appraisal) applies to all temporary employee positions, where the term is greater than three months. For in-service or external temporary or casual employee positions greater than two weeks and less than three months, the Employer will provide the employee with feedback on performance

with a view to identifying areas of strength and those which, if improved, would assist the employee in gaining continuous employment in the position.

38.7 Termination

If a temporary employee is absent on unpaid leave for a period greater than 10 working days during the term, the Employer may end the appointment without notice.

If the circumstances which gave rise to the creation of a temporary position should change, the Employer may end the employment of a temporary employee on 10 days' notice. No layoff or severance benefits pursuant to this agreement are payable, except as may; be required by law.

ARTICLE 39 - ACTING EMPLOYEES

39.1 Definition

An acting employee is a continuous employee who is temporarily placed in a higher paid position.

39.2 Remuneration

An acting employee shall go to the first step on the pay grid for the position being acted in which will result in a salary increase of at least 4% over the employee's current salary.

39.3 Benefits and Entitlements

All benefits and entitlements shall continue to accrue at the specified rates.

39.4 Probation

Article 27 (Probation) applies for all acting employees where the acting term is greater than two months.

39.5 Seniority

An acting employee continues to earn Service Seniority throughout the acting term. Position seniority for the acting position shall only accrue if the employee subsequently becomes a continuous employee in the new position. If an acting employee returns to the former position, the time spent in the acting position shall count towards position seniority in the employee's regular position.

39.6 Right to Position

(a) If an acting employee acts in the same position for a total of 250 days of work in a 12-month period, that employee shall then be entitled to the first same vacant continuous position that becomes available in the employee's work unit. This placement takes precedence over any existing eligibility list for that position, except for a lateral transfer. The determining factor for placement in this position shall be service seniority. This right shall only exist for a period of 12 months from the date the employee ceased to act in the position.

(b) The 250 days of work as per (a) shall not include any form of leave other than vacation leave, paid or unpaid, greater than 20 workdays during the 12-month period.

ARTICLE 40 - TERM OF AGREEMENT

40.1 Duration

This agreement shall be binding and remain in effect to midnight, March 31, 2025. No strike or lockout shall take place during this term.

40.2 Notice to Bargain

(a) Either party may, by giving written notice to the other party, on or after December 1, 2024 require the other party to commence collective bargaining on a renewal or revision of this collective agreement.

(b) Where no notice is given by either party by midnight on January 1, 2025 both parties shall be deemed to have been given notice under this article on January 1, 2025 and thereupon Clause 40.3 (Commencement of Bargaining) applies.

(c) All notices on behalf of the Union shall be given by the President or designate of the Union and similar notices on behalf of the Employer shall be given by the CEO.

40.3 Commencement of Bargaining

Where a party to this agreement has given notice under Clause 40.2 (Notice to Bargain), the parties shall, within 10 days after the notice was given, commence collective bargaining.

40.4 Changes in Agreement

Any change deemed necessary in this agreement may be made by mutual agreement at any time during the life of this agreement by the bargaining Principals.

40.5 Agreement to Continue in Force

Both parties shall adhere fully to the terms of this agreement for the term specified in Clause 40.1 (Duration) and shall continue in effect thereafter until strike or lockout notice is served or the agreement is renewed, revised or replaced.

40.6 Effective Date of Agreement

The provisions of the agreement shall come into full force and effect on the date of ratification, and shall not be retroactive except where specified.

40.7 Essential Services

The parties recognize that in the event of a labour dispute the Employer may submit a written request to the Union to negotiate the continuation of services the Employer believes are essential. Upon receiving this request negotiators shall meet immediately.

SIGNED ON BEHALF OF THE UNION:

SIGNED ON BEHALF OF THE EMPLOYER:

Stephanie Smith President Jelena Vukmirica VP Human Resources and Corporate Services

Steve Kitcher Bargaining Member Colin Millar VP Operations

Tina Gaiardo<u></u> Bargaining Member Ana Rogic Senior Enforcement Manager

Debra Calder Bargaining Member

Deb Wilson Staff Representative, Negotiations

Date: _____

APPENDIX I Salary Grid

General Wage Increases

General Wage Increases for the three-year term (April 1, 2022 - March 31, 2025) as follows:

- April 1, 2022: Increase all rates of pay by a flat rate of \$0.25 per hour and a 3.24% GWI.
- April 1, 2023: Increase all rates of pay by the annualized average of BC CPI over twelve months starting on March 1, 2022 to a minimum of 5.5% and a maximum of 6.75%, subject to the COLA MOU.
- April 1, 2024: Increase rates of pay by the annualized average of BC CPI over twelve months starting on March 1, 2023 to a minimum of 2% and a maximum of 3%, subject to the COLA MOU.

April 1, 2022 - Part 1 of 4									
						Rates for en	nployees who c per Cla	onverted 1 we use 30.3	ek vacation as
Position	Step	Ар	r 1/22	Apr 1	I/22 (CT)		h 37.5 hrs' AL version) with 37.5 hrs' nversion
	-	Yearly	Semi-Monthly	Yearly	Semi-Monthly	Yearly	Semi-Monthly	Yearly	Semi-Monthly
	Step 1 (0-6 mos)	\$53,057.00	\$2,210.71	\$54,024.00	\$2,251.00	\$54,108.00	\$2,254.50	\$55,094.32	\$2,295.60
	Step 2 (7-12 mos)	\$55,156.00	\$2,298.17	\$56,162.00	\$2,340.08	\$56,249.00	\$2,343.71	\$57,275.36	\$2,386.47
Enforcement Officer**	Step 3 (13-24 mos)	\$57,887.00	\$2,411.96	\$58,946.00	\$2,456.08	\$59,034.00	\$2,459.75	\$60,111.16	\$2,504.63
	Step 4 (25-36 mos)	\$60,184.00	\$2,507.67	\$61,282.00	\$2,553.42	\$61,377.00	\$2,557.38	\$62,497.09	\$2,604.05
	Step 5 (37+ mos)	\$63,162.00	\$2,631.75	\$64,282.00	\$2,678.42	\$64,415.00	\$2,683.96	\$65,521.50	\$2,730.06
	Step 1 (0-6 mos)	\$43,077.00	\$1,794.88	\$43,861.00	\$1,827.54	\$43,929.00	\$1,830.38	\$44,727.83	\$1,863.66
	Step 2 (7-12 mos) Step 3	\$44,747.00	\$1,864.46	\$45,561.00	\$1,898.38	\$45,632.00	\$1,901.33	\$46,462.26	\$1,935.93
Enrollment Officer (ENO)	(13-24 mos)	\$46,915.00	\$1,954.79	\$47,769.00	\$1,990.38	\$47,843.00	\$1,993.46	\$48,714.86	\$2,029.79
	Step 4 (25-36 mos)	\$48,736.00	\$2,030.67	\$49,623.00	\$2,067.63	\$49,700.00	\$2,070.83	\$50,605.39	\$2,108.56
	Step 5 (37+ mos)	\$49,700.00	\$2,070.83	\$50,605.00	\$2,108.54	\$50,684.00	\$2,111.83	\$51,607.02	\$2,150.29
	Step 1 (0-6 mos)	\$39,818.00	\$1,659.08	\$40,542.00	\$1,689.25	\$40,605.00	\$1,691.88	\$41,341.82	\$1,722.58
	Step 2 (7-12 mos)	\$41,330.00	\$1,722.08	\$42,081.00	\$1,753.38	\$42,147.00	\$1,756.13	\$42,912.84	\$1,788.04
Enforcement Assistant	Step 3 (13-24 mos)	\$43,296.00	\$1,804.00	\$44,083.00	\$1,836.79	\$44,152.00	\$1,839.67	\$44,954.39	\$1,873.10
	Step 4 (25-36 mos)	\$44,950.00	\$1,872.92	\$45,768.00	\$1,907.00	\$45,839.00	\$1,909.96	\$46,672.59	\$1,944.69
	Step 5 (37+ mos)	\$45,839.00	\$1,909.96	\$46,673.00	\$1,944.71	\$46,746.00	\$1,947.75	\$47,596.92	\$1,983.21
	Step 1 (0-6 mos)	\$40,391.00	\$1,682.96	\$41,125.00	\$1,713.54	\$41,189.00	\$1,716.21	\$41,937.56	\$1,747.40
Enquiry Representative**	Step 2 (7-12 mos)	\$42,931.00	\$1,788.79	\$43,712.00	\$1,821.33	\$43,780.00	\$1,824.17	\$44,576.07	\$1,857.34
	Step 3 (13-24 mos)	\$45,214.00	\$1,883.92	\$46,037.00	\$1,918.21	\$46,108.00	\$1,921.17	\$46,947.90	\$1,956.16

Position

Step

Step 4 (25-36 mos)

	April 1	, 2022 - P	art 1 of 4				
	Арпт	<u>, 2022 - F</u>		Rates for em	ployees who c per Cla	onverted 1 we use 30.3	ek vacation as
Ар	r 1/22	Apr 1	/22 (CT)		n 37.5 hrs' AL ersion	• •) with 37.5 hrs' nversion
Yearly	Semi-Monthly	Yearly	Semi-Monthly	Yearly	Semi-Monthly	Yearly	Semi-Monthly
\$46,108.00	\$1,921.17	\$46,947.00	\$1,956.13	\$47,020.00	\$1,959.17	\$47,875.83	\$1,994.83
\$37,024.00	\$1,542.67	\$37,696.00	\$1,570.67	\$37,755.00	\$1,573.13	\$38,439.49	\$1,601.65
\$38,486.00	\$1,603.58	\$39,185.00	\$1,632.71	\$39,245.00	\$1,635.21	\$39,958.11	\$1,664.92

	Step 1								
	(0-6 mos)	\$37,024.00	\$1,542.67	\$37,696.00	\$1,570.67	\$37,755.00	\$1,573.13	\$38,439.49	\$1,601.65
	Step 2								
	(7-12 mos)	\$38,486.00	\$1,603.58	\$39,185.00	\$1,632.71	\$39,245.00	\$1,635.21	\$39,958.11	\$1,664.92
	Step 3								
Office Assistant	(13-24								
	mos)	\$40,384.00	\$1,682.67	\$41,117.00	\$1,713.21	\$41,181.00	\$1,715.88	\$41,929.42	\$1,747.06
	Step 4								
	(25-36	* / / * * * *	*	A 40 T 40 00	* / = * *	* 40,000,00	A / 300 03	A 40 500 00	A (A (A)
	mos)	\$41,979.00	\$1,749.13	\$42,742.00	\$1,780.92	\$42,808.00	\$1,783.67	\$43,586.30	\$1,816.10
	Step 5	¢ 40,000,00	¢4 700 07	¢ 40 507 00	¢4.040.40	¢40.054.00	¢4.040.00	¢ 4 4 4 4 0 4 0	¢4.050.04
	(37+ mos)	\$42,808.00	\$1,783.67	\$43,587.00	\$1,816.13	\$43,654.00	\$1,818.92	\$44,448.19	\$1,852.01
	25-36 mos	\$41,701.00	\$1,737.54	\$42,459.00	\$1,769.13	\$42,525.00	\$1,771.88	\$43,298.22	\$1,804.09
	37-48 mos	\$43,760.00	\$1,823.33	\$44,556.00	\$1,856.50	\$44,625.00	\$1,859.38	\$45,437.35	\$1,893.22
	49-60 mos	\$45,922.00	\$1,913.42	\$46,758.00	\$1,948.25	\$46,831.00	\$1,951.29	\$47,682.82	\$1,986.78
Logal Assistant**	61-72 mos	\$48,193.00	\$2,008.04	\$49,070.00	\$2,044.58	\$49,147.00	\$2,047.79	\$50,041.85	\$2,085.08
Legal Assistant**	73-84 mos	\$50,579.00	\$2,107.46	\$51,500.00	\$2,145.83	\$51,580.00	\$2,149.17	\$52,519.61	\$2,188.32
	85-96 mos	\$53,081.00	\$2,211.71	\$54,048.00	\$2,252.00	\$54,132.00	\$2,255.50	\$55,119.20	\$2,296.63
	97-108								
	mos	\$54,132.00	\$2,255.50	\$55,119.00	\$2,296.63	\$55,204.00	\$2,300.17	\$56,210.44	\$2,342.10
	Step 1								
	(0-6 mos)	\$39,277.00	\$1,636.54	\$39,820.00	\$1,659.17	\$40,052.00	\$1,668.83	\$40,606.00	\$1,691.92
	Step 2								
	(7-12 mos)	\$40,765.00	\$1,698.54	\$41,329.00	\$1,722.04	\$41,571.00	\$1,732.13	\$42,145.00	\$1,756.04
	Step 3								
Doumont Droppoping Clark	(13-24								
Payment Processing Clerk	mos)	\$42,705.00	\$1,779.38	\$43,295.00	\$1,803.96	\$43,548.00	\$1,814.50	\$44,151.00	\$1,839.63
	Step 4								
	(25-36								
	mos)	\$44,337.00	\$1,847.38	\$44,951.00	\$1,872.96	\$45,214.00	\$1,883.92	\$45,839.00	\$1,909.96
	Step 5								
	(37+ mos)	\$45,214.00	\$1,883.92	\$45,840.00	\$1,910.00	\$46,108.00	\$1,921.17	\$46,747.00	\$1,947.79

CT = compressed time as per Clause 15.3 (Flex Days and Compressed Time) or MOA #6 (Compressed Time Payment Services)

**Includes adjustments per LOU #1 (Service Improvement Fund - Hard to Recruit or Retain Staff Positions)

	April 1, 2022 - Part 2 of 4									
		Rates for employees who converted 1 week F + 7.5 hours' vacation as per Clause 30.3					Rates for employees who converted 1 week + 15 hours' vacation as per Clause 30.3			
Position	Step	-	Apr 1/21 with 45 hrs' AL Conversion		Apr 1/22 (CT) with 45 hrs' AL Conversion		Apr 1/22 with 52.5 hrs' AL Conversion		(CT) with irs' AL ersion	
		Yearly	Semi- Monthly	Yearly	Semi- Monthly	Yearly	Semi- Monthly	Yearly	Semi- Monthly	
	Step 1 (0-6 mos)	\$54,321.00	\$2,263.38	\$55,311.00	\$2,304.63	\$54,535.00	\$2,272.29	\$55,529.00	\$2,313.71	
Enforcement Officer**	Step 2 (7-12 mos)	\$56,471.00	\$2,352.96	\$57,501.00	\$2,395.88	\$56,693.00	\$2,362.21	\$57,727.00	\$2,405.29	
	Step 3 (13-24									
	/	\$59,268.00 \$61,620.00						\$60,586.00 \$62,991.00	\$2,524.42 \$2,624.63	

			April 1, 20	022 - Part	2 of 4				
				who conver n as per Cla				who convert n as per Cla	
Position	Step	Apr 1/21 with 45 hrs' AL Conversion		Apr 1/22 (CT) with 45 hrs' AL Conversion		Apr 1/22 with 52.5 hrs' AL Conversion		52.5 h	(CT) with nrs' AL ersion
		Yearly	Semi- Monthly	Yearly	Semi- Monthly	Yearly	Semi- Monthly	Yearly	Semi- Monthly
	(25-36 mos)								
	Step 5 (37+ mos)	\$64 630 00	\$2 692 92	\$65,777.00	\$2 740 71	\$64 883 00	\$2 703 46	\$66,034.00	\$2,751.42
	Step 1 (0-6 mos)			\$44,905.00					\$1,878.33
	Step 2 (7-12 mos)			\$46,645.00					\$1,951.21
Enrollment Officer	Step 3 (13-24	ψ 1 0,012.00	ψ1,300.03	ψ - 0,0 - 3.00	ψ1,0-0.0-	ψ 1 0,002.00	ψ1,010.00	ψ 1 0,023.00	ψ1,301.21
(ENO)	mos) Step 4	\$48,032.00	\$2,001.33	\$48,907.00	\$2,037.79	\$48,221.00	\$2,009.21	\$49,099.00	\$2,045.79
	(25-36 mos)	\$49,896.00	\$2,079.00	\$50,805.00	\$2,116.88	\$50,093.00	\$2,087.21	\$51,004.00	\$2,125.17
	Step 5 (37+ mos)	\$50,884.00	\$2,120.17	\$51,811.00	\$2,158.79	\$51,083.00	\$2,128.46	\$52,015.00	\$2,167.29
	Step 1 (0-6 mos)	\$40,764.00	\$1,698.50	\$41,505.00	\$1,729.38	\$40,924.00	\$1,705.17	\$41,667.00	\$1,736.13
	Step 2 (7-12 mos)	\$42,313.00	\$1,763.04	\$43,081.00	\$1,795.04	\$42,478.00	\$1,769.92	\$43,251.00	\$1,802.13
Enforcement Assistant	Step 3 (13-24 mos)	\$44,326.00	\$1,846.92	\$45,132.00	\$1,880.50	\$44,499.00	\$1,854.13	\$45,309.00	\$1,887.88
	Step 4 (25-36 mos)	\$46,019.00	\$1,917.46	\$46,857.00	\$1,952.38	\$46,200.00	\$1,925.00	\$47,041.00	\$1,960.04
	Step 5 (37+ mos)	\$46,930.00	\$1,955.42	\$47,784.00	\$1,991.00	\$47,114.00	\$1,963.08	\$47,972.00	\$1,998.83
	Step 1 (0-6 mos)	\$41,351.00	\$1,722.96	\$42,102.00	\$1,754.25	\$41,513.00	\$1,729.71	\$42,268.00	\$1,761.17
	Step 2 (7-12 mos)	\$43,952.00	\$1,831.33	\$44,752.00	\$1,864.67	\$44,124.00	\$1,838.50	\$44,927.00	\$1,871.96
Enquiry Representative**	Step 3 (13-24 mos)	\$46,290.00	\$1,928.75	\$47,132.00	\$1,963.83	\$46,472.00	\$1,936.33	\$47,318.00	\$1,971.58
	Step 4 (25-36 mos)	\$47,205.00	\$1,966.88	\$48,064.00	\$2,002.67	\$47,390.00	\$1,974.58	\$48,253.00	\$2,010.54
	Step 1 (0-6 mos)	\$37,903.00		\$38,591.00					\$1,614.29
	Step 2 (7-12 mos)	\$39,400.00	\$1,641.67	\$40,115.00	\$1,671.46	\$39,554.00	\$1,648.08	\$40,373.00	\$1,682.21
Office Assistant	Step 3 (13-24 mos)	\$41,343.00	\$1,722.63	\$42,094.00	\$1,753.92	\$41,506.00	\$1,729.42	\$42,259.00	\$1,760.79
	Step 4 (25-36 mos)	\$42.978.00	\$1,790.75	\$43.758.00	\$1,823.25	\$43.146.00	\$1,797.75	\$43,930.00	\$1,830.42
	Step 5 (37+ mos)							\$44,798.00	\$1,866.58

		4	April 1, 20	022 - Part	2 of 4				
				who conver n as per Cla		Rates for employees who converted 1 week + 15 hours' vacation as per Clause 30.3			
Position	Step	Apr 1/21 with 45 hrs' AL Conversion		Apr 1/22 (CT) with 45 hrs' AL Conversion		Apr 1/22 with 52.5 hrs' AL Conversion		Apr 1/22 (CT) with 52.5 hrs' AL Conversion	
		Yearly	Semi- Monthly	Yearly	Semi- Monthly	Yearly	Semi- Monthly	Yearly	Semi- Monthly
	25-36 mos			\$43,470.00	\$1,811.25	\$42,862.00	\$1,785.92	\$43,641.00	\$1,818.38
	37-48 mos	\$44,801.00	\$1,866.71	\$45,617.00	\$1,900.71	\$44,978.00	\$1,874.08	\$45,797.00	\$1,908.21
	49-60 mos	\$47,016.00	\$1,959.00	\$47,872.00	\$1,994.67	\$47,202.00	\$1,966.75	\$48,061.00	\$2,002.54
Legal Assistant**	61-72 mos	\$49,341.00	\$2,055.88	\$50,240.00	\$2,093.33	\$49,536.00	\$2,064.00	\$50,438.00	\$2,101.58
Legal Assistant	73-84 mos	\$51,785.00	\$2,157.71	\$52,728.00	\$2,197.00	\$51,989.00	\$2,166.21	\$52,936.00	\$2,205.67
	85-96 mos	\$54,347.00	\$2,264.46	\$55,338.00	\$2,305.75	\$54,562.00	\$2,273.42	\$55,556.00	\$2,314.83
	97-108 mos	\$55,423.00	\$2,309.29	\$56,433.00	\$2,351.38	\$55,642.00	\$2,318.42	\$56,656.00	\$2,360.67
	Step 1 (0-6 mos)	\$40,210.00	\$1,675.42	\$40,766.00	\$1,698.58	\$40,368.00	\$1,682.00	\$40,925.00	\$1,705.21
	Step 2 (7-12 mos)	\$41,734.00	\$1,738.92	\$42,312.00	\$1,763.00	\$41,897.00	\$1,745.71	\$42,477.00	\$1,769.88
Payment Processing	Step 3 (13-24								
Clerk	mos)	\$43,720.00	\$1,821.67	\$44,325.00	\$1,846.88	\$43,891.00	\$1,828.79	\$44,499.00	\$1,854.13
	Step 4 (25-36	\$45 301 00	¢1 801 20	\$46,020,00	¢1 017 50	\$45 570 00	¢1 808 75	\$46,201.00	\$1,925.04
	mos) Step 5	φ4 0,391.00	φ1,091.29	φ4 0,020.00	φ1,917.30	φ4 0,070.00	φ1,080.73	φ4 0,201.00	φ1,920.04
	(37+ mos)	\$46,290.00	\$1,928.75	\$46,931.00	\$1,955.46	\$46,472.00	\$1,936.33	\$47,115.00	\$1,963.13

CT = compressed time as per Clause 15.3 (Flex Days and Compressed Time) or MOA #6 (Compressed Time - Payment Services)

**Includes adjustments per LOU #1 (Service Improvement Fund - Hard to Recruit or Retain Staff Positions)

		Α	pril 1, 202	22 - Part 3	of 4				
			k + 22.5 hrs	es who conv vacation as e 30.3		Rates for employees who converted 1 week + 30 hrs vacation as per Clause 30.3			
Position	Step	Apr 1/22 with 60 hrs' AL Conversion		Apr 1/22 (CT) with 60 hrs' AL Conversion		Apr 1/22 with 67.5 hrs' AL Conversion		Apr 1/22 (CT) with 67.5 hrs' AL Conversion	
		Yearly	Semi- Monthly	Yearly	Semi- Monthly	Yearly	Semi- Monthly	Yearly	Semi- Monthly
	Step 1 (0-6 mos)	\$54,749.00	\$2,281.21	\$55,747.00	\$2,322.79	\$54,962.00	\$2,290.08	\$55,964.00	\$2,331.83
	Step 2 (7-12 mos)	\$56,915.00	\$2,371.46	\$57,954.00	\$2,414.75	\$57,138.00	\$2,380.75	\$58,179.00	\$2,424.13
Enforcement Officer**	Step 3 (13-24 mos)	\$59,734.00	\$2,488.92	\$60,823.00	\$2,534.29	\$59,967.00	\$2,498.63	\$61,061.00	\$2,544.21
	Step 4 (25-36 mos)	\$62,105.00	\$2,587.71	\$63,239.00	\$2,634.96	\$62,347.00	\$2,597.79	\$63,486.00	\$2,645.25
	Step 5 (37+ mos)	\$65,135.00	\$2,713.96	\$66,290.00	\$2,762.08	\$65,386.00	\$2,724.42	\$66,547.00	\$2,772.79
Enrollment Officer	Step 1 (0-6 mos)	\$44,448.00	\$1,852.00	\$45,257.00	\$1,885.71	\$44,621.00	\$1,859.21	\$45,432.00	\$1,893.00
(ENO)	Step 2 (7-12 mos)	\$46,172.00	\$1,923.83	\$47,012.00	\$1,958.83	\$46,351.00	\$1,931.29	\$47,195.00	\$1,966.46

April 1, 2022 - Part 3 of 4									
			k + 22.5 hrs	es who conv vacation as e 30.3				who conver as per Claus	
Position	Step	Apr 1/2 60 hr Conve	s' AL ersion	60 hr	(CT) with s' AL ersion	Apr 1/2 67.5 h Conve	rs' AL ersion	Apr 1/22 67.5 h Conve	rs' AL ersion
		Yearly	Semi- Monthly	Yearly	Semi- Monthly	Yearly	Semi- Monthly	Yearly	Semi- Monthly
	Step 3 (13-24 mos)	\$48,409.00	\$2.017.04	\$49,290.00	\$2.053.75	\$48.598.00	\$2.024.92	\$49,482.00	\$2.061.75
	Step 4 (25-36 mos)			\$51,205.00				\$51,404.00	
	Step 5		-						
	(37+ mos) Step 1	\$51,284.00	\$2,136.83	\$52,218.00	\$2,175.75	\$51,483.00	\$2,145.13	\$52,422.00	\$2,184.25
	(0-6 mos)	\$41,084.00	\$1,711.83	\$41,830.00	\$1,742.92	\$41,244.00	\$1,718.50	\$41,993.00	\$1,749.71
	Step 2 (7-12 mos)	\$42.644.00	\$1.776.83	\$43,420.00	\$1.809.17	\$42.810.00	\$1.783.75	\$43,589.00	\$1.816.21
Enforcement Assistant	Step 3								
	(13-24 mos) Step 4	\$44,673.00	\$1,861.38	\$45,486.00	\$1,895.25	\$44,847.00	\$1,868.63	\$45,664.00	\$1,902.67
	(25-36 mos)	\$46,380.00	\$1,932.50	\$47,225.00	\$1,967.71	\$46,561.00	\$1,940.04	\$47,408.00	\$1,975.33
	Step 5 (37+ mos)	\$47,298.00	\$1,970.75	\$48,159.00	\$2,006.63	\$47,482.00	\$1,978.42	\$48,347.00	\$2,014.46
	Step 1 (0-6 mos)	\$41 675 00	\$1 736 /6	\$42,433.00	\$1 768 0/	\$41 837 00	\$1 7/3 21	\$42,598.00	\$1 77/ 92
	Step 2								
Enquiry Representative**	(7-12 mos) Step 3	\$44,296.00	\$1,845.67	\$45,103.00	\$1,879.29	\$44,469.00	\$1,852.88	\$45,279.00	\$1,886.63
	(13-24 mos)	\$46,653.00	\$1,943.88	\$47,503.00	\$1,979.29	\$46,835.00	\$1,951.46	\$47,687.00	\$1,986.96
	Step 4 (25-36 mos)	\$47,576.00	\$1,982.33	\$48,442.00	\$2,018.42	\$47,761.00	\$1,990.04	\$48,630.00	\$2,026.25
	Step 1								
	(0-6 mos) Step 2	\$38,200.00	\$1,591.67	\$38,893.00	\$1,620.54	\$38,348.00	\$1,597.83	\$39,045.00	\$1,626.88
	(7-12 mos)	\$39,708.00	\$1,654.50	\$40,430.00	\$1,684.58	\$39,862.00	\$1,660.92	\$40,587.00	\$1,691.13
Office Assistant	Step 3 (13-24 mos)	\$41,667.00	\$1,736.13	\$42,424.00	\$1,767.67	\$41,829.00	\$1,742.88	\$42,589.00	\$1,774.54
	Step 4 (25-36 mos)	\$43 314 00	\$1 804 75	\$44 101 00	\$1 837 54	\$43 483 00	\$1 811 79	\$44,273.00	\$1 844 71
	Step 5		-						
	(37+ mos) 25-36 mos							\$45,148.00 \$43,983.00	
	37-48 mos			\$45,976.00				\$46,156.00	
	49-60 mos			\$48,249.00				\$48,438.00	
Legal Assistant**	61-72 mos	\$49,730.00	\$2,072.08	\$50,637.00	\$2,109.88	\$49,925.00	\$2,080.21	\$50,835.00	\$2,118.13
	73-84 mos			\$53,144.00				\$53,352.00	
	85-96 mos	\$54,775.00				\$54,990.00		\$55,993.00	
	97-108 mos	\$55,860.00		\$56,878.00		\$56,079.00		\$57,101.00	
	Step 1								
	(0-6 mos)	 \$40,525.00	\$1,688.54	ֆ41,086.00	\$1,711.92	 φ40,682.00	\$1,095.08	\$41,245.00	\$1,718.54
Payment Processing	Step 2 (7-12 mos)	\$42,061.00	\$1,752.54	\$42,643.00	\$1,776.79	\$42,225.00	\$1,759.38	\$42,809.00	\$1,783.71
Clerk	Step 3 (13-24 mos)	\$44,063.00	\$1,835.96	\$44,673.00	\$1,861.38	\$44,235.00	\$1,843.13	\$44,847.00	\$1,868.63
	Step 4							\$46,562.00	
	20-00 1103/	ψτυ, <i>ι</i> τ0.00	ψι,300.17	ψτυ,υυ Ι.υυ	ψ1,302.04	ψτυ,320.00	ψι,σιο.ου	ψτυ,υυΖ.υυ	ψι,0 1 0.00

		Α	pril 1, 202	22 - Part 3	of 4				
		Rates for employees who converted 1 week + 22.5 hrs vacation as per Clause 30.3				Rates for employees who converted 1 weel + 30 hrs vacation as per Clause 30.3			
Position	Step	60 hr	Apr 1/22 with 60 hrs' AL Conversion		Apr 1/22 (CT) with 60 hrs' AL Conversion		Apr 1/22 with 67.5 hrs' AL Conversion		(CT) with rs' AL ersion
		Yearly	Yearly Semi- Monthly		Semi- Monthly	Yearly	Semi- Monthly	Yearly	Semi- Monthly
	Step 5 (37+ mos)	\$46,653.00	\$1,943.88	\$47,299.00	\$1,970.79	\$54,962.00	\$2,290.08	\$47,483.00	\$1,978.46

CT = compressed time as per Clause 15.3 (Flex Days and Compressed Time) or MOA #6 (Compressed Time Payment Services)

**Includes adjustments per LOU #1 (Service Improvement Fund - Hard to Recruit or Retain Staff Positions)

		April 1, 2022 - Pa	rt 4 of 4		
		Rates for employees	who converted 1 weel	k + 37.5 hrs vacatio	on as per Clause 30.
Position	Step	Apr 1/22 with 75 hr	s' AL Conversion		T) with 75 hrs' onversion
		Yearly	Semi-Monthly	Yearly	Semi-Monthly
	Step 1		* 0.000.00	\$50,400,00	* 0.040.00
	(0-6 mos)	\$55,176.00	\$2,299.00	\$56,182.00	\$2,340.92
	Step 2	¢57,000,00	¢0.000.00	¢50,400,00	¢0 400 F0
	(7-12 mos)	\$57,360.00	\$2,390.00	\$58,406.00	\$2,433.58
Enforcement Officer**	Step 3 (13-24 mos)	\$60,200.00	\$2,508.33	\$61,298.00	\$2,554.08
	Step 4	φ00,200.00	ψ2,000.00	φ01,200.00	φ2,004.00
	(25-36 mos)	\$62,590.00	\$2,607.92	\$63,732.00	\$2,655.50
	Step 5	+,	+=,•••	+	+_,
	(37+ mos)	\$65,637.00	\$2,734.88	\$66,803.00	\$2,783.46
	Step 1				
	(0-6 mos)	\$44,794.00	\$1,866.42	\$45,609.00	\$1,900.38
	Step 2				
	(7-12 mos)	\$46,531.00	\$1,938.79	\$47,378.00	\$1,974.08
Enrollment Officer (ENO)	Step 3				
	(13-24 mos)	\$48,786.00	\$2,032.75	\$49,674.00	\$2,069.75
	Step 4	* -• ••• ••	A0 (() 0 -	A- (AA (AA	
	(25-36 mos)	\$50,680.00	\$2,111.67	\$51,604.00	\$2,150.17
	Step 5	¢E4 C04 00	¢0.450.50	¢50 005 00	¢0 400 74
	(37+ mos)	\$51,684.00	\$2,153.50	\$52,625.00	\$2,192.71
	Step 1 (0-6 mos)	\$41,403.00	\$1,725.13	\$42,156.00	\$1,756.50
	Step 2	ψ+1,405.00	ψ1,725.15	ψ 1 2,150.00	φ1,700.00
	(7-12 mos)	\$42,976.00	\$1,790.67	\$43,758.00	\$1,823.25
	Step 3	<i><i><i>q</i> 12,010.00</i></i>	¢1,100.01	<i>Q</i> 10,100.00	ψ1,020.20
Enforcement Assistant	(13-24 mos)	\$45,021.00	\$1,875.88	\$45,841.00	\$1,910.04
	Step 4				
	(25-36 mos)	\$46,742.00	\$1,947.58	\$47,592.00	\$1,983.00
	Step 5				
	(37+ mos)	\$47,667.00	\$1,986.13	\$48,535.00	\$2,022.29
	Step 1				
Enquiry Representative**	(0-6 mos)	\$41,999.00	\$1,749.96	\$42,763.00	\$1,781.79
	Step 2	*	* 4 000 04		#4 000 00
	(7-12 mos)	\$44,641.00	\$1,860.04	\$45,454.00	\$1,893.92

		April 1, 2022 - Pa	rt 4 of 4		
		Rates for employees	who converted 1 weel	k + 37.5 hrs vacatio	on as per Clause 30.3
Position	Step	Apr 1/22 with 75 hr	s' AL Conversion		CT) with 75 hrs' onversion
		Yearly	Semi-Monthly	Yearly	Semi-Monthly
	Step 3 (13-24 mos)	\$47,017.00	\$1,959.04	\$47,873.00	\$1,994.71
	Step 4 (25-36 mos)	\$47,946.00	\$1,997.75	\$48,819.00	\$2,034.13
	Step 1 (0-6 mos)	\$38,496.00	\$1,604.00	\$39,195.00	\$1,633.13
	Step 2 (7-12 mos)	\$40,017.00	\$1,667.38	\$40,744.00	\$1,697.67
Office Assistant	Step 3 (13-24 mos)	\$41,992.00	\$1,749.67	\$42,755.00	\$1,781.46
	Step 4 (25-36 mos)	\$43,651.00	\$1,818.79	\$44,445.00	\$1,851.88
	Step 5 (37+ mos)	\$44,514.00	\$1,854.75	\$45,323.00	\$1,888.46
	25-36 mos	\$43,365.00	\$1,806.88	\$44,154.00	\$1,839.75
	37-48 mos	\$45,508.00	\$1,896.17	\$46,336.00	\$1,930.67
	49-60 mos	\$47,757.00	\$1,989.88	\$48,626.00	\$2,026.08
Legal Assistant**	61-72 mos	\$50,119.00	\$2,088.29	\$51,033.00	\$2,126.38
	73-84 mos	\$52,601.00	\$2,191.71	\$53,560.00	\$2,231.67
	85-96 mos	\$55,205.00	\$2,300.21	\$56,211.00	\$2,342.13
	97-108 mos	\$56,298.00	\$2,345.75	\$57,324.00	\$2,388.50
	Step 1 (0-6 mos)	\$40,841.00	\$1,701.71	\$41,405.00	\$1,725.21
	Step 2 (7-12 mos)	\$42,389.00	\$1,766.21	\$42,975.00	\$1,790.63
Payment Processing Clerk	Step 3 (13-24 mos)	\$44,406.00	\$1,850.25	\$45,021.00	\$1,875.88
	Step 4 (25-36 mos)	\$46,105.00	\$1,921.04	\$46,743.00	\$1,947.63
	Step 5 (37+ mos)	\$47,017.00	\$1,959.04	\$47,668.00	\$1,986.17

CT = compressed time as per Clause 15.3 (Flex Days and Compressed Time) or MOA #6 (Compressed Time Payment Services)

**Includes adjustments per LOU#1 (Service Improvement Fund - Hard to Recruit or Retain Staff Positions)

Note: The lowest possible starting level for a Legal Assistant is the 25-36 month band. If the Legal Assistant has less than 36 months at the time of entry, their next increment will be when they have completed enough time with the Employer to equal 36 months' experience.

The basic qualifications for a Legal Assistant are:

- (a) direct training and certification as a legal assistant or legal secretary; or
- (b) experience as a titled legal assistant or legal secretary with a law firm for at least two years.

Experience will only be credited for the purpose of placement on this grid, where the person has been employed as a full-time, titled legal assistant or legal secretary in a law firm and has significant experience with family documents and process; this does not include general office duties. The number of years' experience is awarded at the time the job offer is made.

Salary adjustments occur on the date the Legal Assistant reaches a complete year of experience, using both prior time credited and time with BCFMA as a Legal Assistant.

Notwithstanding the above, the Employer will accept an internal applicant's equivalent combination of education and experience in place of a formal educational credential and/or work experience for purposes of hiring and initial placement on the wage grid.

APPENDIX II Geographic Location

OFFICE	LOCATION
Victoria Client Offices	Victoria
Payment Services	Victoria
Northern and Interior Client Office	Kamloops
Lower Mainland Client Office	Burnaby

APPENDIX III Low Wage Redress (LWR) - Annual Wage Gap

Job Group	*Annual Wage Gap as a % of Salaries at Top Step
Office Assistant	-30.1%
Payment Processing Clerk	-19.7%
Enquiry Representative	-17.3%
Enforcement Assistant	-21.4%
Legal Assistant	-3.2%
Enrolment Officer	-18.2%
Enforcement Officer	-10%

*Based on equivalent Public Service BCGEU Main Agreement top step rate as of March 31, 2022

MEMORANDUM OF AGREEMENT #1 Personal Growth Allowance

The Parties agree that Legal Assistants shall receive a Personal Growth Allowance as follows:

- Upon ratification of this agreement \$2000
- On April 1, 2023 \$2000
- On April 1, 2024 \$2000

This Memorandum of Understanding is in effect for the term of this collective agreement. This amount will be paid once annually for the term of the agreement. For clarity, these lump sum payments are neither ongoing nor pensionable.

MEMORANDUM OF AGREEMENT #2 Cost of Living Adjustments (COLA)

The parties agree that in determining the level of any Cost of Living Adjustments (COLAs) that will be paid out starting on the first pay period after April 1, 2023 and April 1, 2024, respectively, the "*annualized average of BC CPI over twelve months*" in Appendix 3 of the collective agreement means the *Latest 12-month Average (Index) % Change* reported by BC Stats in March for British Columbia for the twelve months starting at the beginning of March the preceding year and concluding at the end of the following February. The percentage change reported by BC Stats that will form the basis for determining any COLA increase is calculated to one decimal point. The *Latest 12-month Average Index*, as defined by BC Stats, is a 12-month moving average of the BC consumer price indexes of the most recent 12 months. This figure is calculated by averaging index levels over the applicable 12 months.

The Latest 12-month Average % Change is reported publicly by BC Stats in the monthly BC Stats Consumer Price Index Highlights report. The BC Stats Consumer Price Index Highlights report released in mid-March will contain the applicable figure for the 12-months concluding at the end of February.

For reference purposes only, the annualized average of BC CPI over twelve months from March 1, 2021 to February 28, 2022 was 3.4%.

MEMORANDUM OF AGREEMENT #3 Low Wage Redress (LWR)

Employees who meet the following criteria will receive low wage redress comparability adjustments effective April 1, 2022. These increases will be applied to the wage grids prior to the implementation of the general wage increases available under the employer's proposal *General Wage Increases*.

Classifications where the top step is between 10% and 20% below parity with the top step of equivalent positions in the Public Service BCGEU Main Agreement will receive a 5% increase to each step. In accordance with Appendix III, the classifications falling into this category are:

- Enforcement Officer
- Enrolment Officer
- Enquiry Representative
- Payment Processing Clerk

Classifications where the top step is greater than 20% below parity with the top step of equivalent positions in the Public Service BCGEU Main Agreement will receive an 8% increase to each step. In accordance with Appendix III, the classifications falling into this category are:

Enforcement Assistant Office Assistant

MEMORANDUM OF AGREEMENT #4 Indemnity

The parties agree that the employees who have not been guilty of gross negligence and who have carried out their duties in good faith, will not be subject to any personal liability or expense to defend themselves in a court or administrative body of competent jurisdiction.

MEMORANDUM OF AGREEMENT #5 Job Evaluation Process

Under the current language in the collective agreement (Article 37 [Job Evaluation]), there is no mechanism which lets employees request a job evaluation except where job duties have changed so that "... the employee is spending a substantial part of their workday in performing duties normally performed by a higher classification." (Clause 37.6 [Classification Review]). This Memorandum now permits employees to request a job evaluation and describes the process by which all job evaluations shall be conducted and implemented.

Background:

The work of some classifications has probably changed in such a way that the degree of knowledge, and skill required to do them competently, coupled with the demand to be responsive to clients and exercise increased complexity in decision-making (and therefore accountability), justifies a review of the salary range of the position.

This agreement outlines the process to determine if a job evaluation should be undertaken and then how it will be done and implemented. This process is only for evaluations that fall outside of the current Clause 37.6 (Classification Review).

Principles:

- 1. All components should be able to request a job evaluation when it thinks that the position has materially changed to demand a significantly higher level of knowledge, skill, decision-making and accountability. Increases in volumes of work are not grounds for an evaluation.
- 2. The process by which the Employer decides whether or not to conduct a job evaluation should be as open, transparent and fair as possible.
- 3. Implementation of any salary changes because of a job evaluation exercise are dependent on the following:
 - any required approvals (e.g., PSEC, Ministry) being obtained;
 - there being sufficient money available to provide any increase in salary;
 - agreement of the Bargaining Committee to the proposed increase as per Clause 37.2 (Implementation Procedure).

Note: The job evaluation process is meant to address significant material changes in specific positions. It is not an alternate way of increasing salaries overall. That can only be done through collective bargaining.

Process:

The following describes the process which will be used.

Step 1 - Initial Assessment

An assessment can be initiated either by the Employer or by a group of employees in a particular classification.

If the Employer initiates a reclassification assessment, it will advise all employees in that classification it has done an initial assessment and that it intends to conduct a formal job evaluation review (see Step 2).

If a group of employees wishes to initiate a job evaluation, they must send a letter to the CEO outlining in specific detail why they believe it is warranted. This letter must be signed by at least 50% of the total number of employees in that component. The only grounds for an evaluation (whether initiated by the Employer or requested by the employees) is that the job responsibilities have materially changed so that all of the following have substantially increased:

1. The knowledge (technical and process) required to do the job has become more complex. This does not just mean there is more to know but rather that there is more judgment and analysis required to determine what applies to a given situation.

2. Decision-making has become more complex.

This means that the types of decisions staff have to make in doing most of their work require consideration of more variables and inter-relationships.

3. The consequences of error are materially greater. It also means that the consequences of making inappropriate decisions can lead to greater material consequences for the organization than in the past.

When the employees request a job evaluation, the Employer will perform an initial assessment and will consult with a representative group of staff of the component in question to ensure the employees' perspective is fully understood. A union representative is entitled to sit in on this consultation meeting as an observer. This consultation process is not necessary if the Employer decides it has enough information and agrees that a formal job evaluation should take place as per Step 2.

A written response will be provided to all members of the component in question, and the Union, outlining in detail the Employer's decision whether or not to proceed to a formal review.

The decision of the Employer is final (i.e., it may not be grieved) and another initial assessment will not be conducted for at least eighteen months except where there appear to be special circumstances indicating the job may have materially changed during that period.

Step 2 - Job Evaluation Process

If the preliminary assessment indicates there has been a material change as per the factors outlined above, a formal evaluation will be undertaken. The purpose of a review is to try to determine the degree or extent of change in the position in relation to the three determining factors noted above.

An outside compensation specialist may, at the Employer's discretion, be retained to help prepare a report. A consultation committee will be formed which includes staff from each affected office in the position being reviewed, as well as some management staff.

The specialist's report will detail the findings with an estimate on the overall increase in the degree or approximate magnitude of complexity based on the three factors.

Step 3 - Implementation

If the job evaluation indicates the position's complexity with respect to the three factors has substantially increased, the Employer will propose to the Bargaining Committee (as per Clause 37.2 [Implementation Procedure]) the amount and time schedule of the new salary taking into consideration the following factors:

- the degree to which the three factors have increased the complexity of the job;
- the availability of funds in the budget to pay for any increase;
- overall impact on the organization in terms of such things as internal equity, etc.

The degree or magnitude of the increase in complexity will not usually translate into an equivalent magnitude increase in salary. However, the formula used will apply equally to all components when reclassification is warranted under this Memorandum. Once the bargaining Principals have agreed, no increases can be finalized until any required external approvals are received.

If the bargaining Principals cannot agree on the amount of the increase, Clause 37.2 (Implementation Procedure) shall apply.

MEMORANDUM OF AGREEMENT #6 Compressed Time - Payment Services

Payment Services employees are eligible for compressed time, subject to the following:

1) Employees who are eligible to select the compressed time option work 7.8 hour shifts except in December where the shift length is 7.5 hours. New employees on probation (including temporary employees) are not eligible for compressed time until probation is complete (six months).

a) The 18 minutes can be added to the beginning or end of a 7.5 hours workday;

b) The number of employees who can add the 18 minutes to the beginning of their workday will be determined by the Employer and based on operational requirements;

c) Holiday and sick leave will be deducted at 7.8 hours per day for months that compressed time is being earned.

2) Seven of the nine compressed days shall be taken in the month that they are earned on either a Wednesday or Thursday. Tuesday will be designed as "*flex and compressed free*".

a) The day off will be selected by the employees on service seniority basis one month and then reversed seniority basis the next month;

b) The number of employees who can be absent on a compressed day shall be determined by the Employer and based on operational requirements;

c) If operational requirements necessitate the Employer changing the scheduled day, the employee shall re-schedule the compressed day in consultation with the Employer.

3) Two of the nine compressed days have been converted to salary as per the end note on this memorandum.

4) An employee who wishes to switch from compressed time shifts to regular shifts may do so but the change will not be effective until the following April.

5) The Employer may request mandatory overtime on Wednesdays or Thursdays based on operational requirements.

a) Overtime will be offered on a voluntary basis and granted by seniority;

b) If no employees volunteer, overtime will be required in reverse order of seniority (excluding probationary employees).

MEMORANDUM OF AGREEMENT #7 Workplace Mental Health

The parties share a desire to foster a psychologically healthy workplace and support employees' mental wellness. The parties agree to collaborate to deliver semi-annual mental health workshops or similar in which all employees may participate without loss of pay. Employee participation will be voluntary.

MEMORANDUM OF AGREEMENT #8 Telework

The parties recognize that some service delivery standards have been maintained or improved while some employees have teleworked due to the COVID-19 pandemic. The parties further recognize that the option to telework has been beneficial to some employees. The parties agree to work toward making telework an ongoing option for some employees to the extent operationally practicable.

Within one month of ratification of the renewal agreement the Employer will strike an ad hoc committee to consult employees on policies and procedures for telework as developed by the employer. The committee will consist of the stewards, bargaining committee members, and other employees as selected by the component, so each component is represented. There will be no more than an equal number of Employer representatives. Given the size of the committee, sub-committees may be formed to ensure that employees' input is received.

The ad hoc committee will review telework policies and procedures in a way that includes employee input with respect to their teleworking experiences. The committee will provide a report including recommendations to the parties regarding telework policies by no later than September 30, 2021.

MEMORANDUM OF UNDERSTANDING Clause 31.9 (Legislative Changes)

Changes to the Employer Health Tax or any other premium imposed for purposes similar to the Medical Services Plan premium are excluded and will not be required to be used to increase other employee benefits. If the Employer Health Tax is eliminated and not replaced with another form of Employer paid benefits, Clause 31.9 (Legislative Changes) will be triggered.

If Clause 31.9 (Legislative Changes) is triggered, the liability arising from the amount of savings from the legislative changes to the MSP savings is based on 2017. The parties will endeavour to mutually agree on the liability arising from the MSP savings based on 2017 projected forward. If the parties cannot agree, any party may refer the matter to arbitration.

The liability arising in this Memorandum of Understanding shall expire on March 31, 2022 unless renewed by mutual agreement of the parties.

LETTER OF UNDERSTANDING #1 Service Improvement Fund - Hard to Recruit or Retain Staff Positions

The parties expect that over the life of the collective agreement, the Employer will continue to face recruitment and retention challenges for positions requiring skills that are in high demand. The parties

The Service Improvement Fund presents an opportunity to address these recruitment and retention challenges, through adjustments to grids to address shortages in staffing that negatively impact services to clients.

The amount allocated by the Employer to the fund each year will be in accordance with the following schedule:

Year 1: \$18,000 Year 2: \$19,000 Year 3: \$19,500

positions.

The total amount available over the term of the Agreement shall be \$56,500 and represents the ongoing funding for this initiative.

To address recruitment issues in a hard to recruit classification where services to clients are negatively impacted, the first step on the grid for Enquiry Representatives shall be eliminated effective October 1, 2019.

To address recruitment issues in a hard to recruit classification where services to clients are negatively impacted, the first step on the grid for Legal Assistants shall be eliminated, effective October 1, 2019.

To address retention issues in a hard to retain classification where services to clients are negatively impacted, the top step on the grid for Enforcement Officers shall be increased by \$317.50, effective October 1, 2019, \$635 effective April 1, 2020 and \$635 effective April 1, 2021.

Any funds that may remain in the Service Improvement Fund will be applied to these or other hard to recruit or hard to retain classifications after discussions between the employer and the Union.

The Service Improvement Fund shall apply to actively employed members, including employees on leave, on the date of ratification.

LETTER OF UNDERSTANDING #2 Public Sector General Wage Increase

(1) If a public sector employer as defined in s. 1 of the *Public Sector Employers Act* enters into a collective agreement with an effective date after December 31, 2018 and the first three years of the collective agreement includes a cumulative nominal (not compounded) general wage increase of more than 6%, the general wage increase in the 2019-2022 collective agreement will be adjusted on the third anniversary of the 2019-2022 collective agreement so the cumulative nominal (not compounded) general wage increases are equivalent. This letter of understanding is not triggered by any general wage increase awarded as a result of binding interest arbitration.

(2) A general wage increase and its magnitude in any agreement is as defined by the PSEC Secretariat and reported by the Secretariat to the Minister of Finance.

(3) For certainty, a general wage increase is one that applies to all members of a bargaining unit and does not include wage comparability adjustments, targeted lower wage redress adjustments, labour market adjustment, service improvement allocations, and is net of the value of any changes agreed to by a bargaining agent for public sector employees to obtain a compensation adjustment.

(4) This letter of understanding will be in effect during the term of the 2019-2022 collective agreement.

LETTER OF UNDERSTANDING #3 Joint Training

The parties agree to jointly develop and facilitate a labour relations training session no later than four months after ratification of this agreement.

Participants would include BCGEU bargaining committee members, shop stewards, Senior Managers and Enforcement Managers. The training session will be without loss of pay.

LETTER OF UNDERSTANDING #4 Low Wage Redress

In advance of the development of the next provincial bargaining mandate, the Parties will recommend to PSEC that further significant progress towards parity with equivalent positions in the Public Service should be made in the next round of bargaining.

LETTER OF UNDERSTANDING #5

The Parties agree to refer the matter of the Telework Policy to the joint Union/Employer Committee pursuant to Clause 7.5 (Joint Union/Employer Committee) for review and discussion.

move**up** 07000225v2